

Operator algebras of free wreath products

Pierre Fima, Arthur Troupel

▶ To cite this version:

Pierre Fima, Arthur Troupel. Operator algebras of free wreath products. Advances in Mathematics, 2024, 441, pp.109546. 10.1016/j.aim.2024.109546. hal-04496463v2

HAL Id: hal-04496463 https://hal.science/hal-04496463v2

Submitted on 8 Mar 2024

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers.

L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.





Contents lists available at ScienceDirect

Advances in Mathematics

journal homepage: www.elsevier.com/locate/aim



Operator algebras of free wreath products



Pierre Fima*,1, Arthur Troupel*

Université Paris Cité and Sorbonne Université, CNRS, IMJ-PRG, F-75013 Paris, France

ARTICLE INFO

Article history:
Received 13 February 2023
Received in revised form 15
December 2023
Accepted 27 January 2024
Available online xxxx
Communicated by Dan Voiculescu

MSC: primary 46L67 secondary 46L10, 46L80

Keywords:
Operator algebras
Compact quantum groups
KK-theory
Von Neumann algebras

ABSTRACT

We give a description of operator algebras of free wreath products in terms of fundamental algebras of graphs of operator algebras as well as an explicit formula for the Haar state. This allows us to deduce stability properties for certain approximation properties such as exactness, Haagerup property, hyperlinearity and K-amenability. We study qualitative properties of the associated von Neumann algebra: factoriality, fullness, primeness and absence of Cartan subalgebra and we give a formula for Connes' T-invariant and τ -invariant. We also study maximal amenable von Neumann subalgebras. Finally, we give some explicit computations of K-theory groups for C*-algebras of free wreath products. As an application we show that the reduced C*-algebras of quantum reflection groups are pairwise non-isomorphic.

© 2024 The Authors. Published by Elsevier Inc. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

1. Introduction

After Woronowicz developed the general theory of compact quantum groups [60–62] which covers, in particular, the deeply investigated q-deformations of compact Lie groups

^{*} Corresponding authors.

E-mail addresses: pierre.fima@imj-prg.fr (P. Fima), arthur.troupel@imj-prg.fr (A. Troupel).

¹ P.F. is partially supported by the ANR project ANCG (No. ANR-19-CE40-0002), ANR project AODynG (No. ANR-19-CE40-0008), and Indo-French Centre for the Promotion of Advanced Research - CEFIPRA.

constructed by Drinfeld and Jimbo [20,21,33], new examples of compact quantum groups emerged from the works of Wang [57,58]. A particularly relevant quantum group for the present work is the quantum permutation group S_N^+ (see Section 2.6) for which the representation theory has been completely understood by Banica [1] and its operator algebras (as well as approximation properties) have been extensively studied by Brannan and Voigt [9,54]. In order to find natural quantum subgroups of S_N^+ , quantum automorphism groups of finite graphs were introduced in [7]. Since the classical automorphism group of the graph obtained as the N-times disjoint union of a finite connected graph $\mathcal G$ is given by the wreath product of the classical automorphism group of $\mathcal G$ by S_N , it became important to introduce a free version of the wreath product.

The free wreath product construction was achieved by Bichon in [8]: given a compact quantum group G, Bichon constructed and studied the free wreath product $G \wr_* S_N^+$ of G by S_N^+ (see Section 2.7). Some generalizations of free wreath products by S_N^+ were considered in [39,27,46,29,28]. Concerning the representation theory of $G \wr_* S_N^+$, it has been studied in many specific cases before achieving the general result [10,36,37,39,27, 46,29]. While some approximation properties are known to be preserved by the free wreath product construction, very little is known about the operator algebras associated to $G \wr_* S_N^+$. It seems that the main reason why these operator algebras are not already completely understood is that there is no explicit computation of the Haar state. The first result of the present paper is an explicit formula for the Haar state of $G \wr_* S_N^+$ (Theorem 3.2). While studying free wreath products, we realized that the full C*-algebra $C(G \wr_* S_N^+)$ of the free wreath product can actually be written as the full fundamental C*-algebra of a graph of C*-algebras [22]. However, in contrast with [22], $G \wr_* S_N^+$ is not the fundamental quantum group of a graph of quantum groups. Nevertheless, our description in terms of graph of C*-algebras allows us to produce a specific state, called the fundamental state in [22], which is shown to be the Haar state of $C(G \wr_* S_N^+)$. This allows a complete description of the reduced C*-algebra $C_r(G \wr_* S_N^+)$ as the reduced fundamental C*-algebra of a graph of C*-algebras as well as a description of the von Neumann algebra $L^{\infty}(G \wr_* S_N^+)$ as the fundamental von Neumann algebra of a graph of von Neumann algebras.

Concerning approximation properties, our description in terms of graph of operator algebras of the free wreath product allows us to deduce the following. To state our result, we will the following terminology: we will always denote by \widehat{G} the discrete dual of a compact quantum group G, and we say that \widehat{G} has the Haagerup property if $L^{\infty}(G)$ has the Haagerup property (with respect to the Haar state see [15]) and that \widehat{G} is exact when the reduced C*-algebra $C_r(G)$ is an exact C*-algebra. When G is Kac, we say that its dual \widehat{G} is hyperlinear if the finite von Neumann algebra $L^{\infty}(G)$ embeds in an ultraproduct of the hyperfinite II₁-factor. Finally, G is called co-amenable when the canonical surjection from its full C*-algebra C(G) to $C_r(G)$ is an isomorphism, and we say that \widehat{G} is K-amenable when this canonical surjection is a KK-equivalence.

Theorem A. Given $N \in \mathbb{N}^*$ and G a compact quantum group. The following holds.

- (1) \widehat{G} is exact if and only if $\widehat{G} \wr_* \widehat{S}_N^+$ is exact.
- (2) \widehat{G} has the Haagerup property if and only if $\widehat{G} \wr_* \widehat{S}_N^+$ has the Haagerup property.
- (3) If G is Kac then \widehat{G} is hyperlinear if and only if $\widehat{G} \wr_* \widehat{S}_N^+$ is hyperlinear.
- (4) If $N \geq 5$, $G \wr_* S_N^+$ is not co-amenable for any G.
 - If $N \in \{3,4\}$ then $G \wr_* S_N^+$ is co-amenable if and only if G is trivial.
 - $G \wr_* S_2^+$ is co-amenable if and only if G is trivial or if $G \simeq \mathbb{Z}/2\mathbb{Z}$.
- (5) \widehat{G} is K-amenable if and only if $\widehat{G} \wr_* \widehat{S}_N^+$ is K-amenable.

Let us mention that some of the statements in Theorem A are already known in specific cases but our proofs, based on our description in terms of graphs of operator algebras, are completely different. The stability of exactness under the free wreath product with S_N^+ is known [37,27] for $N \geq 4$ while the stability of the Haagerup property is not known in the form above. Actually, most of the previous proofs of approximation properties were based on a monoidal equivalence argument so that it was not possible to deduce properties which are not stable under monoidal equivalence. To deduce approximation properties, the common trick was to strengthen the definition of these properties so that they become invariant under monoidal equivalence. Hence, what is known in relation with the Haagerup property is actually the stability of the central ACPAP [36,37,27,46]. One exception is the case N=2 which has been completely described by Bichon [7]. From Bichon's description, it has been possible to deduce in [19, Corollary 7.11] that, when G is Kac, $\widehat{G}_{l*}\widehat{S}_2^+$ has the Haagerup property if and only if \widehat{G} has the Haagerup property. Concerning hyperlinearity, it is already known that $\widehat{S_N^+}$ (for all N) and the discrete duals of the quantum reflection groups $\widehat{\mathbb{Z}}_s \wr_* S_N^+$ (for $1 \leq s \leq +\infty$ and $N \geq 4$) are residually finite hence hyperlinear [2]. The now classical argument is based on the topological generation method developed by Brannan-Collins-Vergnioux [3]. The stability of hyperlinearity of Theorem A is completely new and our argument is not based on the topological generation method. The stability of K-amenability is also new. It was known that \hat{S}_N^+ is K-amenable [54] and, again by using a monoidal equivalence argument, it was known [26] that if G is torsion-free and satisfies the strong Baum-Connes conjecture then $G \wr_* S_N^+$ is K-amenable. One of the consequences of our result on K-amenability which seems to have been unknown is that duals of quantum reflection groups are Kamenable. A more interesting application of our description in terms of graph of operator algebras is about the qualitative properties of the von Neumann algebra $L^{\infty}(G_{k}, S_{N}^{+})$ for which nothing is known, outside of the factoriality in some specific cases. It is known that $L^{\infty}(G \wr_* S_N^+)$ is a full II_1 -factor whenever $G = \widehat{\Gamma}$ is the dual of a discrete group and $N \geq 8$ [36] and it is also a non-Gamma II_1 -factor whenever G is a compact matrix quantum group of Kac type and $N \geq 8$ [56]. Using results about the structure of amalgameted free products von Neumann algebra [49,50,32,31] we are able to deduce the following. Let $(\sigma_t^G)_t$ be the modular group of the Haar state on $L^{\infty}(G)$ and T(M) the Connes' T invariant of the von Neumann algebra M.

Theorem B. Suppose that Irr(G) is infinite and $N \geq 4$. Then $L^{\infty}(G \wr_* S_N^+)$ is a non-amenable full and prime factor without any Cartan subalgebra. Moreover:

- If G is Kac then $L^{\infty}(G \wr_* S_N^+)$ is a type II_1 factor.
- If G is not Kac then $L^{\infty}(G \wr_* S_N^+)$ is a type III_{λ} factor for some $\lambda \neq 0$ and:

$$T(\mathcal{L}^{\infty}(G \wr_* S_N^+)) = \{ t \in \mathbb{R} : \sigma_t^G = \mathrm{id} \}.$$

Moreover, the Connes' τ -invariant of the full factor $L^{\infty}(G \wr_* S_N^+)$ is the smallest topology on \mathbb{R} for which the map $(t \mapsto \sigma_t^G)$ is continuous.

To our knowledge, there was no result about the maximal amenable von Neumann subalgebras of $L^{\infty}(G \wr_* S_N^+)$ in the literature. A von Neumann subalgebra $A \subset M$ is called with expectation if there a normal faithful conditional expectation $P \to A$. $A \subset P$ is called maximal amenable if A is amenable and, for any intermediate von Neumann algebra $A \subset P \subset M$ if P is amenable then A = P. $A \subset P$ is called maximal amenable with expectation if A is amenable with expectation and, for any $A \subset P \subset M$, if P is amenable with expectation then A = P.

Recall that $L^{\infty}(G \wr_* S_N^+)$ is generated by N free copies of $L^{\infty}(G)$ given by $\nu_i : L^{\infty}(G) \to L^{\infty}(G \wr_* S_N^+)$, $1 \le i \le N$, and by $L^{\infty}(S_N^+) \subset L^{\infty}(G \wr_* S_N^+)$ plus some relations (see Section 2.7). Let $u = (u_{ij})_{ij} \in M_N(\mathbb{C}) \otimes L^{\infty}(S_N^+)$ be the fundamental representation of S_N^+ so that $L^{\infty}(S_N^+)$ is generated by the coefficients u_{ij} of u. We use [5] to deduce the following.

Theorem C. Let G be a compact quantum group and $N \geq 2$. The following holds.

- (1) If $L^{\infty}(G)$ is amenable and Irr(G) is infinite then, for all $1 \leq i \leq N$, the von Neumann subalgebra of $L^{\infty}(G \wr_* S_N^+)$ generated by $\{\nu_i(a)u_{ij} : a \in L^{\infty}(G), 1 \leq j \leq N\}$ is maximal amenable with expectation in $L^{\infty}(G \wr_* S_N^+)$.
- (2) If G is Kac then $L^{\infty}(S_4^+) \subset L^{\infty}(G \wr_* S_4^+)$ is maximal amenable.

Finally, we compute K-theory groups of the C*-algebras $C(G \wr_* S_N^+)$ and $C_r(G \wr_* S_N^+)$ (recall that they are KK-equivalent whenever \widehat{G} is K-amenable). In the next Theorem, we denote by $C_{\bullet}(G)$ either the full C^* -algebra or the reduced C^* -algebra of G.

Theorem D. For any compact quantum group G and integer $N \in \mathbb{N}^*$ we have,

$$K_0(C_{\bullet}(G \wr_* S_N^+)) \simeq K_0(C_{\bullet}(G)) \otimes (\mathbb{Z}^{N^2}) \oplus K_0(C(S_N^+))/(\mathbb{Z}^{N^2})$$

$$\simeq \begin{cases} K_0(C_{\bullet}(G))^{\oplus N^2}/\mathbb{Z}^{2N-2} & \text{if } N \neq 3, \\ K_0(C_{\bullet}(G))^{\oplus N^2}/\mathbb{Z}^3 & \text{if } N = 3. \end{cases} \text{ and,}$$

$$K_1(C_{\bullet}(G \wr_* S_N^+)) \simeq K_1(C_{\bullet}(G))^{\oplus N^2} \oplus K_1(C(S_N^+))$$

$$\simeq \begin{cases} K_1(C_{\bullet}(G))^{\oplus N^2} \oplus \mathbb{Z} & \text{if } N \ge 4, \\ K_1(C_{\bullet}(G))^{\oplus N^2} & \text{if } 1 \le N \le 3. \end{cases}$$

In particular, for the quantum reflection groups $H_N^{s+} = \widehat{\mathbb{Z}}_s \wr_* S_N^+$, which have K-amenable duals, if $N \geq 4$,

$$K_0(C_{\bullet}(H_N^{s+})) \simeq \begin{cases} \mathbb{Z}^{N^2-2N+2} & \text{if } s = +\infty, \\ \mathbb{Z}^{sN^2-2N+2} & \text{if } s < +\infty, \end{cases} K_1(C_{\bullet}(H_N^{s+})) \simeq \begin{cases} \mathbb{Z}^{N^2+1} & \text{if } s = +\infty, \\ \mathbb{Z} & \text{if } s < +\infty, \end{cases}$$

and if $N \in \{1, 2, 3\}$,

$$K_0(C_{\bullet}(H_N^{s+})) \simeq \begin{cases} \mathbb{Z}^{N!} & \text{if } s = +\infty, \\ \mathbb{Z}^{(s-1)N^2 + N!} & \text{if } s < +\infty, \end{cases} \quad K_1(C_{\bullet}(H_N^{s+})) \simeq \begin{cases} \mathbb{Z}^{N^2} & \text{if } s = +\infty, \\ 0 & \text{if } s < +\infty, \end{cases}$$

It seems that the only attempt to compute the K-theory of free wreath products was done in [26] in which they compute K-theory groups of some quantum groups which are not free wreath product with S_N^+ but only monoidally equivalent to a free wreath products with S_N^+ . Actually the K-theory is computed for some quantum groups of the form $G \wr_* SO_q(3)$, where the free wreath product is in the sense of [27]. The method of computation in [26], which is based on the ideas of the works of Voigt [53,54] and Voigt-Vergnioux [55] is to prove the strong Baum-Connes property and then find an explicit projective resolution of the trivial action. However, they could not do it for free wreath products by S_N^+ but they showed that this can be done for $G_{l_*}SO_q(3)$ with some specific G such as free products of free orthogonal and free unitary quantum groups as well as duals of free groups. While our method cannot provide results on the stability of the strong Baum-Connes conjecture for free wreath products, it has the advantage to be extremely simple and also applicable to any free wreath product with S_N^+ . Our K-theory computations are based on our decomposition as fundamental algebras and the results of [23]. We also use the result of [54] on the computation of the K-theory of $C_{\bullet}(S_N^+)$. As mentioned to us by Adam Skalski, it seems unknown whether or not the reduced C*-algebras of quantum reflection groups H_N^{s+} are isomorphic or not for different values of the parameters N, s. Our K-theory computation in Theorem D solves this question: we show in Corollary 7.1 that for all $N, M \geq 8$ one has $C_r(H_N^{s+}) \simeq C_r(H_M^{t+}) \Leftrightarrow (N, s) =$ (M,t). We do have to restrict the parameters to $N,M\geq 8$ since our proof uses the unique trace result of Lemeux [36], which is valid only for $N \geq 8$. We also compute some other K-theory groups of free wreath products, such as $\widehat{\mathbb{F}}_m \wr_* S_N^+$ and deduce that reduced (as well as full) C*-algebras $C_{\bullet}(\widehat{\mathbb{F}}_m \wr_* S_N^+)$ are pairwise non-isomorphic.

The paper is organized as follows. After the introduction in Section 1, we have included preliminaries in Section 2 in which we detail all our notations concerning operator algebras and quantum groups. We also prove in this preliminary Section that the von Neumann algebra $L^{\infty}(G)$ of a compact quantum group G is diffuse if and only if it is infinite-dimensional. This result, which will be useful in the study of von Neumann algebras of free wreath product, also solves a question left open in [2] in which it is mentioned

that it is unknown if the von Neumann algebra of a quantum reflection group $\mathbb{Z}_s \wr_* S_N^+$ is diffuse when $N \leq 7$. Section 2 also contains important remarks on free product quantum groups, semi-direct product quantum groups, quantum permutation groups and free wreath product quantum groups, as well as amalgamated free wreath products (in the sense of Freslon [28]). In particular, we identify the amalgamated free wreath product $G \wr_{*,H} S_N^+$ at N=2 with a semi-direct product. In Section 3 we explain how operator algebras of amalgamated free wreath products can be described as fundamental algebras of graphs of operator algebras. Section 4 contains the proof of the amalgamated version of Theorem A, Section 5 contains the proof of the amalgamated versions of Theorem B and C. In Section 6, we give a formula describing an amalgamated free wreath product of a fundamental quantum group of a graph of quantum groups by S_N^+ as the fundamental quantum group of a graph of quantum groups given by free wreath products. Finally, section 7 contains the proof of Theorem D and some other K-theory computations.

2. Preliminaries

2.1. Notations

All Hilbert spaces, C*-algebras and preduals of von Neumann algebras considered in this paper are assumed to be separable. The inner product on an Hilbert space H is always linear on the right. The C*-algebra of bounded linear maps from H to H is denoted by $\mathcal{B}(H)$ and, given vectors $\xi, \eta \in H$ we denote by $\omega_{\eta,\xi} \in \mathcal{B}(H)^*$ the bounded linear form on $\mathcal{B}(H)$ defined by $(x \mapsto \langle \eta, x\xi \rangle)$. For $T \in \mathcal{B}(H)$, we denote by $\operatorname{Sp}(T)$ the spectrum of T and by $\operatorname{Sp}_p(T)$ the point spectrum of T. The symbol \otimes will denote the tensor product of Hilbert spaces, von Neumann algebras as well as the minimal tensor product of C*-algebras.

2.2. Full von Neumann algebras

Let M be a von Neumann algebra with predual M_* . We consider on $\operatorname{Aut}(M)$ the topology of pointwise convergence in M_* i.e. the smallest topology for which the maps $\operatorname{Aut}(M) \to M_*$, $\alpha \mapsto \omega \circ \alpha$ are continuous, for all $\omega \in M_*$, where M_* is equipped with the norm topology. It is well known that $\operatorname{Aut}(M)$ is a Polish group (since M_* is separable). When $\omega \in M_*$ is a faithful normal state, we may consider the closed subgroup $\operatorname{Aut}(M,\omega) < \operatorname{Aut}(M)$ of automorphisms preserving ω . Note that the induced topology from $\operatorname{Aut}(M)$ to $\operatorname{Aut}(M,\omega)$ is the smallest topology making the maps $\operatorname{Aut}(M,\omega) \to \operatorname{L}^2(M,\omega)$, $\alpha \mapsto \alpha(a)\xi_\omega$ continuous, for the norm on $\operatorname{L}^2(M,\omega)$, where ξ_ω is the cyclic vector in the GNS construction $\operatorname{L}^2(M,\omega)$ of ω . We record the following Remark for later use.

Remark 2.1. Note that if (M, ω) is a von Neumann algebra with a faithful normal state ω and $A \subset M$ is a von Neumann subalgebra with a ω -preserving normal conditional

expectation $M \to A$ then, the subgroup $\operatorname{Aut}_A(M,\omega) = \{\alpha \in \operatorname{Aut}(M,\omega) : \alpha(A) = A\} < \operatorname{Aut}(M,\omega)$ is closed and the restriction map $\operatorname{Aut}_A(M,\omega) \to \operatorname{Aut}(A,\omega), \ \alpha \mapsto \alpha|_A$ is continuous.

Let $\mathrm{Inn}(M) \subset \mathrm{Aut}(M)$ be the normal subgroup of inner automorphisms and $\mathrm{Out}(M) := \mathrm{Aut}(M)/\mathrm{Inn}(M)$ be the quotient group. Equipped with the quotient topology, $\mathrm{Out}(M)$ is a topological group and it is Hausdorff if and only if $\mathrm{Inn}(M) \subset \mathrm{Aut}(M)$ is closed. In that case (since M_* is assumed to be separable) $\mathrm{Out}(M) := \mathrm{Aut}(M)/\mathrm{Inn}(M)$ is actually a metrizable topological group so the convergence of sequences in $\mathrm{Out}(M)$ completely characterizes the topology. Following Connes [14], a von Neumann algebra is called full if $\mathrm{Inn}(M) \subset \mathrm{Aut}(M)$ is closed. A normal faithful semi-finite weight φ on M is called full the invariant full if full the invariant full of a full von Neumann algebra as the intersection of point spectrum of full the invariant full semi-finite almost-periodic weight and shows that the closure of full is the full semi-finite almost-periodic weight and shows that the closure of full is the full semi-finite almost-periodic weight and

The famous noncommutative Radon-Nykodym Theorem of Connes [13] shows that, for any pair of normal faithful states (actually semi-finite weights) ω_1 and ω_2 on M, their modular groups $\sigma_t^{\omega_1}$ and $\sigma_t^{\omega_2}$ have the same image in $\operatorname{Out}(M)$, for all $t \in \mathbb{R}$. Hence, there is a well defined homomorphism $\delta : \mathbb{R} \to \operatorname{Out}(M)$, called the *modular homomorphism*, defined by $\delta(t) = [\sigma_t^{\omega}]$, where ω is any normal faithful state on M and $[\cdot]$ denotes the class in $\operatorname{Out}(M)$.

Connes defines in [14], the invariant $\tau(M)$ of a full von Neumann algebra M as the smallest topology on \mathbb{R} for which the modular homomorphism δ is continuous. When M is full (and M_* is separable), $\operatorname{Out}(M)$ is clearly a metrizable topological group hence, $(\mathbb{R}, \tau(M))$ is also a metrizable topological group. In particular, the topology $\tau(M)$ is completely characterized by the knowledge of which sequences are converging to zero.

2.3. Compact quantum groups

For a discrete group Γ we denote by $C^*(\Gamma)$ its full C*-algebra, $C_r^*(\Gamma)$ its reduced C*-algebra and L(Γ) its von Neumann algebra. We briefly recall below some elements of the compact quantum group (CQG) theory developed by Woronowicz [60–62].

For a CQG G, we denote by C(G) its maximal C*-algebra, which is the enveloping C*-algebra of the unital *-algebra Pol(G) given by the linear span of coefficients of irreducible unitary representations of G. The set of equivalence classes of irreducible unitary representations will be denoted by Irr(G). We will denote by $\varepsilon_G: C(G) \to \mathbb{C}$ the counit of G which satisfies $(id \otimes \varepsilon_G)(u) = id_H$ for all finite dimensional unitary representations $u \in \mathcal{B}(H) \otimes C(G)$.

Let us recall below the modular ingredients of a CQG. Let us fix a complete set of representatives $u^x \in \mathcal{B}(H_x) \otimes C(G)$ for $x \in \operatorname{Irr}(G)$. It is known that for any $x \in \operatorname{Irr}(G)$ there exists a unique $\overline{x} \in \operatorname{Irr}(G)$, called the contragredient of x, such that $\operatorname{Mor}(1, x \otimes \overline{x}) \neq \{0\}$ and $\operatorname{Mor}(1, \overline{x} \otimes x) \neq \{0\}$, where 1 denotes the trivial representation of G.

Both the spaces $\operatorname{Mor}(1, \overline{x} \otimes x)$ and $\operatorname{Mor}(1, x \otimes \overline{x})$ are actually one-dimensional. Fix non-zero vectors $s_x \in H_x \otimes H_{\overline{x}}$ and $s_{\overline{x}} \in H_{\overline{x}} \otimes H_x$ such that $s_x \in \operatorname{Mor}(1, x \otimes \overline{x})$ and $s_{\overline{x}} \in \operatorname{Mor}(1, \overline{x} \otimes x)$. Let $J_x : H_x \to H_{\overline{x}}$ be the unique invertible antilinear map satisfying $\langle J_x \xi, \eta \rangle = \langle s_x, \xi \otimes \eta \rangle$ for all $\xi \in H_x$, $\eta \in H_{\overline{x}}$ and define the positive invertible operator $Q_x = J_x^* J_x \in \mathcal{B}(H_x)$. Then, there exists a unique normalization of s_x and $s_{\overline{x}}$ such that $||s_x|| = ||s_{\overline{x}}||$ and $J_{\overline{x}} = J_x^{-1}$. With this normalization, Q_x is uniquely determined, we do have $\operatorname{Tr}(Q_x) = \operatorname{Tr}(Q_x^{-1}) = ||s_x||^2$, $Q_{\overline{x}} = (J_x J_x^*)^{-1}$ and $\operatorname{Sp}(Q_{\overline{x}}) = \operatorname{Sp}(Q_x)^{-1}$ (Tr is the unique trace on $\mathcal{B}(H_x)$ such that $\operatorname{Tr}(1) = \dim(H_x)$). The number $\operatorname{Tr}(Q_x)$ is called the quantum dimension of x and is denoted by $\dim_g(x)$. From the orthogonality relations:

$$(\mathrm{id} \otimes h_G)((u^x)^*(\xi\eta^* \otimes 1)u^y) = \frac{\delta_{x,y}1}{\dim_q(x)} \langle \eta, Q_x^{-1}\xi \rangle$$

it is not difficult to check that, with $L^2(G) = \bigoplus_{x \in Irr(G)} H_x \otimes H_{\overline{x}}, \xi_G := 1 \in H_1 \otimes H_{\overline{1}} = \mathbb{C}$ and $\lambda_G : C(G) \to \mathcal{B}(L^2(G))$ the unique unital *-homomorphism such that

$$\lambda_G((\omega_{\eta,\xi} \otimes \mathrm{id})(u^x))\xi_G = \dim_q(x)^{-\frac{1}{2}}\xi \otimes J_{\overline{x}}^*\eta,$$

 ξ_G is λ_G -cyclic and $h_G(x) = \langle \xi_G, \lambda_G(a) \xi_G \rangle \ \forall a \in C(G)$. Hence, the triple $(L^2(G), \lambda_G, \xi_G)$ is an explicit GNS construction for the Haar state h_G on C(G).

Let $C_r(G) = \lambda_G(C(G)) \subset \mathcal{B}(L^2(G))$ be the reduced C*-algebra of G. The surjective unital *-homomorphism $\lambda_G: C(G) \to C_r(G)$ is called the canonical surjection. Recall that G is called co-amenable whenever λ_G is injective. The von Neumann algebra of G is denoted by $L^{\infty}(G) := C_r(G)'' \subset \mathcal{B}(L^2(G))$. We will still denote by h_G the Haar state of G on the C*-algebra $C_r(G)$ as well as on the von Neumann algebra $L^{\infty}(G)$ i.e. $h_G = \langle \xi_G, \xi_G \rangle$ when viewed as a state on $C_r(G)$ or a normal state on $L^{\infty}(G)$. We recall that h_G is faithful on both $C_r(G)$ and $L^{\infty}(G)$. With the explicit GNS construction of h_G given above, it is not difficult to compute the modular ingredients of the normal faithful state h_G on $L^{\infty}(G)$ and we find that the closure S_G of the antilinear operator $x\xi_G \mapsto x^*\xi_G$ has a polar decomposition $S_G = J_G \nabla_G^{\frac{1}{2}}$, where ∇_G is the positive operator on $L^2(G)$ given by $\nabla_G := \bigoplus_{x \in \operatorname{Irr}(G)} Q_x \otimes Q_{\overline{x}}^{-1}$. Hence, the Haar state of a CQG is always almost-periodic and its modular group $(\sigma_t^G)_t$ is the unique one-parameter group of $L^{\infty}(G)$ such that $(id \otimes \sigma_t^G)(u^x) = (Q_x^{it} \otimes 1)u^x(Q_x^{it} \otimes 1)$. Let us recall that G is said to be of Kac type whenever h_G is a trace and it happens if and only if $Q_x = 1$ for all $x \in \operatorname{Irr}(G)$. Let us also recall that the scaling group $(\tau_t^G)_t$ is the one-parameter group of $L^{\infty}(G)$ given by the formula $(id \otimes \tau_t^G)(u^x) = (Q_x^{it} \otimes 1)u^x(Q_x^{-it} \otimes 1)$. It is well known that the scaling group of G is the unique one parameter group $(\tau_t^G)_t$ of $L^{\infty}(G)$ such that $\Delta \circ \sigma_t^G = (\tau_t^G \otimes \sigma_t^G) \circ \Delta \ \forall t \in \mathbb{R}$. Moreover, the scaling group preserves that Haar state: $h_G \circ \tau_t^G = h_G$ (this means that the *scaling constant* of a compact quantum group is 1).

Let us also recall the definition of the T-invariant of a CQG G, introduced by S. Vaes in [51]:

$$T(G) := \{ t \in \mathbb{R} : \exists u \in \mathcal{U}(L^{\infty}(G)) : \Delta(u) = u \otimes u \text{ and } \tau_t^G = \mathrm{Ad}(u) \},$$

where $\operatorname{Ad}(u)$ is the automorphism of $\operatorname{L}^{\infty}(G)$ given by $a\mapsto uau^*$.

The following result is well known [25, Proposition 2.2].

Proposition 2.2. The set $Mod(G) := \bigcup_{x \in Irr(G)} Sp(Q_x)$ is a subgroup of \mathbb{R}_+^* .

Proof. We already remarked in Section 2.3 that $\operatorname{Sp}(Q_{\overline{x}}) = (\operatorname{Sp}(Q_x))^{-1}$. Hence $\operatorname{Mod}(G)$ is stable by inverse. The fact that $\operatorname{Mod}(G)$ is stable by product follows from the relation

$$SQ_z = (Q_x \otimes Q_y)S,$$

for any $x, y, z \in Irr(G)$, and isometry $S \in Mor(z, x \otimes y)$. \square

Remark 2.3. For convenience, we will use the following non-standard notations.

- (1) Let $Sd(G) := \operatorname{Sp}_p(\nabla_G)$. From the explicit computation of ∇_G and since $\operatorname{Sp}(Q_{\overline{x}}) = \operatorname{Sp}(Q_x)^{-1}$ one has $Sd(G) = \bigcup_{x \in \operatorname{Irr}(G)} \operatorname{Sp}(Q_x)^2 \subset \operatorname{Mod}(G)$ and $Sd(G)^{-1} = Sd(G)$.
- (2) We denote by $\tau(G)$ the smallest topology on \mathbb{R} such that the map $(t \mapsto \sigma_t^G)$ is continuous. Hence $\tau(G)$ is smaller than the usual topology on \mathbb{R} . It is not difficult to check that it is the smallest topology on \mathbb{R} for which the maps $f_{\lambda} : \mathbb{R} \to \mathbb{S}^1 : (t \mapsto \lambda^{it})$ are continuous, for all $\lambda \in Sd(G)$, where \mathbb{S}^1 has the usual topology. Note also that, for any topology τ on \mathbb{R} , the set of $\lambda > 0$ for which f_{λ} is τ -continuous is a closed subgroup of \mathbb{R}_+^* (for the usual topology on \mathbb{R}_+^*). Hence, $\tau(G)$ is also the smallest topology on \mathbb{R} for which the maps f_{λ} are continuous, for all $\lambda \in \overline{\langle Sd(G) \rangle}$, the closed subgroup of \mathbb{R}_+^* generated by Sd(G) (for the usual topology on \mathbb{R}_+^*). Moreover, the following is easy to check:
 - G is Kac $\Leftrightarrow Sd(G) = \{1\} \Leftrightarrow \tau(G)$ is the trivial topology. If $Sd(G) \neq \{1\}$ then $(\mathbb{R}, \tau(G))$ is a metrizable topological group so the topology $\tau(G)$ is completely characterized by the knowledge of which sequences are converging to zero.
 - $\overline{\langle Sd(G)\rangle} = \mathbb{R}_+^*$ if and only if $\tau(G)$ is the usual topology on \mathbb{R} .

Recall that a von Neumann algebra is diffuse when it has no non-zero minimal projection. We will use the following simple Lemma. While the arguments for its proof are already present in the literature, the precise statement seems to be unknown.

Lemma 2.4. Let G be a compact quantum group. The following are equivalent.

- (1) Irr(G) is infinite.
- (2) $L^{\infty}(G)$ is diffuse.
- (3) C(G) is infinite-dimensional.

Proof. The implication $(2) \Rightarrow (1)$ is obvious. Let us show $(1) \Rightarrow (2)$. By the general theory, a von Neumann algebra is diffuse if and only if it has no direct summand of the form $\mathcal{B}(H)$. When H is finite dimensional, we may apply [18, Theorem 3.4] (since the

action of G on $L^{\infty}(G)$ given by the comultiplication is ergodic) to deduce that $\mathcal{B}(H)$ is not a direct summand of $L^{\infty}(G)$ whenever Irr(G) is infinite. When H is infinite-dimensional, we may apply [35, Theorem 6.1] to deduce that $\mathcal{B}(H)$ is not a direct summand of $L^{\infty}(G)$ whenever Irr(G) is infinite. The equivalence between (1) and (3) is clear. \square

Let us now recall the notion of a dual quantum subgroup. We are grateful to Kenny De Commer for explaining to us the main argument of the proof of the next Proposition and to Makoto Yamashita for showing to us the reference [11]. Recall that $C_{\bullet}(G)$ denotes either the reduced or the maximal C*-algebra of G.

Proposition 2.5. Let G and H be CQG. The following data are equivalent.

- $\iota: C(H) \to C(G)$ is a faithful unital *-homomorphism intertwining the comultiplications.
- $\iota : \operatorname{Pol}(H) \to \operatorname{Pol}(G)$ is a faithful unital *-homomorphism intertwining the comultiplications.
- $\iota: C_r(H) \to C_r(G)$ is a faithful unital *-homomorphism intertwining the comultiplications.

If one of the following equivalent conditions is satisfied, we view $\operatorname{Pol}(H) \subset \operatorname{Pol}(G)$, $C(H) \subset C(G)$ and $C_r(H) \subset C_r(G)$. Then, the unique linear map $E : \operatorname{Pol}(G) \to \operatorname{Pol}(H)$ such that

$$(\mathrm{id} \otimes E)(u^x) = \begin{cases} u^x & \text{if } x \in \mathrm{Irr}(H), \\ 0 & \text{if } x \in \mathrm{Irr}(G) \setminus \mathrm{Irr}(H). \end{cases}$$

has a unique ucp extension to a map $E_{\bullet}: C_{\bullet}(G) \to C_{\bullet}(H)$ which is a Haar-state-preserving conditional expectation onto the subalgebra $C_{\bullet}(H) \subset C_{\bullet}(G)$. At the reduced level, E_r is faithful and extends to a Haar-state preserving normal faithful conditional expectation $L^{\infty}(G) \to L^{\infty}(H)$.

Proof. If $\iota: C(H) \to C(G)$ is a faithful unital *-homomorphism intertwining the comultiplications then it is clear that its restriction to $\operatorname{Pol}(H)$ has image in $\operatorname{Pol}(G)$ and is still faithful. If now $\iota: \operatorname{Pol}(H) \to \operatorname{Pol}(G)$ is defined at the algebraic level then, since it is faithful and intertwines the comultiplications, it also intertwines the Haar states and so extends to a faithful unital *-homomorphism $\iota: C_r(H) \to C_r(G)$ which is easily seen to intertwine the comultiplications. It is proved in [52] that E extends to a faithful and Haar state preserving conditional expectation at the reduced level as well as at the von Neumann level (which is moreover normal). Also, if $\iota: C_r(H) \to C_r(G)$ is defined at the reduced level, its restriction to $\operatorname{Pol}(H)$ satisfies the second condition. Hence, it suffices to check that if $\operatorname{Pol}(H) \subset \operatorname{Pol}(G)$ is a unital *-subalgebra with the inclusion intertwining the comultiplications then, the canonical extension $\iota: C(H) \to C(G)$ of the

inclusion, which obviously also intertwines the comultiplications, is still faithful. Note that it suffices to show that E extends to a ucp map $E: C(G) \to C(H)$. Indeed, if we have such an extension then E has norm 1 and $E \circ \iota = \mathrm{id}_{C(H)}$, so for all $a \in Pol(H)$ one has $||a||_{C(H)} = ||E(\iota(a))||_{C(H)} \le ||\iota(a)||_{C(G)} \le ||a||_{C(H)}$. Then, ι is an isometry hence faithful. The fact that E extends to a ucp map is proved in [11, Theorem 3.1]. It is clear that the ucp extension preserves the Haar states since it already does at the algebraic level (by definition of E). Now, viewing $C(H) \subset C(G)$, E is ucp and is the identity on C(H) hence, it is a conditional expectation onto C(H).

If one of the above equivalent condition is satisfied, we say that H is a dual quantum subgroup of G and we will view $Pol(H) \subset Pol(G)$, $C(H) \subset C(G)$, $C_r(H) \subset C_r(G)$ as well as $L^{\infty}(H) \subset L^{\infty}(G)$. Let us note that, the ucp extension of E at the maximal level is not, in general, faithful and not even GNS-faithful (meaning that the GNS representation morphism may be non injective). We will usually denote E at the algebraic, full, reduced and von Neumann algebraic level by the same symbol E_H .

Remark 2.6. Let $C(H) \subset C(G)$ be a dual quantum subgroup and define $\operatorname{Pol}(G)^{\circ} := \{a \in \operatorname{Pol}(G) : E_H(a) = 0\}$ and $C_{\bullet}(G)^{\circ} := \{a \in C_{\bullet}(G)^{\circ} : E_{\bullet}(a) = 0\}$. Then $\operatorname{Pol}(G)^{\circ}$ is the linear span of coefficient of irreducible representations $x \in \operatorname{Irr}(G) \setminus \operatorname{Irr}(H)$, $C_{\bullet}(G)^{\circ}$ is the closure in $C_{\bullet}(G)$ of $\operatorname{Pol}(G)^{\circ}$ and $\Delta(\operatorname{Pol}(G)^{\circ}) \subset \operatorname{Pol}(G)^{\circ} \otimes \operatorname{Pol}(G)^{\circ}$. All these statements easily follow from the property of E_H stated in the previous proposition.

Let us recall that if $\mathcal{L}^{\infty}(H) \subset \mathcal{L}^{\infty}(G)$ is a dual compact quantum subgroup then $\sigma_t^G|_{\mathcal{L}^{\infty}(H)} = \sigma_t^H$ and $\tau_t^G|_{\mathcal{L}^{\infty}(H)} = \tau_t^H$ for all $t \in \mathbb{R}$.

Definition 2.7. Following Vergnioux [52], given a dual quantum subgroup $C(H) \subset C(G)$, we introduce an equivalence relation \sim_H on Irr(G) by defining $x \sim_H y \Leftrightarrow \exists s \in Irr(H)$, $Mor(s, \overline{x} \otimes y) \neq \{0\}$. Note in particular that $x \nsim_H y \Leftrightarrow (id \otimes E_H)(u^{\overline{x}} \otimes u^y) = 0$. We define the *index of H in G* by the number of equivalence classes $[G:H] := |Irr(G)/\sim_H|$.

2.4. Free product quantum group

We now recall some well known results about free product quantum groups. Given two compact quantum groups G_1 and G_2 the universal property of the C*-algebra given by the full free product $C(G) := C(G_1) * C(G_2)$ allows to define the unique unital *-homomorphism $\Delta : C(G) \to C(G) \otimes C(G)$ such that $\Delta|_{C(G_k)} = \Delta_{G_k}$ for k = 1, 2. It is easy to check that $G := (C(G), \Delta)$ is a CQG with maximal C*-algebra C(G), reduced C*-algebra given by the reduced free product with respect to the Haar states $C_r(G) = (C_r(G_1), h_1) * (C_r(G_2), h_2)$, and $L^{\infty}(G) = (L^{\infty}(G_1), h_1) * (L^{\infty}(G_2), h_2)$. Moreover the Haar state on $C_r(G)$ is the free product state $h = h_1 * h_2$.

We collect below some important remarks about free products. Most of them are well known to specialists. Since we could not find any explicit statements in the literature, we include a complete proof.

Proposition 2.8. Let G_1, G_2 be non-trivial CQG. The scaling group of $G := G_1 * G_2$ is the free product $\tau_t^G = \tau_t^{G_1} * \tau_t^{G_2}$ and the Vaes' T-invariant of $G_1 * G_2$ is given by:

$$T(G_1 * G_2) = \{ t \in \mathbb{R} : \tau_t^{G_1} = \tau_t^{G_2} = id \}.$$

The following are equivalent.

- (1) $G_1 * G_2$ is co-amenable.
- (2) $|\operatorname{Irr}(G_1)| = 2 = |\operatorname{Irr}(G_2)|$.
- (3) $G_1 \simeq G_2 \simeq \mathbb{Z}/2\mathbb{Z}$.
- (4) $L^{\infty}(G_1 * G_2)$ is an amenable von Neumann algebra.

Moreover, if one of the previous equivalent conditions does not hold then $L^{\infty}(G_1 * G_2)$ is a full and prime factor without any Cartan subalgebras and:

- If G_1 and G_2 are Kac then $L^{\infty}(G_1 * G_2)$ is a type II_1 factor.
- If G_1 or G_2 is not Kac then $L^{\infty}(G_1 * G_2)$ is a type III_{λ} factor with $\lambda \neq 0$ and its Connes' T-invariant is given by $\{t \in \mathbb{R} : \sigma_t^{h_1} = id = \sigma_t^{h_2}\}.$
- $Sd(L^{\infty}(G_1 * G_2)) = \langle Sd(G_1), Sd(G_2) \rangle$ and $\tau(L^{\infty}(G_1 * G_2)) = \langle \tau(G_1), \tau(G_2) \rangle$, where $\langle \cdot \rangle$ means either the group generated by or the topology generated by.

Proof. Since $h = h_1 * h_2$, we have $\sigma_t = \sigma_t^{G_1} * \sigma_t^{G_2}$, where σ_t denotes the modular group of h. Since $\tau_t^{G_k}$ is h_k -invariant, the free product $\tau_t := \tau_t^{G_1} * \tau_t^{G_2}$ makes sense and it defines a one parameter group of $L^{\infty}(G_1 * G_2)$. To show that it is the scaling group, we only have to check that $\Delta \circ \sigma_t = (\tau_t \otimes \sigma_t) \circ \Delta$, which is clear. Let $T' := \{t \in \mathbb{R} : \tau_t^{G_1} = \tau_t^{G_2} = \mathrm{id}\}$. It is clear that $T' \subseteq T(G_1 * G_2)$. Let $t \in T(G_1 * G_2)$ so that there exists $u \in L^{\infty}(G_1 * G_2)$ a unitary such that $\Delta(u) = u \otimes u$ and $\tau_t = \mathrm{Ad}(u)$. It follows that u is a dimension 1 unitary representation of $G_1 * G_2$, hence irreducible. If u is non-trivial, it follows from the classification of irreducible representations of $G_1 * G_2$ [57] that u is a product of non-trivial dimension 1 unitary representations alternating from $Irr(G_1)$ and $Irr(G_2)$ i.e. $u = u_1 u_2 \dots u_n$, where u_k is a unitary in $L^{\infty}(G_{i_k})$ with $\Delta_{G_{i_k}}(u_k) = u_k \otimes u_k$, $h_k(u_k) = 0$ and $i_k \neq i_{k+1}$ for all k. Let $l \in \{1,2\} \setminus \{i_n\}$ and note that, since G_l is non-trivial, there exists a non zero $x \in L^{\infty}(G_l)$ such that $h_l(x) = 0$. Then $uxu^* \in L^{\infty}(G_1) * L^{\infty}(G_2)$ is a reduced operator so $E_l(uxu^*) = 0$, where $E_l : L^{\infty}(G_1) * L^{\infty}(G_2) \to L^{\infty}(G_l)$ denotes the canonical Haar-state-preserving normal and faithful conditional expectation. However, since $\tau_t(x) = \tau_t^{G_l}(x) \in L^{\infty}(G_l)$ one has $\tau_t(x) = E_l(\tau_t(x)) = E_l(uxu^*) = 0$ hence x = 0, leading to a contradiction. It follows that such a u is always trivial and $T(G_1 * G_2) \subseteq T'$.

 $(3) \Rightarrow (2) \Rightarrow (1) \Rightarrow (4)$ are obvious. Also $(2) \Rightarrow (3)$ is easy and well known. Let us repeat however the argument here for the convenience of the reader. Let G be a CQG satisfying $|\operatorname{Irr}(G)| = 2$ and let u be the unique, up to unitary equivalence, non-trivial irreducible representation of G and write 1 for the trivial representation. Since u is non-trivial, \overline{u} also is hence $\overline{u} \simeq u$. It follows that $\dim(1, u \otimes u) = 1$. Hence, $u \otimes u = 1 \oplus du$, where

 $d = \dim(u, u \otimes u) \in \mathbb{N}$. Let us denote by $N \in \mathbb{N}^*$ the dimension of u so that we have $N^2 = 1 + dN$ hence $1 \equiv 0 \pmod{N}$ which implies that N = 1 and then d = 0. Since u is of dimension $1, u \in C(G)$ is a unitary such that $\Delta(u) = u \otimes u, u = u^*$ (hence $u^2 = 1$) and C(G) is generated by u so $G = \mathbb{Z}/2\mathbb{Z}$.

Suppose that (2) does not hold so that $\dim(L^{\infty}(G_1)) + \dim(L^{\infty}(G_2)) \geq 5$. It follows from [48, Theorem 4.1] that there exists a central projection z in $L^{\infty}(G_1 * G_2) = (L^{\infty}(G_1), h_1) * (L^{\infty}(G_2), h_2)$ such that $zL^{\infty}(G_1 * G_2)$ is either a full factor of type Π_1 or a full factor of type Π_1 , $\lambda \neq 0$, with T-invariant given by $\{t \in \mathbb{R} : \sigma_t^{h_1} = \sigma_t^{h_2}\}$ and $(1-z)L^{\infty}(G_1 * G_2)$ is a direct sum of matrix algebras. Hence, $zL^{\infty}(G_1 * G_2)$ is non-amenable (since it is full and not of type I) so $L^{\infty}(G_1 * G_2)$ is non-amenable either and it shows $(4) \Rightarrow (2)$. Moreover, we know from [57] that the set $\operatorname{Irr}(G_1 * G_2)$ is infinite hence, by Lemma 2.4, $L^{\infty}(G_1 * G_2)$ is diffuse so z = 1. Both the primeness and absence of Cartan follow now from [48, Corollary 4.3]. Finally, the Sd and τ invariants are computed in [49, Corollary 2.3] and [49, Theorem 3.2] respectively (recall that our von Neumann algebras are supposed to have separable preduals and that the Haar states on CQG are all almost periodic). \square

Remark 2.9. It is known that, for G a CQG of Kac type, the co-amenability of G is equivalent to the injectivity of the von Neumann algebra $L^{\infty}(G)$ (see Corollary 3.17 in [45] and the discussion after its proof). However, the equivalence for general CQG is open. Proposition 2.8 shows that in the class of CQG which are nontrivial free products the equivalence between co-amenability and injectivity of the von Neumann algebra is true.

Let us now recall the amalgamated free product construction [57,52]. Let G_1, G_2 be two CQG and $C(H) \subset C(G_k)$ a dual quantum subgroup of both G_1, G_2 . Let E_k : $C_r(G_k) \to C_r(H)$ be the faithful CE. The amalgameted free product is introduced in [57] and its Haar state and reduced C*-algebra is understood in [52]. Following [57], let us define $C(G) := C(G_1) *_{C(H)} C(G_2)$ the full amalgamated free product. By universal property there exists a unique unital *-homomorphism $\Delta: C(G) \to C(G) \otimes C(G)$ such that $\Delta|_{C(G_k)} = \Delta_{G_k}$ for k = 1, 2 and it is easy to check [57] that the pair $(C(G), \Delta)$ is a compact quantum group, denoted by $G = G_1 *_{H} G_2$. It is shown in [52] that the reduced C*-algebra is the reduced amalgamated free product with respect to the CE E_k , $C_r(G) = (C_r(G_1), E_1) \underset{C_r(H)}{*} (C_r(G_2), E_2)$ and the Haar state of G is the free product state $h_{G_1} * h_{G_2}$. To study further amalgamated free products, we will need the following lemma. Let us introduce before some terminology. A unitary representation u of a CQG G is called a Haar representation if, for all $k \in \mathbb{Z}^*$ one has $Mor(1, u^{\otimes k}) = \{0\}$, where 1 denotes the trivial representation and, for $k \geq 1$, we define $u^{\otimes -k} := \overline{u}^{\otimes k}$. Two unitary representation u_1, u_2 of G are called *free* if, for $l \geq 1$, any $(i_1, \ldots, i_l) \in \{1, 2\}^l$ such that $i_s \neq i_{s+1}$ for all s and any $k_1, \ldots, k_l \in \mathbb{Z}^*$, one has $\operatorname{Mor}(1, u_{i_1}^{\otimes k_1} \otimes u_{i_2}^{\otimes k_2} \otimes \cdots \otimes u_{i_l}^{\otimes k_l}) = \{0\}.$

Lemma 2.10. Let G be a compact quantum group with two Haar representations which are free then there exists $N \geq 1$ such that $L(\mathbb{F}_2) \subset M_N(\mathbb{C}) \otimes L^{\infty}(G)$ (with a state preserving inclusion). In particular, if G is Kac then $L^{\infty}(G)$ is not amenable.

Proof. Recall that, for $u \in \mathcal{B}(H) \otimes \mathrm{L}^{\infty}(G)$ a finite dimensional unitary representation, its contragredient unitary representation $\overline{u} \in \mathcal{B}(\overline{H}) \otimes \mathrm{L}^{\infty}(G)$ satisfies the following: there exists an invertible operator $Q \in \mathcal{B}(\overline{H})$ and an orthonormal basis $(e_i)_i$ of H such that, writing $u = \sum_{ij} e_{ij} \otimes u_{ij}$, where $(e_{ij})_{ij}$ are the matrix units associated to $(e_i)_i$, then $u^c := (Q \otimes 1)\overline{u}(Q^{-1} \otimes 1) = \sum_{ij} e'_{ij} \otimes u^*_{ij}$, where e'_{ij} are the matrix units associated to $(\overline{e}_i)_i$. Note also that $(u^c)^{\otimes k} := (u^c)_{1,k+1}(u^c)_{2,k+1}\dots(u^c)_{k,k+1} = (Q^{\otimes k} \otimes 1)\overline{u}^{\otimes k}((Q^{-1})^{\otimes k} \otimes 1) \in \mathcal{B}(\overline{H}^{\otimes k}) \otimes \mathcal{L}^{\infty}(G)$ for all $k \geq 1$. Hence, if $\mathrm{Mor}(1,\overline{u}^{\otimes k}) = \{0\}$ one has $(\mathrm{id} \otimes h)((u^c)^{\otimes k}) = Q^{\otimes k}(\mathrm{id} \otimes h)(\overline{u}^{\otimes k})(Q^{-1})^{\otimes k} = 0$, since $(\mathrm{id} \otimes h)(\overline{u}^{\otimes k}) \in \mathcal{B}(\overline{H}^{\otimes k})$ is the orthogonal projection onto $\mathrm{Mor}(1,\overline{u}^{\otimes k})$. It then follows that h(x) = 0 for any coefficient x of $(u^c)^{\otimes k}$. Since the coefficients of $(u^c)^{\otimes k}$ are exactly the products of k adjoints of coefficients of u, we deduce that h(x) = 0 for any product of k adjoints of coefficients of u whenever $\mathrm{Mor}(1,\overline{u}^{\otimes k}) = \{0\}$. We will use this remark in the rest of the proof.

Let $u_k \in \mathcal{B}(H_k) \otimes L^{\infty}(G)$, k = 1, 2, be two free Haar representations and define $v_1 := (u_1)_{13}$ and $v_2 := (u_2)_{23}$ which are both unitary in $\mathcal{B}(H_1 \otimes H_2) \otimes L^{\infty}(G)$ with respect to ω . Consider the faithful normal state $\omega = \operatorname{Tr} \otimes h \in (\mathcal{B}(H_1 \otimes H_2) \otimes L^{\infty}(G))_{*}$. Let us show that v_1 and v_2 are two free Haar unitaries. Since $(id \otimes h)(u_i^{\otimes k})$ (i = 1, 2) is the orthogonal projection onto $\operatorname{Mor}(1, u_i^{\otimes k}) = \{0\}$, for $k \in \mathbb{Z}^*$, it follows that h(x) = 0whenever x is a product of |k| coefficients of u_i or a product of |k| adjoints of coefficients of u_i , for all $k \in \mathbb{Z}^*$. Let \mathcal{C}_i be the linear span of products of coefficients of u_i and of products of adjoints of coefficients of u_i so that $\omega(\mathcal{C}_i) = \{0\}$. Since $v_1^k = \sum_{ij} e_{ij} \otimes 1 \otimes x_{ij}$ and $v_2^k = \sum_{i,j} 1 \otimes e_{ij} \otimes y_{ij}$, where $x_{ij} \in \mathcal{C}_1$, $y_{ij} \in \mathcal{C}_2$, for all $k \in \mathbb{Z}^*$, it follows that $\omega(v_i^k) = (\operatorname{Tr} \otimes \operatorname{id})(\operatorname{id} \otimes h)(v_i^k) = 0 \text{ for all } k \in \mathbb{Z}^*, i \in \{1, 2\}. \text{ Hence, both } v_1 \text{ and } v_2$ are Haar unitaries with respect to ω . Since u_1 and u_2 are free representations, the same argument as before shows that $h(\mathcal{C}) = \{0\}$, where \mathcal{C} is the linear span of operators $x \in L^{\infty}(G)$ of the form $x = y_1 \dots y_l, l \geq 1, y_s$ is a product of $|k_s|$ coefficients of u_{i_s} if $k_s \geq 1$ or adjoints of coefficients of u_{i_s} if $k_s \leq -1$ with $k_s \in \mathbb{Z}^*$ and $i_s \in \{1,2\}$ such that $i_s \neq i_{s+1}$ for all s. Let now $l \geq 1$ and $i_1, \ldots i_l \in \{1, 2\}$ with $i_s \neq i_{s+1}$ and $k_1, \ldots, k_l \in \mathbb{Z}^*$. We can write $v_{i_1}^{k_1} \ldots v_{i_l}^{k_l} = \sum_{i,j,k,l} e_{ij} \otimes e_{kl} \otimes x_{ijkl}$, where $x_{ijkl} \in \mathcal{C}$. It follows that $\omega(v_{i_1}^{k_1} \ldots v_{i_l}^{k_l}) = (\operatorname{Tr} \otimes \operatorname{id})(\operatorname{id} \otimes h)(v_{i_1}^{k_1} \ldots v_{i_l}^{k_l}) = 0$. Hence v_1 and v_2 are free with respect to ω . It follows that there exists a unique normal faithful *-homomorphism $L(\mathbb{F}_2) \to \mathcal{B}(H_1 \otimes H_2) \otimes L^{\infty}(G)$ which maps one generator of \mathbb{F}_2 onto v_1 and the other onto v_2 . Hence, if G is Kac, $\mathcal{B}(H_1 \otimes H_2) \otimes L^{\infty}(G)$ is non amenable either and it implies that $L^{\infty}(G)$ is also non amenable.

Definition 2.11. A dual quantum subgroup $C(H) \subset C(G)$ is called *proper* if there exists an irreducible representation a of G such that $(id \otimes E_H)(a) = 0$ and for any $s \in Irr(H)$, if $s \subset \overline{a} \otimes a$ then s = 1.

Note that if $C(H) \subset C(G)$ is proper then $[G:H] \geq 2$.

Remark 2.12. In the case of duals of discrete groups, the notion of proper dual quantum subgroup coincides with the usual notion of a proper subgroup. However, for quantum groups, there are examples of non proper dual quantum subgroup of index 2. A nice example is the dual quantum subgroup $\operatorname{Aut}^+(M_N(\mathbb{C}))$ of O_N^+ . Indeed, the representation category of O_N^+ is the category such that all the irreducible are self-adjoint, indexed by \mathbb{N} , $(u_n)_{n\in\mathbb{N}}$, with $u_0=\varepsilon$ and fusion rules $u_n\otimes u_m=u_{|n-m|}\oplus u_{|n-m|+2}\oplus \cdots \oplus u_{n+m}$. Taking the full subcategory of $\operatorname{Rep}(O_N^+)$ generated by the irreducible representations $(v_n=u_{2n})$, we get a category isomorphic to $\operatorname{Rep}(\operatorname{Aut}^+(M_N(\mathbb{C})))$, with irreducible representations indexed by \mathbb{N} and fusion rules $v_n\otimes v_m=u_{2n}\otimes u_{2m}=v_{|n-m|}\oplus v_{|n-m|+1}\oplus \cdots \oplus v_{n+m}$. With the fusion rules, we see that it is indeed a full subcategory and that the index of the corresponding subgroup is 2. However, taking any irreducible u_n of O_N^+ , with $n\geq 1$, we have that $v_1=u_2\subset \overline{u_n}\otimes u_n=u_n\otimes u_n$ so the dual quantum subgroup cannot be proper because $v_1\in\operatorname{Irr}(\operatorname{Aut}^+(M_N(\mathbb{C})))$.

Proposition 2.13. Let $G = G_1 *_H G_2$ be an amalgamated free product with H a proper dual quantum subgroup of G_1 and $[G_2 : H] \ge 3$ then there exists N such that $L(\mathbb{F}_2) \subset M_N(\mathbb{C}) \otimes L^{\infty}(G)$.

Proof. There exist an irreducible representation a of G_1 satisfying the conditions of Definition 2.11 and two irreducible representations b_1, b_2 of G_2 such that $(\mathrm{id} \otimes E_H)(b_i) = 0$ and $(\mathrm{id} \otimes E_H)(\overline{b_1} \otimes b_2) = 0$ (so we also have $(\mathrm{id} \otimes E_H)(\overline{b_2} \otimes b_1) = 0$). Define $u_i = a \otimes b_i \otimes \overline{a} \otimes \overline{b_i}$ and let $k \geq 1$. Since $\chi(a) \in \mathrm{L}^\infty(G_1)^\circ, \chi(b_i) \in \mathrm{L}^\infty(G_2)^\circ, \chi(u_i^{\otimes k}) = \chi(u_i)^k = (\chi(a)\chi(b_i)\chi(a)^*\chi(b_i)^*)^k$ is a reduced operator in the amalgamated free product so:

$$\dim \left(\operatorname{Mor}(1, u_i^{\otimes k})\right) = h(\chi(u_i^{\otimes k})) = 0.$$

Also, $\chi(\overline{u}_i^{\otimes k}) = (\chi(b_i)\chi(a)\chi(b_i)^*\chi(a)^*)^k$ is reduced so dim $(\operatorname{Mor}(1,\overline{u}_i^{\otimes k})) = h(\chi(\overline{u}_i^{\otimes k}))$ = 0. Hence, u_i is a Haar representation of G for $i \in \{1,2\}$. Let us show that u_1 and u_2 are free. It suffices to show that, for any $l \geq 1$, $(i_1,\ldots,i_l) \in \{1,2\}^l$ with $i_s \neq i_{s+1}$ and $k_1,\ldots,k_l \in \mathbb{Z}^*$ the operator $x:=\chi(u_{i_1}^{\otimes k_{i_1}}\otimes\cdots\otimes u_{i_l}^{\otimes k_l})$ is in the linear span of reduced operators. The case l=1 is clear by the first part of the proof and the general case can be shown by induction by using the same arguments used in the case l=2 that we present below. Let $x=\chi(u_{i_1}^{\otimes k_1}\otimes u_{i_2}^{\otimes k_2})$. It suffices to show that x is in the linear span of reduced operator. If k_1 and k_2 have the same sign then x is already reduced. If $k_1 \geq 1$ and $k_2 \leq -1$ then,

$$x = \chi(u_{i_1})^{k_1} \chi(\overline{u}_{i_2})^{-k_2}$$

= $\chi(u_{i_1})^{k_l-1} \chi(a) \chi(b_{i_1}) \chi(a)^* \chi(\overline{b}_{i_1} \otimes b_{i_2}) \chi(a) \chi(b_{i_2})^* \chi(a)^* \chi(\overline{u}_{i_2})^{-k_2-1}$

Since $(id \otimes E_H)(\overline{b}_{i_1} \otimes b_{i_2}) = 0$, we have $\chi(b_{i_1} \otimes \overline{b}_{i_2}) \in L^{\infty}(G_2)^{\circ}$ hence x is reduced. If $k_1 \leq -1$ and $k_2 \geq 1$ then

$$x = \chi(\overline{u}_{i_{1}})^{-k_{1}-1}\chi(b_{i_{1}})\chi(a)\chi(b_{i_{1}})^{*}\chi(\overline{a} \otimes a)\chi(b_{i_{2}})\chi(a)^{*}\chi(b_{i_{2}})^{*}\chi(u_{i_{2}})^{k_{2}-1}$$

$$= \chi(\overline{u}_{i_{1}})^{-k_{1}-1}\chi(b_{i_{1}})\chi(a)\chi(\overline{b}_{i_{1}} \otimes b_{i_{2}})\chi(a)^{*}\chi(b_{i_{2}})^{*}\chi(u_{i_{2}})^{k_{2}-1}$$

$$+ \sum_{s \in \operatorname{Irr}(G_{1})\backslash \operatorname{Irr}(H), s \subset \overline{a} \otimes a} \chi(\overline{u}_{i_{1}})^{-k_{1}-1}\chi(b_{i_{1}})\chi(a)\chi(b_{i_{1}})^{*}\chi(s)\chi(b_{i_{2}})\chi(a)^{*}\chi(b_{i_{2}})^{*}\chi(u_{i_{2}})^{k_{2}-1},$$

The right hand side of this equality is in the linear span of reduced operators since $\chi(\overline{b}_{i_1} \otimes b_{i_2}) \in L^{\infty}(G_2)^{\circ}$. \square

Remark 2.14. The index condition is clearly necessary since it is already necessary in the discrete group case. Indeed, if $\Gamma = \Gamma_1 * \Gamma_2$ be a non-trivial amalgamated free product of discrete groups (i.e. $\Sigma \neq \Gamma_k$, k = 1, 2). It is well known and easy to check that Γ is amenable if and only if Σ is amenable and $[\Gamma_k : \Sigma] = 2$ for all $k \in \{1, 2\}$ (actually Γ is an extension of Σ by $D_{\infty} = \mathbb{Z}_2 * \mathbb{Z}_2$).

Example 2.15. The dual quantum subgroup $C(\operatorname{Aut}^+(M_2(\mathbb{C}))) \subset C(O_2^+)$ is not proper, has index 2 and the quantum group $G := O_2^+ \underset{\operatorname{Aut}^+(M_2(\mathbb{C}))}{*} O_2^+$ is co-amenable. The inclusion of $C(\operatorname{Aut}^+(M_2(\mathbb{C})))$ in $C(O_2^+)$ is the map which sends the fundamental representation of $\operatorname{Aut}^+(M_2(\mathbb{C}))$ onto $v \otimes v$, where $v \in M_2(\mathbb{C}) \otimes C(O_2^+)$ is the fundamental representation of O_2^+ . Hence, writing v_l , $l \in \mathbb{N}$, the representatives of the irreducible representations of O_N^+ such that $v_0 = 1$ and $v_1 = v$, $C(\operatorname{Aut}^+(M_2(\mathbb{C})))$ is viewed in $C(O_2^+)$ has the C*-subalgebra generated by the coefficients of representations v_l for $l \in 2\mathbb{N}$. Let $\rho: C(O_2^+) \to C^*(\mathbb{Z}_2)$ be the unique unital *-homomorphism such that $(\operatorname{id} \otimes \rho)(v) = 1 \otimes g$, where g is the generator of \mathbb{Z}_2 . It is not difficult to check that ρ intertwines the comultiplications and,

$$(\mathrm{id} \otimes \rho)(v_l) = \begin{cases} 1 & \text{if } l \in 2\mathbb{N}, \\ g & \text{if } l \in 2\mathbb{N} + 1. \end{cases}$$

In particular, one has $\rho(x) = \varepsilon(x)1$ for all $x \in C(\operatorname{Aut}^+(M_2(\mathbb{C})))$.

It follows from the preceding discussion that, writing $v_{1,l}$ and $v_{2,l}$ the two copies of v_l in C(G), there exists a unique unital *-homomorphism $\pi: C(G) \to C^*(\mathbb{Z}_2 * \mathbb{Z}_2)$ such that $(\mathrm{id} \otimes \pi)(v_{i,1}) = 1 \otimes g_i$, for i = 1, 2 where g_1, g_2 are the two copies of g in $\mathbb{Z}_2 * \mathbb{Z}_2$. In particular π intertwines the comultiplications, $\pi(x) = \varepsilon(x)1$ for all $x \in C(\mathrm{Aut}^+(M_2(\mathbb{C})))$ and, whenever u is a representation of the form $u = v_{i_1,l_1} \otimes \cdots \otimes v_{i_n,l_n}$ with $i_s \neq i_{s+1}$ and $k_s \in 2\mathbb{N} + 1$ for all s one has $(\mathrm{id} \otimes \pi)(u) = 1 \otimes g_{i_1} \dots g_{i_n}$. Let us call such a representation a reduced representation and let \mathcal{C} be the linear span of coefficients of reduced representations. By the previous computation, for all $x \in \mathcal{C}$, $\pi(x)$ is a linear combination of reduced words in $\mathbb{Z}_2 * \mathbb{Z}_2$ hence, $\tau \circ \pi(\mathcal{C}) = \{0\}$, where τ is the canonical tracial state on $C^*(\mathbb{Z}_2 * \mathbb{Z}_2)$. Note also that, since any coefficient of a reduced representation u is a reduced word in the amalgamated free product C(G), one has $E(\mathcal{C}) = \{0\}$ where $E: C(G) \to C(\mathrm{Aut}^+(M_2(\mathbb{C})))$ is the canonical conditional expectation. Since C(G)

is the closed linear span of \mathcal{C} and $C(\operatorname{Aut}^+(M_2(\mathbb{C})))$ we deduce that $\tau \circ \pi = \varepsilon \circ E$. It follows that $\ker(\lambda_G) \subset \ker(\pi)$. Indeed, let $E_r : C_r(G) \to C_r(\operatorname{Aut}^+(M_2(\mathbb{C})))$ be the canonical faithful conditional expectation such that $E_r \circ \lambda_G = \lambda_{\operatorname{Aut}^+(M_2(\mathbb{C}))} \circ E$ and take $x \in \ker(\lambda_G)$. Then, $E_r(\lambda_G(x^*x)) = \lambda_{\operatorname{Aut}^+(M_2(\mathbb{C}))}(E(x^*x)) = 0$. Since $\operatorname{Aut}^+(M_2(\mathbb{C}))$ is co-amenable [1] it follows that $E(x^*x) = 0$ hence $\varepsilon(E(x^*x)) = \tau(\pi(x^*x)) = 0$. Since $\mathbb{Z}_2 * \mathbb{Z}_2$ is amenable, τ is faithful so $x \in \ker(\pi)$. Hence $\pi \prec \lambda_G$ and the co-amenability follows from [34, Theorem 3.11 and 3.12].

2.5. Semi-direct product quantum group

The semi-direct product quantum group is defined and studied in [59]. Let us recall below the basic facts about this construction.

Let G be a compact quantum group and Λ a finite group acting on C(G) by automorphisms of G, meaning that we have a group homomorphism $\alpha: \Lambda \to \operatorname{Aut}(C(G))$ such that $\Delta_G \circ \alpha_g = (\alpha_g \otimes \alpha_g) \circ \Delta_G$ for all $g \in \Lambda$. Define the C*-algebra $C(G \rtimes \Lambda) = C(G) \otimes C(\Lambda)$ with the comultiplication $\Delta: C(G \rtimes \Lambda) \to C(G \rtimes \Lambda) \otimes C(G \rtimes \Lambda)$ such that:

$$\Delta(a \otimes \delta_r) = \sum_{s \in \Lambda} \left[(\mathrm{id} \otimes \alpha_s)(\Delta_G(a)) \right]_{13} (1 \otimes \delta_s \otimes 1 \otimes \delta_{s^{-1}r}).$$

In particular, the inclusion $C(\Lambda) \subset C(G \rtimes \Lambda): x \mapsto 1 \otimes x$ preserves the comultiplications. It is shown in [59] that the pair $(C(G \rtimes \Lambda), \Delta)$ is a compact quantum group in its maximal version, the Haar measure h is given by $h = h_G \otimes \text{tr}$, where h_G is the Haar state on C(G) and tr is the integration with respect to the uniform probability on Λ i.e. $\text{tr}(\delta_r) = \frac{1}{|\Lambda|}$. Hence the reduced C*-algebra is $C_r(G) \otimes C(\Lambda)$, the von Neumann algebra is $L^{\infty}(G) \otimes C(\Lambda)$ and the modular group σ_t of $G \rtimes \Lambda$ is $\sigma_t = \sigma_t^G \otimes \text{id}$. Moreover, the canonical surjection $\Lambda: C(G \rtimes \Lambda) = C(G) \otimes C(\Lambda) \to C_r(G \rtimes \Lambda) = C_r(G) \otimes C(V)$ is $\lambda = \lambda_G \otimes \text{id}$, where λ_G is the canonical surjection $C(G) \to C_r(G)$. The irreducible representations and the fusion rules of $G \rtimes \Lambda$ are described in [59]. We could use the general classification of irreducible representations of $G \rtimes \Lambda$ from [59] to deduce the one-dimensional representations, we prefer to include a self-contained proof in the next Lemma.

Lemma 2.16. The one-dimensional unitary representations of $G \rtimes \Lambda$ are of the form $w \otimes v \in C(G \rtimes \Lambda)$, where $w \in C(G)$ and $v \in C(\Lambda)$ are one-dimensional unitary representations of G and Λ respectively and $\alpha_r(w) = w$ for all $r \in \Lambda$.

Proof. For this proof, we will view $C(G \times \Lambda) = C(\Lambda, C(G))$ and $C(G \times \Lambda) \otimes C(G \times \Lambda) = C(\Lambda \times \Lambda, C(G) \otimes C(G))$. With this identification, the comultiplication becomes $\Delta(u)(r,s) = (\mathrm{id} \otimes \alpha_r)(\Delta_G(u(rs)))$ for all $u \in C(G \times \Lambda)$ and all $r,s \in \Lambda$ and it is then easy to check that the unitary of the form given in the Lemma are indeed one-dimensional unitary representations of $G \times \Lambda$. Conversely, if $u \in C(G \times \Lambda)$ is a unitary

such that $\Delta(u) = u \otimes u$, then $u(r) \in C(G)$ is a unitary for all $r \in \Lambda$ and, for all $r, s \in \Lambda$, $(\mathrm{id} \otimes \alpha_r)(\Delta_G(u(rs)) = u(r) \otimes u(s)$. It follows that w := u(1) is a unitary in C(G) such that $\Delta(w) = w \otimes w$. Moreover, since α_r intertwines the comultiplication of G one has $\varepsilon_G \circ \alpha_r = \varepsilon_G$, where ε_G is the counit of G, and,

$$(\mathrm{id} \otimes \varepsilon_G \circ \alpha_r)(\Delta_G(u(rs))) = (\mathrm{id} \otimes \varepsilon_G)(\Delta_G(u(rs))) = u(rs) = u(r)\varepsilon_G(u(s)).$$

It follows that $v := (r \mapsto v_r)$, where $v_r := \varepsilon_G(u_r)$, is a one-dimensional unitary representation of Λ and, for all $s \in \Lambda$, $u(s) = wv_s$ i.e. $u = w \otimes v \in C(G) \otimes C(\Lambda) = C(G \rtimes \Lambda)$. Using that $\Delta(u) = u \otimes u$, one checks easily that $\alpha_r(w) = w$ for all $r \in \Lambda$. \square

We collect in the following Proposition some easy observations about $G \rtimes \Lambda$ that are not contained in [59]. We use the terminology introduced in the Introduction before the statement of Theorem A and we denote by $\Lambda_{cb}(G)$ the Cowling-Haagerup constant of the von Neumann algebra $L^{\infty}(G)$.

Proposition 2.17. The following holds.

- (1) $G \rtimes \Lambda$ is co-amenable if and only if G is co-amenable.
- (2) $\widehat{G} \times \widehat{\Lambda}$ has the Haagerup property if and only if \widehat{G} has the Haagerup property.
- (3) $\Lambda_{cb}(\widehat{G} \times \widehat{\Lambda}) = \Lambda_{cb}(\widehat{G}).$
- (4) The scaling group τ_t of $G \rtimes \Lambda$ is the one parameter group of $L^{\infty}(G \rtimes \Lambda) = L^{\infty}(G) \otimes C(\Lambda)$ defined by $\tau_t = \tau_t^G \otimes id$, where τ_t^G is the scaling group of G.
- (5) The Vaes' T-invariant $T(G \rtimes \Lambda)$ is:

$$\{t \in \mathbb{R} : \exists w \in \mathcal{U}(C(G)), \tau_t^G = \operatorname{Ad}(w), \Delta_G(w) = w \otimes w, \ \alpha_r(w) = w \ \forall r \in \Lambda\}.$$

(6)
$$Sd(G \rtimes \Lambda) = Sd(G)$$
 and $\tau(G \rtimes \Lambda) = \tau(G)$.

Proof. (1) directly follows from the fact that $\lambda = \lambda_G \otimes \operatorname{id}$ and (2) and (3) follows from $L^{\infty}(G \rtimes \Lambda) \simeq L^{\infty}(G) \otimes \mathbb{C}^K$, where $K = |\Lambda|$. To prove (4), one can easily check that the one parameter group $\tau_t := \tau_t^G \otimes \operatorname{id}$ satisfies $\Delta \circ \sigma_t = (\tau_t \otimes \sigma_t) \circ \Delta$. To prove (6), we note that since $h = h_G \otimes \operatorname{tr}$, the modular operator ∇ of h, is the positive operator on $L^2(G) \otimes l^2(\Lambda)$ given by $\nabla_G \otimes \operatorname{id}$, the equality $Sd(G \rtimes \Lambda) = Sd(G)$ follows, while the equality $\tau(G \rtimes \Lambda) = \tau(G)$ is a direct consequence of $\sigma_t = \sigma_t^G \otimes \operatorname{id}$. Finally, (5) is a consequence of (4) and Lemma 2.16. \square

2.6. Quantum permutation group

For $N \in \mathbb{N}^*$, we denote by S_N^+ the quantum permutation group on N points. We recall that $C(S_N^+)$ is the universal unital C*-algebra generated by N^2 orthogonal projections $u_{ij}, \ 1 \leq i, j \leq N$ with the relations $\sum_{j=1}^N u_{ij} = 1 = \sum_{j=1}^N u_{ji}$ for all $1 \leq i \leq N$. In particular $u = (u_{ij})_{ij} \in M_N(\mathbb{C}) \otimes C(S_N^+)$ is a unitary. The comultiplication on $C(S_N^+)$ is

defined, using the universal property of $C(S_N^+)$, by the relation $\Delta(u_{ij}) = \sum_{k=1}^N u_{ik} \otimes u_{kj}$ for all $1 \leq i, j \leq N$. In particular, u is a unitary representation of S_N^+ , called the fundamental representation. For $1 \leq i \leq N$ we write $L_i := \operatorname{Span}\{u_{ij} : 1 \leq j \leq N\} \subset \operatorname{Pol}(S_N^+)$. Since the family $(u_{ij})_{1 \leq j \leq N}$ is a partition of unity, the vector subspace L_i is actually a unital *-subalgebra of $\operatorname{Pol}(S_N^+)$ and the map $\mathbb{C}^N \to L_i$, $e_j \mapsto u_{ij}$ is a unital *-isomorphism of *-algebras. Since L_i is finite dimensional, we may view $L_i \subset C(S_N^+)$ or $L_i \subset C_r(S_N^+)$ as an abelian finite dimensional \mathbb{C}^* -subalgebra and also $L_i \subset \mathbb{L}^\infty(S_N^+)$ as an abelian finite dimensional von Neumann subalgebra. We use the same symbol h to denote the Haar state of S_N^+ on $C(S_N^+)$, $C_r(S_N^+)$ or $\mathbb{L}^\infty(S_N^+)$. We also recall that $h(u_{ij}) = \frac{1}{N}$ for all $1 \leq i, j \leq N$, where h is the Haar state on $C(S_N^+)$.

The elementary proof of the next Proposition is left to the reader.

Proposition 2.18. Let $1 \le i \le N$. The following holds

(1) The unique trace preserving conditional expectation $E_i: L^{\infty}(S_N^+) \to L_i$ satisfies:

$$E_i(x) = N \sum_{j=1}^{N} h(xu_{ij})u_{ij}$$
 for all $x \in L^{\infty}(S_N^+)$.

- (2) The map $x \mapsto N \sum_{j=1}^{N} h(xu_{ij})u_{ij}$ is a conditional expectation from $C(S_N^+)$ (resp. $C_r(S_N^+)$) onto L_i . All these maps will be denoted by E_i .
- (3) The conditional expectation $E_i: C_r(S_N^+) \to L_i$ is faithful.

We collect below some elementary computations concerning the conditional expectation onto L_i .

Lemma 2.19. Let $a \in C(S_N^+)$ and $1 \le i \le N$ be such that $E_i(a) = 0$. Then,

- (1) For any $1 \leq s, j \leq N$, we have $(h \otimes id)(\Delta(a)(u_{is} \otimes u_{sj})) = 0$.
- (2) For all $1 \leq s \leq N$, we have $(h \otimes id) (\Delta(a)(u_{is} \otimes 1)) = 0$.

Proof. (1). We show that $\forall \omega \in C(S_N^+)^*$, $1 \leq s, j \leq N$, $(h \otimes \omega)(\Delta(a)(u_{is} \otimes u_{sj})) = 0$. Let ω, s, j be as above, and define $\mu = \omega(\cdot u_{sj}) \in C(S_N^+)^*$. Using the Sweedler notation,

$$\Delta(au_{ij}) (1 \otimes u_{sj}) = \sum_{t=1}^{N} a_{(1)} u_{it} \otimes a_{(2)} u_{tj} u_{sj} = a_{(1)} u_{is} \otimes a_{(2)} u_{sj} = \Delta(a) (u_{is} \otimes u_{sj}).$$

Applying $(h \otimes \omega)$, we get

$$(h \otimes \omega) (\Delta(a)(u_{is} \otimes u_{sj})) = (h \otimes \omega) (\Delta(au_{ij})(1 \otimes u_{sj})) = (h \otimes \mu)(\Delta(au_{ij}))$$
$$= \mu(1)h(au_{ij}) = 0,$$

where we used the invariance of the Haar state h and the fact that $h(au_{ij}) = 0$ for all $1 \le j \le N$ since $E_i(a) = 0$ and by definition of E_i .

(2). It suffices to sum the relation (1) for $1 \le j \le N$ and use $\sum_{j=1}^{N} u_{sj} = 1 \ \forall s$. \square

We will use the following Lemma, which is an easy consequence of the factoriality of $L^{\infty}(S_N^+)$ when $N \geq 8$ [9], but we will need the next result for all $N \geq 4$.

Lemma 2.20. For all $N \geq 4$ and all $1 \leq k \leq N$ one has $L^{\infty}(S_N^+)' \cap L_k = \mathbb{C}1$.

Proof. Pol (S_N^+) being weakly dense in $L^{\infty}(S_N^+)$ one has $L^{\infty}(S_N^+)' \cap L_k = \text{Pol}(S_N^+)' \cap L_k$. By the universal property of $\text{Pol}(S_N^+)$, for all $\sigma \in S_N$, there exist unique unital *-homomorphisms $R_{\sigma}, C_{\sigma} : \text{Pol}(S_N^+) \to \text{Pol}(S_N^+)$ such that $R_{\sigma}(u_{ij}) = u_{\sigma(i)j}$ and $C_{\sigma}(u_{ij}) = u_{i\sigma(j)}$, for all $1 \leq i, j \leq N$. Note that R_{σ} and C_{σ} are *-isomorphisms since $R_{\sigma^{-1}}R_{\sigma} = R_{\sigma}R_{\sigma^{-1}} = \text{id}$ and $C_{\sigma^{-1}}C_{\sigma} = C_{\sigma}C_{\sigma^{-1}} = \text{id}$. Denoting by $(1, k) \in S_N$ the transposition of 1 and k, one has $R_{(1,k)}(L_k) = L_1$ hence, it suffices to show that $\text{Pol}(S_N^+)' \cap L_1 = \mathbb{C}1$.

Fix a Hilbert space H with two non-commuting orthogonal projections $P, Q \in \mathcal{B}(H)$ and let us denote by $A \in M_4(\mathcal{B}(H))$ the matrix

$$A := \begin{pmatrix} P & 1-P & 0 & 0 \\ 1-P & P & 0 & 0 \\ 0 & 0 & Q & 1-Q \\ 0 & 0 & 1-Q & Q \end{pmatrix},$$

and by $B \in M_N(\mathcal{B}(H))$ the block matrix $B = \begin{pmatrix} A & 0 \\ 0 & I \end{pmatrix}$, where I is the identity matrix.

Note that B is a magic unitary. Hence, writing $B = (b_{ij})$, there exists a unique unital *-homomorphism $\pi : \operatorname{Pol}(S_N^+) \to \mathcal{B}(H)$ such that $\pi(u_{ij}) = b_{ij}$ for all $1 \leq i, j \leq N$.

To show that $\operatorname{Pol}(S_N^+)' \cap L_1 = \mathbb{C}1$, it suffices to show that the only orthogonal projections in L_1 that commutes with $\operatorname{Pol}(S_N^+)$ are 0 and 1. A projection $p \in L_1 \setminus \{0, 1\}$ is of the form $p = \sum_{j \in I} u_{1j}$ for $I \subset \{1, \dots, N\}$ with $1 \leq |I| \leq N-1$. Let $\sigma \in S_N$ be a permutation such that $\sigma(I) = \{1, \dots, |I|+1\} \setminus \{2\}$ so that $q := C_{\sigma}(p) = u_{11} + u_{13} + u_{14} + \dots + u_{1,|I|+1}$. It suffices to show that q does not commute with u_{33} and this follows from $\pi(q) = P$ and $\pi(u_{33}) = Q$. \square

2.7. Free wreath products

For a compact quantum group G and an integer N, the free wreath product of G by S_N^+ , as defined by Bichon in [8], is the CQG $G \wr_* S_N^+$ with

$$C(G \wr_* S_N^+) = C(G)^{*N} * C(S_N^+)/I,$$

where we consider the full free product and I is the two-sided closed ideal generated by:

$$\{\nu_i(a)u_{ij} - u_{ij}\nu_i(a) : a \in C(G), 1 \le i, j \le N\}$$
 (1)

and $\nu_i: C(G) \to C(G)^{*N} \subset C(G)^{*N} * C(S_N^+)$ is the unital *-homomorphism on the i^{th} -copy of C(G) in $C(G)^{*N}$, $u_{ij} \in C(S_N^+)$ are the coefficients of the fundamental representation. If G has a dual quantum subgroup H i.e. $C(H) \subset C(G)$, then we define, following [28], the free wreath product with amalgamation $G \wr_{*,H} S_N^+$. This is the CQG with $C(G \wr_{*,H} S_N^+) = C(G)^{*H} \times C(S_N^+)/I$, where the full free product is taken amalgamated over C(H) and the ideal I is the same as in (1). It is easy to check (using both universal properties) that $C(G \wr_{*,H} S_N^+) = C(G \wr_{*,H} S_N^+)/J$, where J is the closed two-sided ideal generated by $\{\nu_i(a) - \nu_j(a) : a \in C(H), 1 \leq i, j \leq N\}$. Note that G always admits the trivial group $\{e\}$ as a dual quantum subgroup, and we have, for $H = \{e\}$, $C(G \wr_{*,H} S_N^+) \simeq C(G \wr_{*} S_N^+)$.

Remark 2.21. The surjective *-homomorphism $a\mapsto a+I: C(G)^{*N}*C(S_N^+)\to C(G\wr_*S_N^+)$ is injective when restricted to $C(G)^{*N}$ as well as to $C(S_N^+)$. By the universal property, there exists a unique unital *-homomorphism $\pi: C(G\wr_*S_N^+)\to C(G)^{*N}\otimes C(S_N^+)$ such that $\pi(x+I)=x\otimes 1$ if $x\in C(G)^{*N}$ and $\pi(x+I)=1\otimes x$ if $x\in C(S_N^+)$. The composition of $x\mapsto x+I$ and π is the map sending $C(G)^{*N}$ and $C(S_N^+)$ to their respective copies in $C(G)^{*N}\otimes C(S_N^+)$, which is injective. The same holds for $C(G)^{*H}$ and $C(S_N^+)$ in the amalgamated case.

Following the previous Remark, we will always view $C(G)^{*_H N}$, $C(S_N^+) \subset C(G\wr_{*,H}S_N^+)$. We endow the unital C^* -algebra $C(G\wr_{*,H}S_N^+)$ with the unique unital *-homomorphism $\Delta: C(G\wr_{*,H}S_N^+) \to C(G\wr_{*,H}S_N^+) \otimes C(G\wr_{*,H}S_N^+)$ satisfying:

$$\Delta(\nu_i(a)) = \sum_{j=1}^N (\nu_i \otimes \nu_j)(\Delta_G(a))(u_{ij} \otimes 1) \text{ and } \Delta(u_{ij}) = \sum_{k=1}^N u_{ik} \otimes u_{kj}.$$

Remark 2.22.

- (1) Both G^{*_HN} and S_N^+ are compact quantum subgroups of $G \wr_{*,H} S_N^+$ via the maps $(\mathrm{id} \otimes \varepsilon_{S_N^+}) \circ \pi : C(G \wr_{*,H} S_N^+) \to C(G)^{*_{H}N}$ and $(\varepsilon_{G^{*_N}} \otimes \mathrm{id}) \circ \pi : C(G \wr_{*,H} S_N^+) \to C(S_N^+)$ (which obviously intertwine the comultiplications), where π is the map defined in Remark 2.21.
- (2) Let $\nu: C(H) \to C(G \wr_{*,H} S_N^+)$ be the common restriction of the maps ν_i on $C(H) \subset C(G)$, $1 \le i \le N$. Then, ν is faithful and, since $\sum_j u_{ij} = 1$ we see that ν intertwines the comultiplications. Hence, H is a dual compact subgroup of $G \wr_{*,H} S_N^+$.

Let us now describe another specific compact quantum subgroup of $G \wr_{*,H} S_N^+$. The only mentions of this subgroup that we could find in the literature are in [8, Proposition 2.6], when N=2 and $G=\widehat{\Gamma}$ is the dual of a discrete group, and in Gromada's recent work [30] where it appears under the name *wreath product*, in the non-amalgamated case. We would like to thank the referee for showing us this reference.

Fix $\sigma \in S_N$. By the universal property of $C(G)^{*_HN}$, there exists a unique unital *homomorphism $\alpha_{\sigma}: C(G)^{*_HN} \to C(G)^{*_HN}$ such that $\alpha_{\sigma} \circ \nu_i = \nu_{\sigma(i)}$ for all $1 \leq i \leq N$. Since we clearly have $\alpha_{\sigma}\alpha_{\tau} = \alpha_{\sigma\tau}$, for all $\sigma, \tau \in S_N$, σ is an action of S_N on the C*-algebra $C(G)^{*_HN}$ by unital *-isomorphisms. Moreover, it is easy to see that, for all $\sigma \in \Sigma$, $(\alpha_{\sigma} \otimes \alpha_{\sigma}) \circ \Delta_{G^{*_N}} = \Delta_{G^{*_N}} \circ \alpha_{\sigma}$, where $\Delta_{G^{*_N}}$ is the comultiplication on $C(G)^{*_HN}$. Note that this action by automorphisms of G^{*_HN} is actually the restriction of the action of S_N^+ on the compact quantum group G^{*_HN} to the compact quantum subgroup S_N which was described in [8, Proposition 2.1]. Let us now consider the semi-direct product quantum group $G^{*_HN} \rtimes S_N$ associated to this action, as described in Section 2.5. We show below that the quantum group $G^{*_HN} \rtimes S_N$ is actually a compact quantum subgroup of $G \wr_{*_H} S_N^+$.

Proposition 2.23. Given $N \in \mathbb{N}^*$, there exists a unique unital *-homomorphism

$$\pi : C(G \wr_{*,H} S_N^+) \to C(G^{*_H N} \rtimes S_N) = C(G)^{*_H N} \otimes C(S_N) \text{ s.t. } \begin{cases} \pi(\nu_i(a)) = \nu_i(a) \otimes 1 \\ \pi(u_{ij}) = 1 \otimes \chi_{ij} \end{cases}$$

where $\chi_{ij} \in C(S_N)$ is the characteristic function of $A_{i,j} := \{ \sigma \in S_N : \sigma(j) = i \}$. Moreover,

- (1) π is surjective and intertwines the comultiplications.
- (2) π is an isomorphism if and only if $N \in \{1,2\}$ or if N = 3 and the inclusion $C(H) \hookrightarrow C(G)$ is an isomorphism (a special case being when G is the trivial group).

Proof. The existence of π is a direct consequence of the universal property of the C*-algebra $C(G \wr_{*,H} S_N^+)$ and the fact that the matrix $(\chi_{ij})_{ij} \in M_N(C(S_N))$ is a magic unitary.

(1). Since $C(S_N)$ is generated by the χ_{ij} , the surjectivity of π is clear. Let us check that π intertwines the comultiplications. Recall that the comultiplication on $G\wr_{*,H}S_N^+$ is denoted by Δ , the one on G by Δ_G , the one on G^{*_HN} by $\Delta_{G^{*_N}}$ (it satisfies $\Delta_{G^{*_N}}\circ\nu_i=\nu_i\otimes\nu_i\circ\Delta_G$) and let us denote the one on $C(G^{*_HN}\rtimes S_N)$ by Δ_s . On the one hand, $\forall 1\leq i,j\leq N$, $a\in C(G)$,

$$\Delta_{s}(\pi(u_{ij})) = \Delta_{s}(1 \otimes \chi_{ij}) = \sum_{\sigma \in S_{N}, \, \sigma(j) = i} \Delta_{s}(1 \otimes \delta_{\sigma})$$

$$= \sum_{\sigma \in S_{N}, \, \sigma(j) = i} \sum_{\tau \in S_{N}} (1 \otimes \delta_{\tau} \otimes 1 \otimes \delta_{\tau^{-1}\sigma}),$$

$$\Delta_{s}(\pi(\nu_{i}(a))) = \sum_{\sigma \in S_{N}} \Delta_{s}(\nu_{i}(a) \otimes \delta_{\sigma})$$

$$= \sum_{\tau, \sigma \in S_{N}} [(\mathrm{id} \otimes \alpha_{\tau})(\Delta_{G^{*N}}(\nu_{i}(a)))]_{13} (1 \otimes \delta_{\tau} \otimes 1 \otimes \delta_{\tau^{-1}\sigma})$$

$$= \sum_{\tau \in S_N} \left[(\mathrm{id} \otimes \alpha_\tau) ((\nu_i \otimes \nu_i)(\Delta_G(a))) \right]_{13} (1 \otimes \delta_\tau \otimes 1 \otimes 1).$$

On the other hand, for all $1 \le i, j \le N$ and all $a \in C(G)$,

$$(\pi \otimes \pi)(\Delta(\nu_{i}(a))) = \sum_{j=1}^{N} (\pi \otimes \pi)(\nu_{i} \otimes \nu_{j})(\Delta_{G}(a))(\pi(u_{ij}) \otimes 1)$$

$$= \sum_{j=1}^{N} [(\nu_{i} \otimes \nu_{j})(\Delta_{G}(a))]_{13} (1 \otimes \chi_{ij} \otimes 1 \otimes 1)$$

$$= \sum_{j=1}^{N} \sum_{\tau \in S_{N}, \tau(j)=i} [(\mathrm{id} \otimes \alpha_{\tau})(\nu_{i} \otimes \nu_{i})(\Delta_{G}(a))]_{13} (1 \otimes \delta_{\tau} \otimes 1 \otimes 1)$$

$$= \sum_{\tau \in S_{N}} [(\mathrm{id} \otimes \alpha_{\tau})(\nu_{i} \otimes \nu_{i})(\Delta_{G}(a))]_{13} (1 \otimes \delta_{\tau} \otimes 1 \otimes 1)$$

where, in the last equality, we use the partition $S_N = \bigsqcup_{j=1}^N \{\tau \in S_N : \tau(j) = i\}$. Moreover,

$$(\pi \otimes \pi)(\Delta(u_{ij})) = \sum_{k=1}^{N} \pi(u_{ik}) \otimes \pi(u_{kj}) = \sum_{k=1}^{N} 1 \otimes \chi_{ik} \otimes 1 \otimes \chi_{kj}$$
$$= \sum_{k=1}^{N} \sum_{\substack{\sigma, \tau \in S_N, \\ \tau(k) = i, \ \sigma(j) = k}} 1 \otimes \delta_{\tau} \otimes 1 \otimes \delta_{\sigma} = \sum_{\substack{\sigma, \tau \in S_N, \\ \sigma(j) = i}} 1 \otimes \delta_{\tau} \otimes 1 \otimes \delta_{\tau^{-1}\sigma},$$

where we use, in the last equality, the following partition:

$$\{(\tau, \tau^{-1}\sigma) \in S_N^2 : \sigma(j) = i\} = \bigsqcup_{k=1}^N \{(\tau, \sigma) \in S_N^2 : \tau(k) = i \text{ and } \sigma(j) = k\}.$$

(2). Suppose that N=2. In that case, Bichon's proof of [8, Proposition 2.1] can be directly adapted. It is well known that $C(S_2^+)$ is commutative: the surjection $C(S_2^+) \to C(S_2)$, $u_{ij} \mapsto \chi_{ij}$ is an isomorphism. Hence, we view $C(S_2^+) = C(S_2)$ and we have $u = \begin{pmatrix} \delta_1 & \delta_\tau \\ \delta_\tau & \delta_1 \end{pmatrix}$, where τ is the unique non-trivial element in S_2 . Now, for $i \in \{1,2\}$ and $a \in C(G)$, $\nu_i(a) \in C(G \wr_{*,H} S_N^+)$ is commuting with the line i of u hence, it is also commuting with the other line. Hence, $C(G)^{*_H 2}$ and $C(S_2^+)$ are commuting in $C(G \wr_{*_H} S_N^+)$ and it follows that π is actually injective.

Suppose that $N \geq 4$. In that case, it is well known that $C(S_N^+)$ is infinite-dimensional (the classical argument is actually contained in the proof of Lemma 2.19). In particular, $C(S_N^+) \mapsto C(S_N)$, $u_{ij} \mapsto \chi_{ij}$ is not injective which implies that π itself is not injective.

The non injectivity in the case N=3 is a consequence of Theorem 3.2 (the proof of which does not rely on this isomorphism even in the case N=3), using the fact that an isomorphism between these quantum groups would intertwine the Haar measures. It is then enough, when G is non-trivial to show that there is a reduced operator in $C(G \wr_{*,H} S_3^+)$, hence of Haar measure 0, which is sent to an element of $C(G^{*_H 3} \rtimes S_3) \simeq$ $C(G)^{*_{H}3} \otimes \mathbb{C}^{6}$. Since H is a strict (not necessarily proper) dual quantum subgroup of G, then Rep(H) is a full subcategory of Rep(G), with a non trivial irreducible representation v of G which is not a representation of H, and $a, b \in C(G) \setminus C(H)$ non trivial coefficients of v and \overline{v} respectively such that ab has non zero Haar measure (using the unique intertwiner between the trivial representation and $v \otimes \overline{v}$). Then we can consider the element $\nu_1(a)u_{22}\nu_1(b) \in C(G_{\ell_*,H}S_3^+)$, which is sent to $\nu_1(a)\nu_1(b) \otimes u_{22} = \nu_1(ab) \otimes u_{22} \in$ $C(G)^{*_{H}3} \otimes \mathbb{C}^{6}$. However, the word $\nu_{1}(a)u_{22}\nu_{1}(b)$ is reduced in the sense of 3.2, because a and b are coefficients of a non trivial irreducible representation, and therefore of Haar measure 0 and the element u_{22} is such that $E_{L_1}(u_{22})$ is equal to 0 if $N \geq 3$ and to u_{22} if $N \in \{0,1\}$, while the element $\nu_1(ab) \otimes u_{22}$ is of Haar measure $h_G(ab)/3 \neq 0$, because the Haar measure on $C(G^{*H^3} \times S_3) \simeq C(G)^{*H^3} \otimes \mathbb{C}^6$ is the tensor product of the Haar measures on $C(G)^{*_H 3}$ and on $C(S_3)$.

If G is the trivial group then the surjection is the canonical morphism $C(S_N^+) \twoheadrightarrow C(S_N)$ which is an isomorphism if and only if $N \leq 3$. \square

Remark 2.24. The condition $C(H) \hookrightarrow C(G)$ not being an isomorphism is equivalent to the index [G:H] being greater than 2, as defined in 2.7.

Let us show that this semi-direct product agrees with the wreath product by S_N as defined in [30]. We first start by defining a well-known family of compact quantum subgroups of $G \wr_{*,H} S_N^+$.

Proposition-Definition 2.25. For any compact quantum subgroup $K_N \leq S_N^+$ with $\mu: C(S_N^+) \to C(K_N)$, we have a corresponding compact quantum subgroup $G \wr_{*,H} K_N \leq G \wr_{*,H} S_N^+$, with $C(G \wr_{*,H} K_N) := C(G)^{*_H N} * C(K_N)/I$, where the double-sided closed ideal I is once again defined as in (1), replacing the elements u_{ij} by their images through μ in $C(K_N)$, with the comultiplication defined in the same way as for the free wreath product with S_N^+ . In particular $G \wr_{*,H} S_N$ is a compact quantum subgroup of $G \wr_{*,H} S_N^+$, where S_N is seen as a compact quantum group of S_N^+ .

Proof. It is enough to show that the quotient map

$$\psi: C(G \wr_{*,H} S_N^+) \to C(G \wr_{*,H} K_N) \text{ s.t. } \begin{cases} \psi(\nu_i(a)) = \nu_i(a) \\ \psi(u_{ij}) = \mu(u_{ij}) \end{cases}$$

intertwines the comultiplications but this is obvious by the definitions. \Box

Remark 2.26. Note that the considered dimension of the generating matrix of the compact quantum group K_N matters a lot, as it changes the numbers of free copies of C(G) in the definition of the free wreath product. Take H to be the trivial group and K_N and K_M for $N \neq M$ two presentations of the trivial group (seen as acting on \mathbb{C}^N and \mathbb{C}^M respectively), then we have that $G \wr_{*,H} K_N = G^{*N}$ and $G \wr_{*,H} K_M = G^{*M}$, which are oftentimes not isomorphic.

Proposition 2.27. The wreath product $G \wr S_N$ from [30] is isomorphic to the semi-direct product $G^{*N} \rtimes S_N$.

Proof. It is enough to prove that the map $\pi: C(G \wr_* S_N^+) \twoheadrightarrow C(G^{*N} \rtimes S_N) \simeq C(G)^{*N} \otimes C(S_N)$ defined in Proposition 2.23 is the composition of the maps $C(G \wr_* S_N^+) \twoheadrightarrow C(G \wr_* S_N) \otimes C(G \wr$

Remark 2.28. For many quantum groups G we can also use K-theory to distinguish between the free wreath product and the semi-direct product. Indeed, the K-theory of the algebra of the semi-direct product $G^{*3} \rtimes S_3$ is just the K-theory of the tensor product $C(G)^{*3} \otimes \mathbb{C}^6$ which is

$$K_0(C(G)^{*3} \otimes \mathbb{C}^6) \simeq (K_0(C(G))^3/\mathbb{Z}^2)^3$$
, and, $K_1(C(G)^{*3} \otimes \mathbb{C}^6) \simeq K_1(C(G))^{18}$,

which can only coincide with the K-theory of the free wreath product algebra if $K_0(C(G)) \simeq \mathbb{Z}$ and $K_1(C(G)) = 0$, using the computations of Theorem D which are independent from Proposition 2.23

In [2, Section 3.4] it is mentioned that the question whether the von Neumann algebra of a quantum reflection group $H_N^{s+} := \mathbb{Z}_s \wr_* S_N^+$ is diffuse is open when $N \leq 7$. We add the following Proposition in order to provide a complete answer to this question.

Proposition 2.29. The von Neumann algebra $L^{\infty}(G \wr_{*,H} S_N^+)$ is diffuse if and only if at least one of the following conditions holds:

- $N \geq 4$,
- Irr(G) is infinite,
- $[G:H] \ge 2$ (i.e. $C(H) \hookrightarrow C(G)$ is not surjective) and $N \ge 2$.

Proof. If G is trivial, it directly follows from Lemma 2.4 since $S_N^+ \simeq S_N$ for $N \leq 3$ and $C(S_N^+)$ is infinite-dimensional for $N \geq 4$. Suppose that G is non-trivial and let $N \geq 3$. Let us denote by $u = (u_{ij})$ and $v = (v_{ij})$ the fundamental representations of S_N^+ and S_{N-1}^+ respectively. By the universal property, there exists a unique unital *-homomorphism

 $\rho: C(G\wr_{*,H}S_N^+) \to C(G\wr_{*,H}S_{N-1}^+)$ such that $\rho(u_{ij}) = v_{ij}$ if $1 \le i, j \le N-1$, $\rho(u_{ij}) = \delta_{i,j}$ if i or j is equal to N and $\rho(\nu_i(a)) = \nu_i(a)$ if $1 \le i \le N-1$, $\rho(\nu_N(a)) = \nu_{N-1}(a)$ for all $a \in C(G)$. Since ρ is clearly surjective a direct induction implies that, for all $N \ge 2$, $C(G\wr_{*,H}S_N^+)$ is infinite-dimensional (hence $L^\infty(G\wr_{*,H}S_N^+)$ is diffuse by Lemma 2.4) whenever $C(G\wr_{*,H}S_2^+)$ is. By Proposition 2.23, $C(G\wr_{*,H}S_2^+) \simeq C(G*_HG) \otimes \mathbb{C}^2$. Hence, it is infinite-dimensional as soon as $C(G*_HG)$ is. There is an embedding $C(G) \hookrightarrow C(G*_HG)$, so the case $|\operatorname{Irr}(G)| = \infty$ is a direct consequence of Lemma 2.4. If $C(H) \hookrightarrow C(G)$ is not surjective, then there is a non zero element $a \in C(G)$ such that $E_H(G) = 0$ (for example, take a coefficient of an irreducible representation of G which is not in the representation category of G. Then the words G0 is even, form a family of linearly independent reduced elements in $G(G*_HG)$, which is therefore infinite dimensional. \Box

2.8. Graphs of operator algebras

We recall below some notions and results from [22,23]. If \mathcal{G} is a graph in the sense of [44, Def 2.1], its vertex set will be denoted $V(\mathcal{G})$ and its edge set will be denoted $E(\mathcal{G})$. We will always assume that \mathcal{G} is at most countable. For $e \in E(\mathcal{G})$ we denote by s(e) and r(e) respectively the source and range of e and by \overline{e} the inverse edge of e. An orientation of \mathcal{G} is a partition $E(\mathcal{G}) = E^+(\mathcal{G}) \sqcup E^-(\mathcal{G})$ such that $e \in E^+(\mathcal{G}) \Leftrightarrow \overline{e} \in E^-(\mathcal{G})$.

The data $(\mathcal{G}, (A_q)_{q \in V(\mathcal{G})}, (B_e)_{e \in E(\mathcal{G})}, (s_e)_{e \in E(\mathcal{G})})$ will be called a graph of C*-algebras if:

- \mathcal{G} is a connected graph.
- For every $q \in V(\mathcal{G})$ and every $e \in E(\mathcal{G})$, A_q and B_e are unital C*-algebras.
- For every $e \in E(\mathcal{G})$, $B_{\overline{e}} = B_e$.
- For every $e \in E(\mathcal{G})$, $s_e : B_e \to A_{s(e)}$ is a unital faithful *-homomorphism.

Define $r_e = s_{\overline{e}} : B_e \to A_{s(e)}$.

Fix a maximal subtree $\mathcal{T} \subset \mathcal{G}$ and define $P = \pi_1(\mathcal{G}, A_q, B_e, \mathcal{T})$ to be the maximal fundamental C*-algebra of the graph of C*-algebras $(\mathcal{G}, A_q, B_e, s_e)$ with respect to the maximal subtree \mathcal{T} . This means that P is the universal unital C*-algebra generated by A_q , for $q \in V(\mathcal{G})$ and unitaries u_e , for $e \in E(\mathcal{G})$ with the following relations:

- For every $e \in E(\mathcal{G})$, $u_{\overline{e}} = u_e^*$.
- For every $e \in E(\mathcal{G})$ and every $b \in B_e$, $u_{\overline{e}}s_e(b)u_e = r_e(b)$.
- For every $e \in E(\mathcal{T})$, $u_e = 1$.

It is known that P is non-zero and that the canonical unital *-homomorphisms $A_q \to P$ are all faithful [23, Remark 2.2]. Hence, we will always view $A_q \subset P$ for all $q \in V(\mathcal{G})$.

Assume now that there exists, for all $e \in E(\mathcal{G})$ a conditional expectation $E_e^s : A_{s(e)} \to s_e(B_e)$. For $p \in V(\mathcal{G})$, an element $a \in P$ will be called a reduced operator from p to p

if it is of the form $a=a_0u_{e_1}a_1\dots u_{e_n}a_n$ where $n\geq 1,\ (e_1,\dots,e_n)$ is a path in $\mathcal G$ from p to p (i.e. $e_k\in E(\mathcal G)$ are such that $r(e_k)=s(e_{k+1})$ and $s(e_1)=r(e_n)=p),\ a_0\in A_p,$ $a_k\in A_{r(e_k)}$ and, for all $1\leq k\leq n-1$, if $e_{k+1}=\overline{e}_k$ then $E^s_{e_{k+1}}(a_k)=0$. Then one can construct [23], the unital C*-algebra P_r called the vertex reduced fundamental algebra which is the unique (up to canonical isomorphism) quotient $\lambda:P\to P_r$ of P satisfying the following:

- (1) There exists, for all $p \in V(\mathcal{G})$, a ucp map $E_p : P_r \to A_p$ such that $E_p(\lambda(a)) = a$ for all $a \in A_p$ and $E_p(\lambda(a)) = 0$ for all reduced operators $a \in P$ from p to p. Moreover, the family $\{E_p : \in V(\mathcal{G})\}$ is GNS-faithful.
- (2) For any unital C*-algebra C with a surjective unital *-homomorphism $\rho: P \to C$ and a GNS-faithful family of ucp map $\varphi_p: C \to A_p, p \in V(\mathcal{G})$, such that $\varphi_p(\rho(a)) = a$ for all $a \in A_p$ and $\varphi_p(\rho(a)) = 0$ for all reduced operators $a \in P$ from p to p there exists a unique unital *-isomorphism $\nu: P_r \to C$ such that $\nu \circ \lambda = \rho$.

We recall that, given unital C*-algebras $A, B_i, i \in I$, a family of ucp maps $\varphi_i : A \to B_i$ is called GNS-faithful if $\bigcap_{i \in I} \operatorname{Ker}(\pi_i) = \{0\}$, where (H_i, π_i, ξ_i) is the GNS-construction of φ_i .

Note that a ucp map satisfying (1) is necessarily unique and property (1) implies that $\lambda: P \to P_r$ is faithful on A_p , for all $p \in V(\mathcal{G})$ so that we may and will view $A_p \subset P_r$ for all $p \in V(\mathcal{G})$ and the ucp maps $E_p: P_r \to A_p$ become conditional expectations under this identification. If all the ucp maps E_e^s are supposed to be GNS-faithful then the vertex reduced fundamental algebra P_r is the same as the reduced fundamental algebra constructed in [22] and the conditional expectations $E_p: P_r \to A_p$ are all GNS-faithful.

3. Free wreath products as fundamental algebras

Let $N \in \mathbb{N}^*$ and \mathcal{T}_N be the rooted tree with N+1 vertices p_0, \ldots, p_N , p_0 being the root and 2N edges $v_1, \ldots, v_N, \overline{v}_1, \ldots \overline{v}_N$, source maps $s(v_k) = p_0$ and range maps $r(v_k) = p_k$ for all $1 \le k \le N$.

3.1. The full version

Consider the graph of C*-algebras over \mathcal{T}_N given by $\mathcal{A}_{p_0} = C(H) \otimes C(S_N^+)$, $\mathcal{A}_{p_k} := C(G) \otimes \mathbb{C}^N$, $\mathcal{B}_{v_k} = \mathcal{B}_{\overline{v}_k} = C(H) \otimes \mathbb{C}^N$ ($1 \leq k \leq N$) with source map $s_{v_k} : C(H) \otimes \mathbb{C}^N \to C(H) \otimes L_k \subset C(H) \otimes C(S_N^+)$, $h \otimes e_j \mapsto h \otimes u_{kj}$ and range map $r_{v_k} : C(H) \otimes \mathbb{C}^N \to C(G) \otimes \mathbb{C}^N$ being the canonical inclusion. Note that our graph of C*-algebras has conditional expectations $\mathrm{id} \otimes E_k : C(H) \otimes C(S_N^+) \to C(H) \otimes L_k$ (Proposition 2.18) which can be non GNS-faithful and $E_H \otimes \mathrm{id} : C(G) \otimes \mathbb{C}^N \to C(H) \otimes \mathbb{C}^N$, coming from Proposition 2.5, which can also be non-GNS faithful.

Let us denote by \mathcal{A} the maximal fundamental C*-algebra of this graph of C*-algebras relative to the unique maximal subtree \mathcal{T}_N itself so that \mathcal{A} is the universal unital C*-

algebra generated by \mathcal{A}_{p_k} for $0 \leq k \leq N$ with the relations $s_{v_k}(a) = r_{v_k}(a)$ for all $a \in C(H) \otimes \mathbb{C}^N$ and all $1 \leq k \leq N$. Recall that $\nu_i : C(G) \to C(G)^{*_H N} \subset C(G \wr_{*,H} S_N^+)$ denotes the i^{th} -copy of C(G) in $C(G)^{*_H N}$, we also denote by ν the common restriction of the ν_i 's to C(H)

Proposition 3.1. There is a unique isomorphism $\pi: \mathcal{A} \to C(G \wr_{*,H} S_N^+)$ such that $\pi(h \otimes u_{ij}) = \nu(h)u_{ij}$ $(h \otimes u_{ij} \in \mathcal{A}_{p_0} = C(H) \otimes C(S_N^+))$ and, $\pi(a \otimes e_j) = \nu_i(a)u_{ij}$ for $a \otimes e_j \in \mathcal{A}_{p_i} = C(G) \otimes \mathbb{C}^N$, $1 \leq i, j \leq N$.

Proof. Let us show the existence of π . Note first that, since $\nu_i(a)$ and u_{ij} commute in $C(G \wr_{*,H} S_N^+)$, there exists a unique unital *-homomorphism $\pi_i : \mathcal{A}_{p_i} = C(G) \otimes \mathbb{C}^N \to C(G \wr_{*,H} S_N^+)$ such that $\pi_i(a \otimes e_j) = \nu_i(a)u_{ij}$, for all $a \in C(G)$ and $1 \leq j \leq N$. We also define $\pi_0 : \mathcal{A}_{p_0} = C(H) \otimes C(S_N^+) \to C(G \wr_{*,H} S_N^+)$, $\pi_0(h \otimes u_{ij}) = \nu(h)u_{ij}$. Next, for all $1 \leq i,j \leq N$, and $h \in C(H)$, one has $\pi_0(s_{v_i}(h \otimes e_j)) = \pi_0(h \otimes u_{ij}) = \nu(h)u_{ij}$ and $\pi_i(r_{v_i}(h \otimes e_j)) = \pi_i(h \otimes e_j) = \nu(h)u_{ij}$. By the universal property of \mathcal{A} , there exists a unique unital *-homomorphism $\pi : \mathcal{A} \to C(G \wr_{*,H} S_N^+)$ satisfying the properties of the proposition. Moreover, since the image of π contains u_{ij} for all i, j and $\sum_{j=1}^N \pi_i(a \otimes e_j) = \sum_{j=1}^N \nu_i(a)u_{ij} = \nu_i(a)$ for all $a \in A$ and $1 \leq i \leq N$, it follows that π is surjective.

To show that it is an isomorphism, we will give an inverse, using this time the universal property of $C(G \wr_{*,H} S_N^+)$. By the universal property of the full free product, there exists a unique unital *-homomorphism $\mu: C(G)^{*N} * C(S_N^+) \to \mathcal{A}$ such that, for all $1 \leq i \leq N$, $a \in C(G)$, $b \in C(S_N^+)$,

$$\mu(\nu_i(a)) = a \otimes 1 \in C(G) \otimes \mathbb{C}^N = \mathcal{A}_{p_i} \subset \mathcal{A} \text{ and } \mu(b) = 1 \otimes b \in C(H) \otimes C(S_N^+) = \mathcal{A}_{p_0} \subset \mathcal{A}.$$

Recall that $C(G \wr_{*,H} S_N^+) = C(G)^{*_H N} * C(S_N^+)/I$, where I is defined in Equation (1). Note that

$$\mu(\nu_{i}(a)u_{ij}) = (a \otimes 1)(1 \otimes u_{ij}) = \mu(\nu_{i}(a))s_{v_{i}}(1 \otimes e_{j}) = \mu(\nu_{i}(a))r_{v_{i}}(1 \otimes e_{j})$$
$$= (a \otimes 1)(1 \otimes e_{j}) = (1 \otimes e_{j})(a \otimes 1) = r_{v_{i}}(1 \otimes e_{j})\mu(\nu_{i}(a))$$
$$= s_{v_{i}}(1 \otimes e_{j})\mu(\nu_{i}(a)) = (1 \otimes u_{ij})\mu(\nu_{i}(a)) = \mu(u_{ij}\nu_{i}(a)).$$

Moreover, we have that for every $h \in C(H)$, and $1 \le i, j \le N$,

$$\mu(\nu_i(\iota(h))) = (\iota(h) \otimes 1) = s_{v_i}(h \otimes 1) = r_{v_i}(h \otimes 1) = (h \otimes 1)$$
$$= r_{v_j}(h \otimes 1) = s_{v_j}(h \otimes 1) = (\iota(h) \otimes 1) = \mu(\nu_j(\iota(h))),$$

Hence, the images of C(H) through μ coincide and $I \subset \ker(\mu)$ so there exists a unique unital *-homomorphism $\rho: C(G \wr_{*,H} S_N^+) \to \mathcal{A}$ which factorizes μ . It is clear that ρ is surjective and it is easy to check that ρ is the inverse of π . \square

3.2. The reduced version

Consider the graph of C*-algebras over \mathcal{T}_N given by $A_{p_0} = C_r(H) \otimes C_r(S_N^+)$, $A_{p_k} := C_r(G) \otimes \mathbb{C}^N$, $B_{v_k} = B_{\overline{v}_k} = C_r(H) \otimes \mathbb{C}^N$ for all $1 \leq k \leq N$ with source map $s_{v_k} : C_r(H) \otimes \mathbb{C}^N \to C_r(H) \otimes L_k \subset C_r(H) \otimes C_r(S_N^+)$, $h \otimes e_j \mapsto h \otimes u_{kj}$ and range map $r_{v_k} : C_r(H) \otimes \mathbb{C}^N \to C_r(G) \otimes \mathbb{C}^N$, being the canonical inclusion. Note that our graph of C*-algebras has faithful conditional expectations $\mathrm{id} \otimes E_k : C_r(H) \otimes C_r(S_N^+) \to C_r(H) \otimes L_k$ (Proposition 2.18) and $E_H \otimes \mathrm{id} : C_r(G) \otimes \mathbb{C}^N \to C_r(H) \otimes \mathbb{C}^N$, thanks to Proposition 2.5. Let A the vertex reduced fundamental C*-algebra of this graph of C*-algebras with faithful conditional expectations and view $A_{p_k} \subset A$ for all $0 \leq k \leq N$. By the universal property of A defined in Section 3.1, there exists a unique unital *-homomorphism (which is surjective) $\lambda' : A \to A$ such that $\lambda'|_{A_{p_0}} = \lambda_H \otimes \lambda_{S_N^+}$ and $\lambda'|_{A_{p_k}} = \lambda_G \otimes \mathrm{id}_{\mathbb{C}^N}$ for all $1 \leq k \leq N$. Let $E : A \to C_r(H) \otimes C_r(S_N^+)$ the GNS-faithful conditional expectation and define $\omega := h_H \otimes h_{S_N^+} \circ E$ so that $\omega|_{A_{p_0}} = h_H \otimes h_{S_N^+}$ and $\omega(c) = 0$ for $c \in A$ a reduced operator. The state $\omega \in A^*$ is called the fundamental state. Let $\lambda : C(G \wr_{*,H} S_N^+) \to C_r(G \wr_{*,H} S_N^+)$ be the canonical surjection.

Theorem 3.2. The Haar state $h \in C(G \wr_{*,H} S_N^+)^*$ is the unique state such that:

$$h(a_0\nu_{i_1}(b_1)a_1\nu_{i_2}(b_2)\dots\nu_{i_n}(b_n)a_n)=0$$

whenever $a_k \in C(S_N^+) \subset C(G \wr_{*,H} S_N^+)$, and $b_k \in C(G)$ are such that $E_H(b_k) = 0$ for all k and if $i_k = i_{k+1}$, then $E_{i_k}(a_k) = 0$.

There exists a unique unital *-isomorphism $\pi_r: A \to C_r(G \wr_{*,H} S_N^+)$ such that $\lambda \circ \pi = \pi_r \circ \lambda'$, where $\pi: A \to C(G \wr_{*,H} S_N^+)$ is the isomorphism of Proposition 3.1. Moreover, π intertwines the Haar state on $C_r(G \wr_{*,H} S_N^+)$ and the fundamental state $\omega \in A^*$.

An element of the form $a_0\nu_{i_1}(b_1)a_1\nu_{i_2}(b_2)\dots\nu_{i_n}(b_n)a_n \in C(G\wr_{*,H}S_N^+)$ and satisfying the conditions of Theorem 3.2 will be called a reduced operator.

Proof. Consider the state $\widetilde{\omega} := \omega \circ \lambda_N \circ \mu = h_H \otimes h_{S_N^+} \circ E \circ \lambda_N \circ \mu \in C(G\wr_{*,H} S_N^+)^*$, where $\mu := \pi^{-1} : C(G\wr_{*,H} S_N^+) \to \mathcal{A}$ has been constructed in the proof of Proposition 3.1. Let $\mathcal{C} \subset \mathcal{A}$ be the linear span of $C(S_N^+)$, $\nu(C(H))$ and all reduced operators in $C(G\wr_{*,H} S_N^+)$. Note that \mathcal{C} is dense in \mathcal{A} . By construction, the state $\widetilde{\omega}$ satisfies $\widetilde{\omega}|_{C(S_N^+)} = h_{S_N^+}, \widetilde{\omega} \circ \nu = h_H$ and $\widetilde{\omega}(c) = 0$ for $c \in C(G\wr_{*,H} S_N^+)$ a reduced operator so $\widetilde{\omega}$ satisfies the property of the state h stated in the Theorem hence $h = \widetilde{\omega}$ by density of \mathcal{C} . We will show that $\widetilde{\omega}$ is Δ -invariant, which will imply that it is the Haar state, and thus (since E is GNS faithful and $h_H \otimes h_{S_N^+}$ is faithful on $C_r(H) \otimes C_r(S_N^+)$) that A is isomorphic to the algebra obtained through the GNS-construction applied to the Haar state, namely the reduced C^* -algebra $C_r^*(G\wr_{*,H} S_N^+)$, which will complete the proof of the Theorem. It is enough to show that $\widetilde{\omega}$ is Δ invariant when restricted to \mathcal{C} . An element in \mathcal{C} is the sum of an element $x_0 \in C(S_N^+) \otimes C(H)$ and elements of the form $x = a_0 \nu_{i_1}(b_1) a_1 \dots a_{n-1} \nu_{i_n}(b_n) a_n$

with $n \geq 1$, $a_k \in C(S_N^+)$, $b_k \in C(G)^\circ$, where $C(G)^\circ := \{b \in C(G), E_H(b) = 0\}$, for all $0 \leq k \leq N$, and if there is k such that $i_k = i_{k+1}$, then $E_{i_k}(a_k) = 0$. It suffices to show $\Delta(x) \in \mathcal{C} \odot C(G \wr_{*,H} S_N^+)$ (where \odot is the algebraic tensor product). Using the Sweedler notation,

$$\Delta(x) = \sum_{(a_0)_{(1)}} \nu_{i_1}((b_1)_{(1)}) u_{i_1,s_1}(a_1)_{(1)} \nu_{i_2}((b_2)_{(1)}) u_{i_2,s_2} \dots (a_n)_{(1)}$$

$$\otimes (a_0)_{(2)} \nu_{s_1}((b_1)_{(2)}) (a_1)_{(1)} \nu_{s_2}((b_2)_{(2)}) \dots (a_n)_{(2)}.$$

By Remark 2.6 one has $E_H((b_k)_{(1)})=0$ hence, the left part of every term of the sum is a reduced operator whenever there is no k such that $i_k=i_{k+1}$ and $E_{i_k}((a_k)_{(1)})\neq 0$, we need only to take care of the remaining cases. Assume that we have k such that $i_k=i_{k+1}$, and $E_{i_k}((a_k)_{(1)})\neq 0$, with k being the smallest such integer. Then we can write $(a_k)_{(1)}=E_{i_k}((a_k)_{(1)})+(a_k)_{(1)}^{\circ}$, with $E_{i_k}((a_k)_{(1)}^{\circ})=0$. Fixing $s=(s_1,s_2,\ldots,s_n)\in N^n$, we can write

$$(a_0)_{(1)}\nu_{i_1}((b_1)_{(1)})u_{i_1,s_1}(a_1)_{(1)}\dots(a_n)_{(1)}\otimes(a_0)_{(2)}\nu_{s_1}((b_1)_{(2)})(a_1)_{(1)}\nu_{s_2}((b_2)_{(2)})\dots(a_n)_{(2)}$$

$$=(a_0)_{(1)}\nu_{i_1}((b_1)_{(1)})u_{i_1,s_1}(a_1)_{(1)}\dots(a_k)_{(1)}^{\circ}\dots(a_n)_{(1)}\otimes(a_0)_{(2)}\dots(a_n)_{(2)}$$

$$+(a_0)_{(1)}\nu_{i_1}((b_1)_{(1)})u_{i_1,s_1}(a_1)_{(1)}\dots E_{i_k}((a_k)_{(1)})\dots(a_n)_{(1)}\otimes(a_0)_{(2)}\dots(a_n)_{(2)},$$

we see that the first term of the decomposition is itself reduced up to the k-th term, and we can iterate the process with the next term such that $i_{k'} = i_{k'+1}$. In the end we get a sum of tensor such that the first term is always reduced, and we only have to show that the second vanishes. The second term is always of the form

$$\alpha_{k,s} = (a_0)_{(1)} \nu_{i_1}((b_1)_{(1)}) u_{i_1,s_1}(a_1)_{(1)} \dots E_{i_k}((a_k)_{(1)}) \dots (a_n)_{(1)} \otimes (a_0)_{(2)} \dots (a_n)_{(2)},$$

with maybe other conditional expectations appearing after rank k. The definition of the conditional expectations (E_i) gives $E_{i_k}((a_k)_{(1)}) = N \sum_{t=1}^N h((a_k)_{(1)} u_{i_k t}) u_{i_k t}$ hence,

$$\alpha_{k,s} = N \sum_{t=1}^{N} h((a_k)_{(1)} u_{i_k t})(a_0)_{(1)} \nu_{i_1}((b_1)_{(1)}) u_{i_1,s_1}(a_1)_{(1)} \dots \nu_{i_k}((b_k)_{(1)}) u_{i_k,t} \nu_{i_k}((b_{k+1})_{(1)})$$

$$\dots (a_n)_{(1)} \otimes (a_0)_{(2)} \dots (a_n)_{(2)}$$

$$= N \sum_{t=1}^{N} h((a_k)_{(1)} u_{i_k t})(a_0)_{(1)} \nu_{i_1}((b_1)_{(1)}) u_{i_1,s_1}(a_1)_{(1)} \dots \nu_{i_k}((b_k)_{(1)}(b_{k+1})_{(1)}) u_{i_k,t}$$

$$\dots (a_n)_{(1)} \otimes (a_0)_{(2)} \dots (a_n)_{(2)}$$

$$= N \sum_{t=1}^{N} h((a_k)_{(1)'} u_{i_k t})(a_0)_{(1)} \nu_{i_1}((b_1)_{(1)}) u_{i_1,s_1}(a_1)_{(1)} \dots \nu_{i_k}((b_k)_{(1)}(b_{k+1})_{(1)}) u_{i_k,t}$$

$$\dots (a_n)_{(1)} \otimes (a_0)_{(2)} \dots (a_k)_{(1)'} \dots (a_n)_{(2)}$$

The symbols (1)' and (2)' are used to differentiate between the summation indices of the scalars we will now move to the right side of the tensor product and the others summation indices:

$$\alpha_{k,s} = N \sum_{t=1}^{N} (a_0)_{(1)} \nu_{i_1}((b_1)_{(1)}) u_{i_1,s_1}(a_1)_{(1)} \dots \nu_{i_k}((b_k)_{(1)}(b_{k+1})_{(1)}) u_{i_k,t} \dots (a_n)_{(1)}$$

$$\otimes (a_0)_{(2)} \dots h((a_k)_{(1)'} u_{i_k t}) (a_k)_{(2)'} \dots (a_n)_{(2)}.$$

Now, by Lemma 2.19, we get $\sum_{(1)',(2)'} h((a_k)_{(1)'} u_{i_k t})(a_k)_{(2)'} = 0$. This is true for every value of t, hence α_k vanishes whenever we considered k as before. This proves that $\widetilde{\omega}$ is Δ -invariant. \square

The explicit computation of the Haar state can be used to show the following property of dual quantum subgroups in free wreath products.

Proposition 3.3. Let $C(H) \subset C(G)$ be a dual quantum subgroup and consider, by the universal property of $C(H \wr_* S_N^+)$, the unique unital *-homomorphism $\iota : C(H \wr_* S_N^+) \to C(G \wr_* S_N^+)$ such that $\iota \circ \nu_i = \nu_i : C(H) \to C(G \wr_* S_N^+)$ for all $1 \leq i \leq N$ and $\iota|_{C(S_N^+)}$ is the inclusion $C(S_N^+) \subset C(G \wr_* S_N^+)$. Then, ι is faithful so $H \wr_* S_N^+$ is a dual quantum subgroup of $G \wr_* S_N^+$.

Proof. It is clear that ι intertwines the comultiplications and, by Theorem 3.2, it also intertwines the Haar states so the restriction $\iota|_{\operatorname{Pol}(H\wr_*S_N^+)}:\operatorname{Pol}(H\wr_*S_N^+)\to\operatorname{Pol}(G\wr_*S_N^+)$ is injective (by faithfulness of the Haar states) and still intertwines the comultiplication. Hence, $H\wr_*S_N^+$ is a dual quantum subgroup of $G\wr_*S_N^+$ and ι itself faithful by Proposition 2.5. \square

3.3. The von Neumann version

Consider the graph of von Neumann algebras over \mathcal{T}_N given by $M_{p_0} = L^{\infty}(H) \otimes L^{\infty}(S_N^+)$, $M_{p_k} := L^{\infty}(G) \otimes \mathbb{C}^N$, $N_{e_k} = N_{\overline{e_k}} = L^{\infty}(H) \otimes \mathbb{C}^N$ for all $1 \leq k \leq N$ with source map $s_{e_k} : L^{\infty}(H) \otimes \mathbb{C}^N \to L^{\infty}(H) \otimes L_k \subset L^{\infty}(H) \otimes L^{\infty}(S_N^+)$, $h \otimes e_j \mapsto h \otimes u_{kj}$ and range map $r_{e_k} : L^{\infty}(H) \otimes \mathbb{C}^N \to L^{\infty}(G) \otimes \mathbb{C}^N$ being the canonical inclusion. We also consider family of faithful normal states $\omega_{p_0} = h_H \otimes h_{S_N^+}$, $\omega_{p_k} = h_G \otimes \operatorname{tr}$ (where tr is the normalized uniform trace on \mathbb{C}^N i.e. $\operatorname{tr}(e_j) = \frac{1}{N}$) and $\omega_{e_k} = h_H \otimes \operatorname{tr}$ for all $1 \leq k \leq N$. Hence, we get a graph of von Neumann algebras as defined in [22, Definition A.1]. Let us denote by M the fundamental von Neumann algebra at p_0 and view $M_{p_0} = L^{\infty}(H) \otimes L^{\infty}(S_N^+) \subset M$. Let φ the associated fundamental normal faithful state which is a trace if and only if G is Kac from the results of [22], as H is automatically Kac in that case. As a direct consequence of Proposition 3.2 and [22, Section A.5] we have the following. Note that, by Theorem 3.2, the unital faithful *-homomorphism

 $\nu_i: C(G) \to C(G \wr_{*,H} S_N^+)$ preserves the Haar states hence, it induces a unital normal faithful *-homomorphism $\nu_i: L^{\infty}(G) \to L^{\infty}(G \wr_{*,H} S_N^+)$.

Proposition 3.4. There exists a unique unital normal *-isomorphism $\pi''_r: M \to L^{\infty}(G \wr_{*,H} S_N^+)$ extending the isomorphism $\pi_r: A \to C_r(G \wr_{*,H} S_N^+)$ from Theorem 3.2. Moreover, $h \circ \pi''_r = \varphi$, where h also denotes the Haar state on $L^{\infty}(G \wr_{*,H} S_N^+)$.

Recall that
$$\operatorname{Aut}_{L^{\infty}(H)}(L^{\infty}(G), h_G) = \{\beta \in \operatorname{Aut}(L^{\infty}(G), h_G), \beta(L^{\infty}(H)) = L^{\infty}(H)\}.$$

Proposition 3.5. For all $\alpha \in \operatorname{Aut}_{L^{\infty}(H)}(L^{\infty}(G), h_G)$ there exists a unique h-preserving automorphism $\psi(\alpha)$ of $L^{\infty}(G \wr_{*,H} S_N^+)$ such that

$$\psi(\alpha)|_{\mathbf{L}^{\infty}(S_{\Sigma}^{+})} = \mathrm{id} \quad and \quad \psi(\alpha) \circ \nu_{i} = \nu_{i} \circ \alpha \quad \forall 1 \leq i \leq N.$$
 (2)

Moreover, $\psi : \operatorname{Aut}_{L^{\infty}(H)}(L^{\infty}(G), h_G) \to \operatorname{Aut}(L^{\infty}(G \wr_{*,H} S_N^+), h)$ is a continuous group homomorphism.

Proof. Let $\alpha \in \operatorname{Aut}_{L^{\infty}(H)}(L^{\infty}(G), h_G)$ and write $(L^2(G \wr_{*,H} S_N^+), \lambda, \xi)$ the GNS of the Haar state. To show that $\psi(\alpha)$ exists, it suffices to show that there exists a well defined unitary $U_{\alpha} \in \mathcal{B}(L^2(G \wr_{*,H} S_N^+))$ such that, for all reduced operator $a_0\nu_{i_1}(b_1)\dots\nu_{i_n}(b_n)a_n \in L^{\infty}(G \wr_{*,H} S_N^+)$,

$$U_{\alpha}(a_0\nu_{i_1}(b_1)\dots\nu_{i_n}(b_n)a_n\xi) = a_0\nu_{i_1}(\alpha(b_1))\dots\nu_{i_n}(\alpha(b_n))a_n\xi$$

Indeed, if such a unitary is constructed, then $\psi(\alpha)(x) := U_{\alpha}xU_{\alpha}^*$ does the job. To show that such a U_{α} exists it suffices to check that, for any reduced operator $x = a_0\nu_{i_1}(b_1)\dots\nu_{i_n}(b_n)a_n \in L^{\infty}(G\wr_{*,H}S_N^+)$ one has $h(x^*x) = h(y^*y)$, where $y := a_0\nu_{i_1}(\alpha(b_1))\dots\nu_{i_n}(\alpha(b_n))a_n$. This can be shown by induction on n, by first observing that $E_H \circ \alpha = \alpha \circ E_H$, where $E_H : L^{\infty}(G) \to L^{\infty}(H)$ is the canonical normal faithful conditional expectation. Indeed, since $\alpha^{-1} \circ E_H \circ \alpha$ is a normal h-preserving conditional expectation onto $L^{\infty}(H)$, we have, by uniqueness in Takesaki's Theorem, that $\alpha^{-1} \circ E_H \circ \alpha = E_H$. It follows that the element y is reduced whenever x is. Then, one can easily use the Haar state formula in Theorem 3.2 to prove our claim by induction. The uniqueness of $\psi(\alpha)$ and the fact that it preserves the Haar state are clear.

The fact that it is a group homomorphism follows by uniqueness. To prove continuity, it suffices to check that the map ψ_x : $\operatorname{Aut}_{L^\infty(H)}(\operatorname{L}^\infty(G),h_G)) \to \operatorname{L}^2(G\wr_{*,H}S_N^+,h)$, $(\alpha\mapsto\psi(\alpha)(x)\xi)$, is continuous for all $x\in\operatorname{L}^\infty(G\wr_{*,H}S_N^+)$, where $\xi\in\operatorname{L}^2(G\wr_{*,H}S_N^+,h)$ is the canonical cyclic vector. Define $\mathcal{C}\subset\operatorname{L}^\infty(G\wr_{*,H}S_N^+)$ to be the linear span of $\operatorname{L}^\infty(S_N^+)$ and the reduced operators and observe that \mathcal{C} is a σ -strongly dense unital *-subalgebra of $\operatorname{L}^\infty(G\wr_{*,H}S_N^+)$. Note that the subset:

$$\{x \in \mathcal{L}^{\infty}(G \wr_{*,H} S_N^+) : \psi_x \text{ is continuous }\} \subseteq \mathcal{L}^{\infty}(G \wr_{*,H} S_N^+)$$

is a clearly a subspace and it is σ -strongly closed since, $\psi(\alpha)$ being h-preserving, we have,

$$\|\psi_x(\alpha) - \psi_y(\alpha)\|^2 = \|\psi(\alpha)(x)\xi - \psi(\alpha)(y)\xi\|^2 = \|\psi(\alpha)(x-y)\xi\|^2 = h((x-y)^*(x-y))$$

for all $\alpha \in \operatorname{Aut}_{L^{\infty}(H)}(L^{\infty}(G), h_G)$ and $x, y \in L^{\infty}(G \wr_{*,H} S_N^+)$. Hence, it suffices to show that ψ_x is continuous for all $x \in L^{\infty}(S_N^+)$ or for x a reduced operator. When $x \in L^{\infty}(S_N^+)$ the map ψ_x is the constant map equals to $x\xi$. When $x = a_0\nu_{i_1}(b_1)\dots\nu_{i_n}(b_n)a_n \in L^{\infty}(G \wr_{*,H} S_N^+)$ is a reduced operator, we have, by induction on n, $\psi_x(\alpha) - \psi_x(\mathrm{id}) = \sum_{k=1}^n \varphi_k(\alpha)$, where:

$$\varphi_k(\alpha) := a_0 \nu_{i_1}(b_1) a_1 \dots \nu_{i_{k-1}}(b_k) a_k \nu_{i_k}(\alpha(b_k) - b_k) a_k \nu_{i_{k+1}}(\alpha(b_{k+1})) a_{k+1} \dots \nu_{i_n}(\alpha(b_n)) a_n \xi$$

with the natural conventions so that $\varphi_0(\alpha)$ and $\varphi_n(\alpha)$ make sense. Using the formula for the Haar state given in Theorem 3.2 and the fact that α is h_G preserving, a direct computation gives:

$$\|\varphi_k(\alpha)\|^2 = \left(\prod_{l=0}^n \|a_l\|_{2,S_N^+}^2\right) \left(\prod_{1 \le l \le n, l \ne k} \|b_l\|_{2,G}^2\right) \|\alpha(b_k) - b_k\|_{2,G}^2$$

where, for a CQG G, $\|\cdot\|_{2,G}$ is the 2-norm given by the Haar state on $L^{\infty}(G)$. It follows that if $\alpha_s \to_s$ id in $\operatorname{Aut}_{L^{\infty}(H)}(L^{\infty}(G), h_G)$ then, for all $1 \leq k \leq n$, $\varphi_k(\alpha_s) \to_s 0$ in $L^2(G \wr_{*,H} S_N^+)$ for the norm. Hence, ψ_x is continuous at id so it is continuous since it is a group homomorphism. \square

3.4. An inductive construction for free wreath products algebras

We define inductively the C*-algebra \mathcal{A}_i $(1 \leq i \leq N)$ containing $C(H) \otimes C(S_N^+)$ with a state $\omega_i \in \mathcal{A}_i^*$ by $\mathcal{A}_0 = C(H) \otimes C(S_N^+)$ and $\omega_0 = h_H \otimes h_{S_N^+}$ the Haar state on \mathcal{A}_0 . Let $A_0 := C_r(H) \otimes C_r(S_N^+)$ be the GNS-construction of ω_0 and $M_0 := L^{\infty}(H) \otimes L^{\infty}(S_N^+) = A_0''$ the von Neumann algebra generated in the GNS construction. Write $\lambda_0 : \mathcal{A}_0 \to A_0$ the GNS morphism. We still denote by ω_0 the associated faithful state on A_0 (resp. faithful normal state on M_0). Suppose that $(\mathcal{A}_i, \omega_i)$ is constructed and let A_i be the C*-algebra of the GNS construction of ω with GNS morphism $\lambda_i : \mathcal{A}_i \to A_i$ and $M_i = A_i''$ the von Neumann algebra generated in the GNS construction. Define the full free product $\mathcal{A}_{i+1} := (C(G) \otimes \mathbb{C}^N) {}_{C(H) \otimes \mathbb{C}^N} \mathcal{A}_i$, where the amalgamation is with respect to the faithful normal unital *-homomorphisms $r_{e_{i+1}} : C(H) \otimes \mathbb{C}^N \to C(G) \otimes \mathbb{C}^N$ and $s_{e_{i+1}} : C(H) \otimes \mathbb{C}^N \to C(H) \otimes L_{i+1} \subset C(H) \otimes C(S_N^+) \subset \mathcal{A}_i$. Let $A_{i+1} := (C_r(G) \otimes \mathbb{C}^N, h_G \otimes \operatorname{tr}) {}_{C_r(H) \otimes \mathbb{C}^N} (A_i, \omega_i)$ be the reduced free product with amalgamation with respect to the same maps but at the reduced level. Let $\lambda_{i+1} : \mathcal{A}_{i+1} \to A_i$ the unique surjective unital *-homomorphism such that $\lambda_{i+1}(a \otimes x) = \lambda_G(a) \otimes x$ for all $a \in C(G)$,

 $x \in \mathbb{C}^N$ and $\lambda_{i+1}|_{\mathcal{A}_i} = \lambda_i$. Define $\omega_{i+1} := ((h_G \otimes \operatorname{tr}) * \omega_i) \circ \lambda_{i+1}$ and note that A_{i+1} is the C*-algebra of the GNS construction of ω_{i+1} and $\lambda_{i+1} : \mathcal{A}_{i+1} \to A_{i+1}$ is the GNS morphism. It follows that $M_{i+1} = A_{i+1}'' = (\operatorname{L}^{\infty}(G) \otimes \mathbb{C}^N) \underset{\operatorname{L}^{\infty}(H) \otimes \mathbb{C}^N}{*} M_i$, where the amalgamated free product von Neumann algebra is with respect to the faithful normal unital *-homomorphisms $r_{e_i} : \operatorname{L}^{\infty}(H) \otimes \mathbb{C}^N \to \operatorname{L}^{\infty}(G) \otimes \mathbb{C}^N$ and $s_{e_i} : \operatorname{L}^{\infty}(H) \otimes \mathbb{C}^N \to \operatorname{L}^{\infty}(H) \otimes L_{i+1} \subset \operatorname{L}^{\infty}(H) \otimes \operatorname{L}^{\infty}(S_N^+) \subset M_i$ and with respect to the faithful normal states $h_G \otimes \operatorname{tr}$ and ω_i . We still denote by ω_{i+1} the faithful free product state on A_{i+1} and the faithful normal free product state on M_{i+1} .

Proposition 3.6. There exists unique state preserving *-isomorphisms

$$\rho: (C(G \wr_{*,H} S_N^+), h) \to (\mathcal{A}_N, \omega_N) \quad and \quad \rho_r: (C_r(G \wr_{*,H} S_N^+), h) \to (\mathcal{A}_N, \omega_N)$$

and a normal *-isomorphism $\rho_r'': (L^{\infty}(G \wr_{*,H} S_N^+), h) \to (M_N, \omega_N)$, where h the Haar state on $G \wr_{*,H} S_N^+$, such that, writing $A_N = \cup_i^{\dagger} A_i$, ρ maps $C(S_N^+)$ onto $A_0 = C(H) \otimes C(S_N^+)$ via $x \mapsto 1 \otimes x$ and, for all $1 \leq i \leq N$, $\rho(\nu_i(a)) = a \otimes 1 \in C(G) \otimes \mathbb{C}^N \subset A_i = (C(G) \otimes \mathbb{C}^N) \underset{C(H) \otimes \mathbb{C}^N = C(H) \otimes L_i}{*} A_{i-1}$ and, $\lambda \circ \rho = \rho_r \circ \lambda_N$, $\lambda \circ \rho_r'' = \rho_r'' \circ \lambda_N$, where $\lambda : C(G \wr_{*,H} S_N^+) \to C_r(G \wr_{*,H} S_N^+)$ is the canonical surjection.

Proof. Uniqueness being obvious, it suffices to show the existence and that ρ intertwines the states. The existence of ρ_r and ρ_r'' follows from that. It is easy to see, by the universal property of $C(G \wr_{*,H} S_N^+) = C(G)^{*_H N} * C(S_N^+)/I$ that there exists a unital *-homomorphism ρ satisfying the conditions of the statement. To show that ρ is an isomorphism, we construct an inverse $\rho': A_N \to C(G \wr_{*,H} S_N^+)$. To do so, we construct inductively unital *-homomorphisms $\rho'_i: \mathcal{A}_i \to C(G \wr_{*,H} S_N^+)$ by letting $\rho'_0: \mathcal{A}_0 = C(H) \otimes C(S_N^+) \to C(G \wr_{*,H} S_N^+), \ a \otimes b \mapsto \nu(a)b \text{ and, if } \rho'_{i-1}: \mathcal{A}_{i-1} \to \mathcal{A}_{i-1}$ $C(G \wr_{*,H} S_N^+)$ is defined, we use the universal property of the full amalgamated free product $\mathcal{A}_i = (C(G) \otimes \mathbb{C}^N) \underset{C(H) \otimes \mathbb{C}^N = C(H) \otimes L_i}{*} \mathcal{A}_{i-1}$ to define $\rho_i' : \mathcal{A}_i \to C(G \wr_{*,H} S_N^+)$ to be the unique unital *-homomorphism such that $\rho'_i|_{A_{i-1}} = \rho'_{i-1}$ and $\rho'_i(a \otimes e_i) = \nu_i(a)u_{ij}$ for all $a \otimes e_j \in C(G) \otimes \mathbb{C}^N$, where $(e_j)_{1 \leq j \leq N}$ is the canonical orthonormal basis of \mathbb{C}^N . Then, since $\rho'_{i|A_{i-1}} = \rho'_{i-1}$, there exists a unique unital *-homomorphism $\rho': \mathcal{A}_N = \cup_i^{\uparrow} \mathcal{A}_i \to C(G \wr_{*,H} S_N^+)$ such that $\rho'|_{\mathcal{A}_i} = \rho'_i$. It is then easy to check that ρ' is the inverse of ρ . It remains to show that ρ intertwines the states. By the uniqueness property of the state h stated in Theorem 3.2, it suffices to show that $\omega_N(\rho(c)) = 0$ for $c \in C(G_{k,H}, S_N^+)$ a reduced operator in the sense of Theorem 3.2. So let $c = a_0 \nu_{i_1}(b_1) a_1 \dots \nu_{i_n}(b_n) a_n \in C(G_{*,H} S_N^+)$ be a reduced operator and note that, if at least one the i_k is equal to N then $\rho(c)$ is a word with letters alternating from $C(G) \otimes \mathbb{C}^N \oplus C(H) \otimes \mathbb{C}^N$, and $A_{N-1} \oplus C(H) \otimes L_N$ (where, when, for $C \subset B$ with conditional expectation, $B \ominus C$ denotes the kernel of the conditional expectation onto C) so, by definition of the free product state, $\omega_N(\rho(c)) = 0$. If for all $k, i_k \leq N-1$,

then we can view $\rho(c) \in \mathcal{A}_{N-1}$ and since $\omega_N|_{\mathcal{A}_{N-1}} = \omega_{N-1}$, we can repeat the argument inductively and eventually deduce that $\omega_N(\rho(c)) = 0$. \square

In the setting of von Neumann algebras, the inductive construction implies the following.

Proposition 3.7. The following holds for $G \wr_{*,H} S_N^+$, with ψ defined in Equation (2),

- (1) The modular group of the Haar state h is $\sigma_t := \psi(\sigma_t^G)$, for all $t \in \mathbb{R}$, where $(\sigma_t^G)_{t \in \mathbb{R}}$ is the modular group of the Haar state of G on $L^{\infty}(G)$.
- (2) The scaling group is $\tau_t := \psi(\tau_t^G)$, $\forall t \in \mathbb{R}$, where τ_t^G is the scaling group of G.
- (3) $T(G \wr_* S_N^+) = \{t \in \mathbb{R} : \tau_t^G = \mathrm{id}\} \ \forall N \geq 2 \ and \ N \neq 3 \ and \ \tau(G \wr_{*,H} S_N^+) = \tau(G) \ \forall N \in \mathbb{N}^*.$

Proof. (1). We first note that, for all $t \in \mathbb{R}$, $\sigma_t^G \in \operatorname{Aut}_{L^{\infty}(H)}(L^{\infty}(G), h_G)$. From [47, Theorem 2.6], the modular group of the free product state ω_1 on the amalgamated free product $M_1 = (L^{\infty}(G) \otimes \mathbb{C}^N) \underset{L^{\infty}(H) \otimes \mathbb{C}^N = L^{\infty}(H) \otimes L_1}{*} (L^{\infty}(H) \otimes L^{\infty}(S_N^+))$ satisfies

 $\sigma_t^{\omega_1}|_{\mathcal{L}^{\infty}(G)} = \sigma_t^{h_{S_N^+}} = \text{id} \text{ and } \sigma_t(x \otimes 1) = \sigma_t^G(x) \otimes 1 \text{ for all } x \in \mathcal{L}^{\infty}(G).$ Identifying $(M_N, \omega_N) \simeq (\mathcal{L}^{\infty}(G)_{*,H} S_N^+), h)$ with the isomorphism of Proposition 3.6 and applying inductively [47, Theorem 2.6] to our finite sequence of von Neumann algebra $(M_i)_{1 \leq i \leq N}$ gives the result.

(2). Note that, for all $t \in \mathbb{R}$, $\tau_t^G \in \operatorname{Aut}_{L^{\infty}(H)}(L^{\infty}(G), h_G)$. To show that $\tau_t := \psi(\tau_t^G)$ is the scaling group we have to show that $\Delta \sigma_t = (\tau_t \otimes \sigma_t)\Delta$. Using the properties of σ_t , τ_t and the definition of Δ we have, for all $1 \le i \le N$ and all $a \in \operatorname{Pol}(G)$,

$$\Delta(\sigma_t(\nu_i(a))) = \Delta(\nu_i(\sigma_t^G(a))) = \sum_{j=1}^N (\nu_i \otimes \nu_j) (\Delta_G(\sigma_t^G(a))) (u_{ij} \otimes 1)$$

$$= \sum_{j=1}^N (\nu_i \otimes \nu_j) ((\tau_t^G \otimes \sigma_t^G) \Delta_G(a)) (u_{ij} \otimes 1)$$

$$= \sum_{j=1}^N (\tau_t \otimes \sigma_t) (\nu_i \otimes \nu_j) (\Delta_G(a)) (u_{ij} \otimes 1) = (\tau_t \otimes \sigma_t) \Delta(\nu_i(a)),$$

where, in the last equality, we used that $\tau_t(u_{ij}) = u_{ij}$. Since the modular group and scaling group act as the identity on $L^{\infty}(S_N^+)$, the equality $\Delta \sigma_t(u_{ij}) = (\tau_t \otimes \sigma_t)\Delta(u_{ij})$ is clear.

(3). Let us compute the T-invariant in the non-amalgamated case. For $N \geq 4$, the irreducible representations of $G \wr_* S_N^+$ are completely classified in [27] and it follows that the only dimension 1 irreducible representation is the trivial representation. Hence, $T(G \wr_* S_N^+) = \{t \in \mathbb{R} : \tau_t = \mathrm{id}\}$ and we conclude the proof using assertion (1). For N = 2,

we know from Proposition 2.23 that $G
vert_* S_2^+ \simeq G^{*2} \rtimes S_2$ and the *T*-invariant of such quantum groups is computed in Proposition 2.17. Using also Proposition 2.8 concludes the computation.

To compute the τ -invariant, we may and will assume that $N \geq 2$. Since ψ is continuous, one has $\tau(G \wr_{*,H} S_N^+) \subseteq \tau(G)$. Let us show that $\tau(G) \subseteq \tau(G \wr_{*,H} S_N^+)$. Applying Remark 2.1 with $M = \mathcal{L}^{\infty}(G \wr_{*,H} S_N^+) = \left(\mathcal{L}^{\infty}(G) \otimes \mathbb{C}^N\right)_{\mathcal{L}^{\infty}(H) \otimes \mathbb{C}^N = \mathcal{L}^{\infty}(H) \otimes \mathcal{L}_N} M_{N-1}$, ω the Haar state and $A = \mathcal{L}^{\infty}(G) \otimes \mathbb{C}^N$ we see that the map $\mathbb{R} \to \operatorname{Aut}(\mathcal{L}^{\infty}(G))$, $t \mapsto \sigma_t|_A = \sigma_t^G \otimes \operatorname{id}$ is $\tau(G \wr_{*,H} S_N^+)$ -continuous. Hence, $(t \mapsto \sigma_t^G)$ is $\tau(G \wr_{*,H} S_N^+)$ -continuous so $\tau(G) \subseteq \tau(G \wr_{*,H} S_N^+)$. \square

4. Approximation properties

Theorem A is a consequence of the following more general statement.

Theorem 4.1. For G a CQG with H a dual quantum subgroup and $N \geq 1$, the following holds

- (1) \widehat{G} is exact if and only if $\widehat{G}_{\wr_*,H} S_N^+$ is exact.
- (2) If Irr(H) is finite, then \widehat{G} has the Haagerup property if and only if $G \wr_{*,H} \widehat{S}_N^+$ has the Haagerup property.
- (3) If G is Kac and H is co-amenable then \widehat{G} is hyperlinear if and only if $\widehat{G}_{\ell_*,H} S_N^+$ is hyperlinear.
- (4) If $N \geq 5$, $G \wr_{*,H} S_N^+$ is not co-amenable for all G and all H.
 - If N ∈ {3,4} and H is a proper dual quantum subgroup of G (2.11) then G\(\cap{\chi}_{*,H} S_N^+\)
 is not co-amenable whenever G is Kac.
 - $G \wr_{*,H} S_2^+$ is co-amenable if and only if $G^{*_H 2}$ is co-amenable.
- (5) \widehat{G} is K-amenable if and only if $\widehat{G}_{1,H} \stackrel{\frown}{S_N}$ is K-amenable.

Proof. (1). It is known from [9] that $C_r(S_N^+)$ is exact. Hence, the result follows from [22, Corollary 3.30].

(2). Note that, at $N \leq 4$, S_N^+ is a co-amenable Kac CQG [1] so that $L^{\infty}(S_N^+)$ has the Haagerup property and, at $N \geq 5$, $L^{\infty}(S_N^+)$ has the Haagerup property by [9]. We show the result by using the inductive construction of $L^{\infty}(G \wr_{*,H} S_N^+) = M_N$, with $M_0 = L^{\infty}(H) \otimes L^{\infty}(S_N^+)$ and $M_{i+1} = (L^{\infty}(G) \otimes \mathbb{C}^N) {}_{L^{\infty}(H) \otimes \mathbb{C}^N} M_i$. At i = 0, $M_0 = L^{\infty}(H) \otimes L^{\infty}(S_N^+)$ has the Haagerup property, because $L^{\infty}(H)$ is finite-dimensional. Assume that M_i has the Haagerup property. It follows from [12, Corollary 8.2] that $M_{i+1} = (L^{\infty}(G) \otimes \mathbb{C}^N) {}_{L^{\infty}(H) \otimes \mathbb{C}^N} M_i$ also has the Haagerup property whenever \widehat{G} has the Haagerup property. By induction, $M_N = L^{\infty}(G \wr_{*,H} S_N^+)$ has the Haagerup property. Suppose now that $\widehat{G \wr_{*,H} S_N^+}$ has the Haagerup property. It is easy to check, using the definition of Haagerup property of [15] with respect to a faithful normal state (f.n.s.)

that if (M,ω) , $\omega\in M_*$ a f.n.s., has the Haagerup property and $P\subset M$ is a von Neumann subalgebra with a normal and state-invariant conditional expectation $E:M\to P$ then, $(P,\omega_{|P})$ also has the Haagerup property. Hence, since by construction of amalgamated free products we do have such conditional expectations on each leg of the amalgamated free product we first deduce from the inductive construction state-preserving isomorphism

$$(\mathrm{L}^{\infty}(G \wr_{*,H} S_N^+), h) \simeq \left((\mathrm{L}^{\infty}(G) \otimes \mathbb{C}^N) \underset{\mathrm{L}^{\infty}(H) \otimes \mathbb{C}^N}{*} M_{N-1}, \omega_N \right)$$

that $L^{\infty}(G) \otimes \mathbb{C}^N$ has the Haagerup property which implies that $L^{\infty}(G)$ also has it.

- (3). By Proposition [2, Corollary 3.7], $\widehat{S_N^+}$ is hyperlinear for all $N \in \mathbb{N}^*$. Assuming that G is Kac (so that $G\wr_{*,H}S_N^+$ also is), and that H is co-amenable so that $L^\infty(H)$ is amenable, we can apply the strategy of the proof of (2) by using [4, Corollary 4.5] to deduce the fact that if \widehat{G} is hyperlinear then so is $G\wr_{*,H}S_N^+$. The converse is clear.
- (4). The case N=2 is a consequence of Propositions 2.23 and 2.17. Suppose that $N\geq 3$ and $G\wr_{*,H}S_N^+$ is co-amenable. Since G^{*_HN} and S_N^+ are both compact quantum subgroup of $G\wr_{*,H}S_N^+$ by Remark 2.22, it follows that G^{*_HN} and S_N^+ are both co-amenable [43]. It implies that $N\leq 4$, since S_N^+ is not co-amenable for all $N\geq 5$ [1]. Moreover, since $N\geq 3$, G^{*_HN} is not co-amenable whenever H is proper by Proposition 2.13 since H has infinite index in $G*_HG$.
- (5). By [54] $\widehat{S_N^+}$ is K-amenable hence, if \widehat{G} is K-amenable, so is \widehat{H} and, a direct application of [23, Theorem 5.1 and 5.2], implies that $G\wr_{*,H}S_N^+$ is K-amenable. Let us prove the converse. To ease the notations we write $\mathcal{A}=C(G\wr_{*,H}S_N^+)$ and $A=C_r(G\wr_{*,H}S_N^+)$ during this proof. Let $\lambda:\mathcal{A}\to A$ be the canonical surjection and assume that $G\wr_{*,H}S_N^+$ is K-amenable i.e. there exists $\alpha\in KK(A,\mathbb{C})$ such that $[\lambda]\otimes\alpha=[\varepsilon]\in KK(\mathcal{A},\mathbb{C})$, where \otimes denotes here the Kasparov product and $\varepsilon:\mathcal{A}\to\mathbb{C}$ the counit of the free wreath product. Let us show that G^{*_HN} is K-amenable (and so \widehat{G} also is). Consider the canonical inclusion $\pi:C(G)^{*_HN}\to\mathcal{A}$. Note that π does not intertwine the comultiplications so that we can not deduce the K-amenability of G^{*_HN} by using the stability of K-amenability by dual quantum subgroup. However, by Theorem 3.2, the canonical inclusion $\pi:C(G)^{*_HN}\to\mathcal{A}$ intertwines the Haar states hence, there exists a unital *homomorphism $\pi_r:C_r(G^{*_HN})\to\mathcal{A}$ intertwines the Haar states hence, there exists a unital *homomorphism $\pi_r:C_r(G^{*_HN})\to\mathcal{A}$ such that $\pi_r\circ\lambda_G=\lambda\circ\pi$, where $\lambda_G:C(G^{*_HN})\to C_r(G^{*_HN})$ is the canonical surjection. Define $\beta:=[\pi_r]\otimes\alpha\in KK(C_r(G^{*_HN}),\mathbb{C})$. One has:

$$[\lambda_G] \otimes \beta = [\lambda_G] \otimes [\pi_r] \otimes \alpha = [\pi_r \circ \lambda_G] \otimes \alpha = [\lambda \circ \pi] \otimes \alpha$$
$$= [\pi] \otimes [\lambda] \otimes \alpha = [\pi] \otimes [\varepsilon] = [\varepsilon \circ \pi] = [\varepsilon_G].$$

This shows that $\widehat{G^{*_H N}}$ is K-amenable. \square

Remark 4.2. To deduce Theorem A it remains to deduce the co-amenability statement for N=2 which follows from Proposition 2.8.

Remark 4.3. As explained in the Introduction, except for N=2, the stability of the Haagerup property under the free wreath product construction was open. In the case where $G=\widehat{\Gamma}$ is the dual of a discrete group, and H is a finite subgroup, then the stability of the Haagerup property and of the weak amenability with constant 1 has been proved in [28]. In the general but non-amalgamated case, only the stability of the central ACPAP, which is a strengthening of both the Haagerup property and the weak amenability with constant 1 was known. However, it follows from Proposition 2.17 that $\Lambda_{cb}(\widehat{G}) \cdot \widehat{S}_2^+ = \Lambda_{cb}(\widehat{G})$. For $N \geq 3$, we believe that weak amenability with constant 1 is also stable under the free wreath product construction and one obvious way to do the proof would be by induction, using our inductive construction of the von Neumann algebra as well as an amalgamated version (over a finite dimensional subalgebra) of the result of Xu-Ricard [42]. Let us mention that for classical wreath products it has been proved by Ozawa [38] that for any non-trivial discrete group Λ and any non-amenable discrete group Γ , the classical wreath product $\Lambda \wr \Gamma$ is not weakly amenable.

5. The von Neumann algebra of a free wreath product

This section contains the proofs of Theorems B and C. Recall that $(\sigma_t^G)_{t\in\mathbb{R}}$ denotes the modular group of the Haar state on $L^{\infty}(G)$.

Theorem B is a direct consequence of the following more general statement.

Theorem 5.1. Suppose that Irr(G) is infinite, Irr(H) is finite, $L^{\infty}(G)' \cap L^{\infty}(H) = \mathbb{C}1$ and $N \geq 4$. Then $L^{\infty}(G)_{*,H} S_N^+$ is a non-amenable full and prime factor without any Cartan subalgebra and the following holds.

- If G is Kac then $L^{\infty}(G \wr_{*,H} S_N^+)$ is a type II_1 factor.
- If G is not Kac then $L^{\infty}(G \wr_{*,H} S_N^+)$ is a type III_{λ} factor for some $\lambda \neq 0$ and:

$$T(\operatorname{L}^{\infty}(G \wr_{*,H} S_N^+)) = \{t \in \mathbb{R} : \exists u \in \mathcal{U}(\operatorname{L}^{\infty}(H)), \ \sigma_t^G = \operatorname{Ad}(u)\}.$$

Moreover, in the non-amalgamated case, we have $\tau(L^{\infty}(G \wr_* S_N^+)) = \tau(G)$.

Proof of Theorem B. First observe that, by Lemma 2.4, both $L^{\infty}(G)$ and $L^{\infty}(S_N^+)$ are diffuse (since $N \geq 4$). Hence, $L^{\infty}(G) \otimes \mathbb{C}^N$ and $L^{\infty}(H) \otimes L^{\infty}(S_N^+)$ are also diffuse and the inclusions $L^{\infty}(H) \otimes \mathbb{C}^N \subset L^{\infty}(G) \otimes \mathbb{C}^N$ and $L^{\infty}(H) \otimes L_k \subset L^{\infty}(H) \otimes L^{\infty}(S_N^+)$ are without trivial corner for all $1 \leq k \leq N$, in the sense of [32] (since $L^{\infty}(H) \otimes \mathbb{C}^N$ is finite dimensional hence purely atomic, see [32, Lemma 5.2]).

We show the result by using the inductive construction of $L^{\infty}(G \wr_{*,H} S_N^+) = M_N$, with $M_0 = L^{\infty}(H) \otimes L^{\infty}(S_N^+)$ and $M_{i+1} = (L^{\infty}(G) \otimes \mathbb{C}^N) \underset{L^{\infty}(H) \otimes \mathbb{C}^N}{*} M_i$. At i = 0, we know that

 M_0 and $L^{\infty}(G) \otimes \mathbb{C}^N$ are diffuse so the inclusions $B = L^{\infty}(H) \otimes \mathbb{C}^N \subset L^{\infty}(G) \otimes \mathbb{C}^N$ and $B = L^{\infty}(H) \otimes L_1 \subset M_0$ are without trivial corner. Moreover, it follows from Lemma 2.20 that:

$$M'_0 \cap (L^{\infty}(G) \otimes \mathbb{C}^N)' \cap B \subset (L^{\infty}(G)' \cap L^{\infty}(H)) \otimes (L^{\infty}(S_N^+)' \cap L_1) = \mathbb{C}.$$

Hence, we may apply [32, Theorem E] to deduce that M_1 is a non-amenable prime factor. In particular M_1 is diffuse and a direct induction shows that M_k is a non-amenable prime factor for all $1 \leq k \leq N$, in particular, $M_N = L^{\infty}(G \wr_{*,H} S_N^+)$ is a non-amenable prime factor.

Suppose that $A \subset M := L^{\infty}(G \wr_{*,H} S_N^+) = \left(L^{\infty}(G) \otimes \mathbb{C}^N\right) \underset{L^{\infty}(H) \otimes \mathbb{C}^N}{*} M_{N-1}$ is a Cartan subalgebra. By the previous discussion, M_{N-1} is a non-amenable factor. Hence, it has no amenable direct summand and we may apply [6, Theorem B] to deduce that $A \prec_M L^{\infty}(H) \otimes \mathbb{C}^N$. Since $L^{\infty}(H) \otimes \mathbb{C}^N$ is finite dimensional, this contradicts the fact that A, being maximal abelian in the diffuse von Neumann algebra M, is itself diffuse.

To see that $L^{\infty}(G\wr_{*,H}S_N^+)$ is full, we may now apply [50, Theorem 4.10] (since $L^{\infty}(G)\otimes \mathbb{C}^N$ is diffuse and $L^{\infty}(H)\otimes \mathbb{C}^N$ is finite dimensional so the inclusion $L^{\infty}(H)\otimes \mathbb{C}^N\subset L^{\infty}(G)\otimes \mathbb{C}^N$ is entirely non trivial and, as shown before, M_{N-1} is diffuse).

When G is Kac, $G \wr_{*,H} S_N^+$ is also Kac hence, $L^{\infty}(G \wr_{*,H} S_N^+)$ is a II₁-factor. When G is not Kac, $L^{\infty}(G \wr_{*,H} S_N^+)$ is a non-amenable factor by the first part of the proof but not a finite factor by [24, Theorem 8]. Hence, it is a type III_{λ} factor with $\lambda \neq 0$ since it is a full factor [14, Proposition 3.9]. To compute the T-invariant, we view again $M := L^{\infty}(G \wr_{*,H} S_N^+) = (L^{\infty}(G) \otimes \mathbb{C}^N) \underset{L^{\infty}(H) \otimes \mathbb{C}^N}{*} M_{N-1}$ and we also note that, since M_{N-1} is diffuse and the amalgam $B := L^{\infty}(H) \otimes L_N = L^{\infty}(H) \otimes \mathbb{C}^N$ is finite dimensional, we have $M_{N-1} \not\prec_M B$. We can now use [31] to deduce that $T := T(L^{\infty}(G \wr_{*,H} S_N^+)) =$ $\{t \in \mathbb{R} : \exists u \in \mathcal{U}(B) \ \sigma_t = \mathrm{Ad}(u)\}, \text{ where } (\sigma_t)_{t \in \mathbb{R}} \text{ is the modular group of the Haar state}$ on $L^{\infty}(G \wr_{*,H} S_N^+)$, which coincides with the free product state. In particular one has $\sigma_t|_{L^{\infty}(G)\otimes\mathbb{C}^N} = \sigma_t^G \otimes \mathrm{id}$. Let $T' := \{t \in \mathbb{R} : \sigma_t^G = \mathrm{id}\}$. It is clear that $T' \subseteq T$. Let $t \in T$ and $u \in \mathcal{U}(B)$ such that $\sigma_t = \mathrm{Ad}(u)$. Write $u = \sum_{k=1}^N u_k \otimes e_k \in \mathrm{L}^{\infty}(H) \otimes \mathbb{C}^N$; where $u_k \in L^{\infty}(H)$ is unitary for all $1 \leq k \leq N$. For all $b \in L^{\infty}(G)$ and all $1 \leq k \leq N$ we find $\sigma_t(b\otimes e_k) = \sigma_t^G(b)\otimes e_k = u(b\otimes e_k)u^* = u_kbu_k^*\otimes e_k$. Hence, $\sigma_t^G = \mathrm{Ad}(u_1)$ which implies that $t \in T'$. Note that we could also have computed the T-invariant by using the results of [50]. Let us now compute Connes' τ -invariant in the non-amalgamented case. It suffices to show that, for any sequence $(t_n)_n$ of real numbers, one has $t_n \to 0$ for $\tau(G \wr_* S_N^+)$ if and only if $t_n \to 0$ for $\tau(G)$. Assume that $t_n \to 0$ for $\tau(L^{\infty}(G \wr_* S_N^+))$. Then, viewing again $L^{\infty}(G \wr_* S_N^+) = (L^{\infty}(G) \otimes \mathbb{C}^N) *_{B} M_{N-1}$, where this time $B = \mathbb{C}^N = L_N$, we may apply [31, Theorem B], to deduce that there exists a sequence of unitaries $v_n \in \mathcal{U}(B)$ such that $\alpha_n := \mathrm{Ad}(v_n) \circ \sigma_{t_n} \to \mathrm{id}$. To conclude the proof, we apply the restriction argument from Remark 2.1 to deduce that $\alpha_n|_{L^{\infty}(G)\otimes\mathbb{C}^N}\to \mathrm{id}$ in $\mathrm{Aut}(\mathrm{L}^{\infty}(G)\otimes\mathbb{C}^N)$. Note that, since B is in the center of $L^{\infty}(G) \otimes \mathbb{C}^N$ one has $\alpha_n|_{L^{\infty}(G) \otimes \mathbb{C}^N} = \sigma_{t_n}^G \otimes id$ and we deduce that $\sigma_{t_n}^G \to \text{id i.e. } t_n \to 0 \text{ in } \tau(G)$. The converse statement follows from the continuity of the group homomorphism ψ defined by Equation (2). \square

Remark 5.2. The assumption $N \neq 2$ in Theorem B is necessary. Indeed, when N = 2, we know from Proposition 2.23 and Section 2.5 that $L^{\infty}(G \wr_* S_2^+) = (L^{\infty}(G^{*2})) \otimes \mathbb{C}^2$. In particular, $L^{\infty}(G \wr_* S_2^+)$ is never a factor. For N = 3, our proof does not work $(L^{\infty}(S_3^+) \simeq \mathbb{C}^6$ is not diffuse) but we don't know if $L^{\infty}(G \wr_* S_3^+)$ can nonetheless be a factor. However, whenever $|\operatorname{Irr}(G)| \geq 3$ if N = 2 or whenever G is non-trivial if N = 3, Proposition 2.8 shows that $L^{\infty}(G^{*N})$ is a full and prime factor without any Cartan subalgebras and of type II_1 if G is Kac or of type III_{λ} , $\lambda \neq 0$, with T invariant given by $\{t \in \mathbb{R} : \sigma_t^G = \operatorname{id}\}$, if G is not Kac. Finally, let us mention that the assumption $|\operatorname{Irr}(G)| = \infty$ in Theorem B does not seem necessary but is useful in our proof.

Theorem C is a direct consequence of the following.

Theorem 5.3. Suppose that Irr(G) is infinite, Irr(H) is finite and $N \geq 2$. The following holds.

- (1) If $L^{\infty}(G)$ is amenable then, $\forall 1 \leq i \leq N$, the von Neumann subalgebra of $L^{\infty}(G)_{*,H}$ S_N^+ generated by $\{\nu_i(a)u_{ij} : a \in L^{\infty}(G), 1 \leq j \leq N\}$ is maximal amenable with expectation.
- (2) If G is Kac then $L^{\infty}(H) \otimes L^{\infty}(S_4^+) \simeq (\nu(L^{\infty}(H)) \cup L^{\infty}(S_4^+))'' \subset L^{\infty}(G \wr_* S_4^+)$ is maximal amenable.
- **Proof.** (1). We use Serre's dévissage in the following way. Fix $1 \leq i \leq N$ and Let \mathcal{T}'_N be the graph obtained from \mathcal{T}_N by removing the edge v_i as well as its inverse edge \overline{v}_i . This graph is still connected. Let us denote by M_2 the fundamental von Neumann algebra of our graph of von Neumann algebras restricted to \mathcal{T}'_N so that we have $L^{\infty}(H) \otimes L^{\infty}(S_N^+) \subset M_2$. It follows from [22] that $L^{\infty}(G \wr_{*,H} S_N^+)$ is canonically isomorphic to $M_1 \underset{B}{*} M_2$, where the amalgamation is $B = L^{\infty}(H) \otimes \mathbb{C}^N \subset M_1 := L^{\infty}(G) \otimes \mathbb{C}^N$ and $B = L^{\infty}(H) \otimes L_i \subset L^{\infty}(H) \otimes L^{\infty}(S_N^+) \subset M_2$. Note that the von Neumann algebra generated by $\{\nu_i(a)u_{ij}: a \in C(G), 1 \leq j \leq N\}$ is then identified with $M_1 \subset M_1 \underset{B}{*} M_2$ and $M_1 \simeq L^{\infty}(G) \otimes \mathbb{C}^N \simeq L^{\infty}(G \times \widehat{\mathbb{Z}/N\mathbb{Z}})$ is diffuse, by Lemma 2.4. Hence $M_1 \not\prec_{M_1} B$ and since M_1 is amenable, we may apply [5, Main Theorem] to deduce that M_1 is maximal amenable.
- (2). Consider the inductive construction $L^{\infty}(H) \otimes L^{\infty}(S_4^+) = M_0 \subset M_1 \subset \cdots \subset M_N = L^{\infty}(G \wr_{*,H} S_4^+)$, where $M_{k+1} = \left(L^{\infty}(G) \otimes \mathbb{C}^N\right) \underset{L^{\infty}(H) \otimes \mathbb{C}^N}{*} M_k$. Recall that $L^{\infty}(S_4^+)$ is diffuse and amenable and so is $L^{\infty}(H) \otimes L^{\infty}(S_4^+) = M_0$. Let $M_0 \subset Q \subset M_N$ be an amenable von Neumann algebra. Then $M_0 \subset Q \cap M_{N-1}$ so $Q \cap M_{N-1} \not\prec L^{\infty}(H) \otimes \mathbb{C}^N$ since M_0 is diffuse. Hence, we can apply [5, Main Theorem] to the amalgamated free

product $M_N = \left(L^{\infty}(G) \otimes \mathbb{C}^N \right) \underset{L^{\infty}(H) \otimes \mathbb{C}^N}{*} M_{N-1}$ and deduce that $Q \subset M_{N-1}$. By a direct induction we find that $Q \subset M_0$. Note that the Kac assumption on G is to ensure that each M_k is finite so the inclusion $Q \cap M_k \subset M_k$ is always with expectation. We do not know if this result still holds in the non Kac case. \square

6. Free wreath product of a fundamental quantum group

Let $(\mathcal{G}, (G_q)_{q \in V(\mathcal{G})}, (G_e)_{e \in E(\mathcal{G})}, (s_e)_{e \in E(\mathcal{G})})$ be a graph of CQG over the connected graph \mathcal{G} i.e.:

- For every $q \in V(\mathcal{G})$ and every $e \in E(\mathcal{G})$, G_q and G_e are CQG.
- For all $e \in E(\mathcal{G})$, $G_{\overline{e}} = G_e$.
- For every $e \in E(\mathcal{G})$, $s_e : C(G_e) \to C(G_{s(e)})$ is a faithful unital *-homomorphism intertwining the comultiplications (so G_e is a dual quantum subgroup of $G_{s(e)}$).

Consider the graph of C^* -algebras $(\mathcal{G}, C(G_q), C(G_e), s_e)$ and fix a maximal subtree $\mathcal{T} \subset \mathcal{G}$. Define C(G) as the maximal fundamental C^* -algebra of the graph of C^* -algebras $(\mathcal{G}, C(G_q), C(G_e), s_e)$ relative to the maximal subtree \mathcal{T} . By the universal property of C(G), there exists a unique unital *-homomorphism $\Delta: C(G) \to C(G) \otimes C(G)$ such that $\Delta|_{C(G_q)} = \Delta_{C(G_q)}$ and $\Delta(u_e) = u_e \otimes u_e$ for all $q \in V(\mathcal{G})$ and all $e \in E(\mathcal{G})$. The pair $G:=(C(G),\Delta)$ is a CQG, known as the fundamental quantum group and studied in [22]. We will denote this CQG by $G=\pi_1(\mathcal{G},G_p,G_e,\mathcal{T})$. It is known from [22] that C(G) is indeed the full C^* -algebra of G. Let us note that, since s_e identifies G_e as a dual quantum subgroup of $G_{s(e)}$, we have conditional expectations $E_e^s: C(G_{s(e)}) \to s_e(C(G_e))$ which are non-necessary GNS-faithful.

Since $G_e \subset G_{s(e)}$ is a dual quantum subgroup s_e factorizes to a faithful unital *-homomorphism $s_e: C_r(G_e) \to C_r(G_{s(e)})$ intertwining the comultiplications and we have Haar-state-preserving faithful conditional expectations $C_r(G_{s(e)}) \to s_e(C_r(G_e))$. Hence we get a graph of C*-algebras $(\mathcal{G}, C_r(G_q), C_r(G_e), s_e)$ with faithful conditional expectations whose vertex-reduced fundamental C*-algebra is the reduced C*-algebra $C_r(G)$ of G [22].

Definition 6.1. The *loop subgroup* of G is the group $\Gamma \subset \mathcal{U}(C(G))$ generated by $\{u_e : e \in E(\mathcal{G})\}$.

Note that the inclusion $\Gamma \subset \mathcal{U}(C(G))$ extends to a unital *-homomorphism $\pi : C^*(\Gamma) \to C(G)$.

Proposition 6.2. For $G = \pi_1(\mathcal{G}, G_p, G_e, \mathcal{T})$ with loop subgroup Γ the following holds.

(1) $\pi: C^*(\Gamma) \to C(G)$ is faithful and intertwines the comultiplications.

(2) For all $e \in E(\mathcal{G})$, $N \in \mathbb{N}^*$, there exists a unique unital *-homomorphism

$$s_e^N: C\left(G_e \wr_* S_N^+\right) \to C\left(G_{s(e)} \wr_* S_N^+\right) \text{ such that } s_e^N \circ \nu_i = \nu_i \circ s_e, \ s_e^N|_{C(S_N^+)} = \mathrm{id}, \ \forall i.$$

Moreover, s_e^N intertwines the comultiplications and restricts to a faithful map $\operatorname{Pol}(G_e \wr_* S_N^+) \hookrightarrow \operatorname{Pol}(G_{s(e)} \wr_* S_N^+)$ i.e. $G_e \wr_* S_N^+$ is a dual quantum subgroup of $G_{s(e)} \wr_* S_N^+$.

Proof. (1). Let C_{Γ} be the image of π i.e. the C*-algebra generated by $\{u_e : e \in E(\mathcal{G})\}$ in C(G). To show that that π is faithful, it suffices to check that C_{Γ} satisfies the universal property of $C^*(\Gamma)$. Recall that $\varepsilon_{s(e)} \circ s_e(b) = \varepsilon_e(b)$, where, for $p \in V(\mathcal{G})$, ε_p is the counit on $C(G_p)$ and, for $e \in E(\mathcal{G})$, ε_e is the counit on $C(G_e)$. Let now $\rho: \Gamma \to \mathcal{U}(H)$ be a unitary representation of Γ . By the universal property of $C(G) = \pi_1(\mathcal{G}, C(G_p), C(G_e), \mathcal{T})$, there exists a unique unital *-homomorphism $\widetilde{\rho}: C(G) \to \mathcal{B}(H)$ such that $\widetilde{\rho}(u_e) = \rho(u_e)$ and $\widetilde{\rho}(a) = \varepsilon_p(a) \mathrm{id}_H$, for all $a \in C(G_p)$ and all $p \in V(\mathcal{G})$. Hence, $\widetilde{\rho}|_{C_\Gamma}: C_\Gamma \to \mathcal{B}(H)$ is a unital *-homomorphism such that $\widetilde{\rho}(g) = \rho(g)$, for all $g \in \Gamma$. It follows that π is faithful. The fact that π intertwines the comultiplications follows from the equality $\Delta(u_e) = u_e \otimes u_e$ for all $e \in E(\mathcal{G})$.

(2). The existence and the properties of the morphism s_e^N follows from Proposition 3.3. \square

By the previous Proposition, we will always view $C^*(\Gamma) = C(\widehat{\Gamma}) \subset C(G)$ as the C*-algebra generated by $\{u_e : e \in E(\mathcal{G})\}$ and such that the inclusion intertwines the comultiplications. In particular, $\widehat{\Gamma}$ is a dual quantum subgroup of G. Moreover, the previous Proposition shows that $(\mathcal{G}, (G_q \wr_* S_N^+)_{q \in V(\mathcal{G})}, (G_e \wr_* S_N^+)_{e \in E(\mathcal{G})}, (s_e^N)_{e \in E(\mathcal{G})})$ is a graph of quantum groups. Its fundamental quantum group is determined in the following Theorem.

Theorem 6.3. If $G = \pi_1(\mathcal{G}, G_p, G_e, \mathcal{T})$ with loop subgroup Γ then

$$G \wr_{*,\widehat{\Gamma}} S_N^+ \simeq \pi_1 \left(\mathcal{G}, G_p \wr_* S_N^+, G_e \wr_* S_N^+, \mathcal{T} \right).$$

Proof. Define $\mathcal{A} := C\left(\pi_1\left(\mathcal{G}, G_p \wr_* S_N^+, G_e \wr_* S_N^+, \mathcal{T}\right)\right)$ and $\mathcal{B} := C\left(G \wr_{*,\widehat{\Gamma}} S_N^+\right)$. For all $p \in V(\mathcal{G})$, we have a unital *-homomorphism $\varphi_p : C(G_p \wr_* S_N^+) \to \mathcal{B}$ coming from the inclusion $C(G_p) \subset C(G)$. Moreover, $\forall e \in E(\mathcal{G}) \backslash E(\mathcal{T})$, there is a unitary $u_e \in \mathcal{B}$ such that

$$u_e^* \varphi_{s(e)} \circ s_e(a) u_e = \varphi_{r(e)} \circ r_e(a) \quad \forall a \in C(G_e \wr_* S_N^+),$$

the unitary being the one in $C^*(\Gamma)$, which is unique because of the amalgamation and works for every copy of $C(G_e)$ in $C(G_e)$ in $C(G_e)$ by construction of G. Hence, the universal property of \mathcal{A} gives a morphism $\pi: \mathcal{A} \to \mathcal{B}$ which clearly intertwines the comultiplications. We will give an inverse of this map, using the universal property of \mathcal{B} . For $1 \leq i \leq N$, there is for any $p \in V(\mathcal{G})$ a map $\nu_i: C(G_p) \to \mathcal{A}$ sending $C(G_p)$ on its

i-th copy in the free product defining $C(G_p \wr_* S_N^+)$ in \mathcal{A} . There is also for every edge $e \in E(\mathcal{G}) \setminus E(\mathcal{T})$ a unitary $u_e \in \mathcal{A}$, which satisfies the relations of the universal property of \mathcal{B} , and is the same for every one of the N copies of $C(G_e)$. Therefore, we get a map $\psi_0 : C(G \wr_* S_N^+) \to \mathcal{A}$ which factors through \mathcal{B} , because the unitaries u_e are independent of the copy of $C(G_e)$, thus we can factor the map through the quotient amalgamating the algebras generated by the u_e^i for $1 \le i \le N$, which is exactly \mathcal{B} . We get a morphism $\psi : \mathcal{B} \to \mathcal{A}$ for which it is easy to see that it intertwines the comultiplications. The maps are inverse of each other, because they send the N copies of C(G) and respect the unitaries from the fundamental algebra construction. \square

Example 6.4. Suppose that \mathcal{G} has two edges e and \overline{e} . We have two cases.

(1) If $s(e) \neq r(e)$ then \mathcal{G} is a tree so $\mathcal{T} = \mathcal{G}$ and the loop subgroup is trivial. We are in the situation of an amalgamated free product $G = G_1 *_H G_2$ and, by Theorem 6.3,

$$G \wr_* S_N^+ \simeq G_1 \wr_* S_N^+ \underset{H \wr_* S_N^+}{*} G_2 \wr_* S_N^+.$$

(2) If s(e) = r(e) then \mathcal{T} has no edges and the loop subgroup is $\Gamma = \langle u_e \rangle \simeq \mathbb{Z}$. We are in the situation of an HNN extension i.e. H and Σ are CQG such that $C(\Sigma) \subset C(H)$ is a dual quantum subgroup and $\theta : C(\Sigma) \hookrightarrow C(H)$ is a faithful unital *-homomorphism intertwining the comultiplications then the HNN extension $G := \text{HNN}(H, \Sigma, \theta)$ is the CQG with C(G) the universal unital C*-algebra generated by C(H) and a unitary $u \in C(G)$ with the relations $\theta(b) = ubu^*$ for all $b \in C(\Sigma) \subset C(H)$ and comultiplication given by $\Delta(u) = u \otimes u$ and $\Delta|_{C(H)} = \Delta_H$. The loop subgroup $\Gamma = \mathbb{Z}$ satisfies $C^*(\Gamma) = \langle u \rangle \subset C(G)$ and, by Theorem 6.3, $\text{HNN}(H, \Sigma, \theta) \wr_{*,\hat{\Gamma}} S_N^+ \simeq \text{HNN}(H \wr_* S_N^+, \Sigma \wr_* S_N^+, \tilde{\theta})$.

7. K-theory

In [23] is obtained a 6-term exact sequences for the KK-theory of the reduced and the full fundamental algebras of any graph of C^* -algebras $(\mathcal{G}, A_p, B_e, s_e)$ with (non-necessary GNS-faithful) conditional expectations. It is shown in [23] that the canonical surjection from the full fundamental C^* -algebra P to the vertex-reduced fundamental C^* -algebra P_r is a KK-equivalence and, denoting by P_{\bullet} either P or P_r one has the following exact sequences, for any C^* -algebra C.

$$\bigoplus_{e \in E^{+}(\mathcal{G})} KK^{0}(C, B_{e}) \xrightarrow{\sum s_{e}^{*} - r_{e}^{*}} \bigoplus_{p \in V(\mathcal{G})} KK^{0}(C, A_{p}) \longrightarrow KK^{0}(C, P_{\bullet})$$

$$\downarrow \qquad \qquad \downarrow \qquad \qquad \downarrow$$

$$KK^{1}(C, P_{\bullet}) \longleftarrow \bigoplus_{p \in V(\mathcal{G})} KK^{1}(C, A_{p}) \xrightarrow{\sum s_{e}^{*} - r_{e}^{*}} \bigoplus_{e \in E^{+}(\mathcal{G})} KK^{1}(C, B_{e}),$$

$$\bigoplus_{e \in E^{+}(\mathcal{G})} KK^{0}(B_{e}, C) \underset{s_{e_{*}} - r_{e_{*}}}{\longleftarrow} \bigoplus_{p \in V(\mathcal{G})} KK^{0}(A_{p}, C) \longleftarrow KK^{0}(P_{\bullet}, C)$$

$$\downarrow \qquad \qquad \uparrow \qquad \qquad \downarrow$$

$$KK^{1}(P_{\bullet}, C) \longrightarrow \bigoplus_{p \in V(\mathcal{G})} KK^{1}(A_{p}, C) \xrightarrow{s_{e_{*}} - r_{e_{*}}} \bigoplus_{e \in E^{+}(\mathcal{G})} KK^{1}(B_{e}, C). \tag{4}$$

Recall that, given a CQG G, $C_{\bullet}(G)$ denotes either C(G) or $C_r(G)$.

Proof of Theorem D. We can use the exact sequence (3) with $C = \mathbb{C}$ and with the graph of C*-algebra over \mathcal{T}_N from Section 3.1 in the case of C(G) and from Section 3.2 in the case of $C_r(G)$ since both graphs of C^* -algebras have injective connecting maps and conditional expectations.

$$\bigoplus_{v \in E^{+}} K_{0}(C_{\bullet}(H) \otimes \mathbb{C}^{N}) \xrightarrow{\sum s_{v}^{*} - r_{v}^{*}} \bigoplus_{p \in V \setminus \{p_{0}\}} K_{0}(C_{\bullet}(G) \otimes \mathbb{C}^{N}) \oplus K_{0}(C_{\bullet}(S_{N}^{+})) \xrightarrow{} K_{0}(C_{\bullet}(G)_{*,H} S_{N}^{+}))$$

$$\downarrow \qquad \qquad \downarrow \qquad \downarrow \qquad \qquad \qquad \downarrow \qquad$$

Using the compatibility of the K-theory with direct sums, we get the following:

$$K_0(C_{\bullet}(H)) \otimes \mathbb{Z}^{N^2} \xrightarrow{\sum s_v^* - r_v^*} \left(K_0(C_{\bullet}(G)) \otimes \mathbb{Z}^{N^2} \right) \oplus K_0(C_{\bullet}(S_N^+)) \longrightarrow K_0(C_{\bullet}(G \wr_{*,H} S_N^+))$$

$$\downarrow \qquad \qquad \downarrow$$

$$K_1(C_{\bullet}(G \wr_{*,H} S_N^+)) \longleftarrow \left(K_1(C_{\bullet}(G)) \otimes \mathbb{Z}^{N^2} \right) \oplus K_1(C_{\bullet}(S_N^+)) \xrightarrow{\sum s_v^* - r_v^*} K_1(C_{\bullet}(H)) \otimes \mathbb{Z}^{N^2}.$$

In what follows we restrict to the cases where H is the trivial group, so that the maps $\sum s_v^* - r_v^*$ are always injective, as shown just below. We will however come back to the above sequence in some special cases. Let $e_j \in \mathbb{C}^N$, $1 \le j \le N$ the canonical basis made of pairwise orthogonal rank one projections so that $\{[e_j] : 1 \le j \le N\}$ is a basis of $K_0(\mathbb{C}^N) \simeq \mathbb{Z}^N$. One checks easily that $\{[1 \otimes e_j] : 1 \le j \le N\}$ is linearly independent in $K_0(C_{\bullet}(G) \otimes \mathbb{C}^N)$ hence for every $1 \le i \le N$, the elements

$$[1 \otimes e_i] - [u_{ii}] \in K_0((C(G) \otimes \mathbb{C}^N)) \oplus K_0(C_{\bullet}(S_N^+)), \ 1 \leq j \leq N$$

are also linearly independent and the map

$$s_{v_i}^* - r_{v_i}^* : K_0(\mathbb{C}^N) = \mathbb{Z}^N \to K_0((C(G) \otimes \mathbb{C}^N)) \oplus K_0(C_{\bullet}(S_N^+)),$$

$$(s_{v_i}^* - r_{v_i}^*)[e_j] = [1 \otimes e_j] - [u_{ji}]$$

is injective, for all $1 \leq i \leq N$. Now, let us denote by $[e_{ij}] \in (\mathbb{Z}^N)^{\oplus N}$ the element $[e_j]$ viewed in the i^{th} -copy of \mathbb{Z}^N so that $\{[e_{ij}]: 1 \leq i, j \leq N\}$ is a basis of $(\mathbb{Z}^N)^{\oplus N}$ and, similarly, we denote by $[1 \otimes e_{ij}] \in K_0(C_{\bullet}(G) \otimes \mathbb{C}^N)^{\oplus N}$ the element $[1 \otimes e_j]$ viewed in

the i^{th} copy of $K_0(C_{\bullet}(G) \otimes \mathbb{C}^N)$ so that the map $\psi := \sum_{v \in E^+} s_v^* - r_v^* : (\mathbb{Z}^N)^{\oplus N} \to K_0(C_{\bullet}(G) \otimes \mathbb{C}^N)^{\oplus N} \bigoplus K_0(C_{\bullet}(S_N^+))$ is given by $\psi([e_{ij}]) = [1 \otimes e_{ij}] - [u_{ji}]$. As before, since $\{[1 \otimes e_{ij}] : 1 \leq i, j \leq N\}$ is linearly independent in $K_0(C_{\bullet}(G) \otimes \mathbb{C}^N)^{\oplus N}$, it follows that $\{[1 \otimes e_{ij}] - [u_{ji}] : 1 \leq i, j \leq N\}$ is also linearly independent in $K_0(C_{\bullet}(G) \otimes \mathbb{C}^N)^{\oplus N}$ $\bigoplus K_0(C_{\bullet}(S_N^+))$ so ψ is injective.

From this we get the isomorphism for K_1 and, for K_0 , we have:

$$K_0(C_{\bullet}(G \wr_* S_N^+)) \simeq (K_0(C_{\bullet}(G) \otimes \mathbb{C}^N)^{\oplus N} \oplus K_0(C_{\bullet}(S_N^+))) / \operatorname{Im}(\psi)$$

The K-theory of $C_{\bullet}(S_N^+)$, where the two algebras are KK-equivalent because $\widehat{S_N^+}$ is K-amenable, is computed by Voigt in [54] for $N \geq 4$ and, for $1 \leq N \leq 3$ we have $C_{\bullet}(S_N^+) = C(S_N) = \mathbb{C}^{N!}$ hence,

$$K_0(C_{\bullet}(S_N^+)) \simeq \begin{cases} \mathbb{Z}^{N^2-2N+2} & \text{if } N \geq 4, \\ \mathbb{Z}^{N!} & \text{if } 1 \leq N \leq 3. \end{cases} \quad \text{and} \quad K_1(C_{\bullet}(S_N^+)) \simeq \begin{cases} \mathbb{Z} & \text{if } N \geq 4, \\ 0 & \text{if } 1 \leq N \leq 3. \end{cases}$$

When $N \geq 4$, a family of generators of $K_0(C_{\bullet}(S_N^+))$ is given by the classes of $(N-1)^2$ coefficients of the fundamental representation of S_N^+ , $[u_{ij}]$, $1 \leq i, j \leq N-1$, and the class [1] of the unit.

$$K_{0}(C_{\bullet}(G \wr_{*} S_{N}^{+}))$$

$$\simeq \left(K_{0}(C_{\bullet}(G) \otimes \mathbb{C}^{N})\right)^{\oplus N} \oplus K_{0}(C_{\bullet}(S_{N}^{+})) / \operatorname{Im}(\psi)$$

$$\simeq \left(K_{0}(C_{\bullet}(G)) \otimes \mathbb{Z}^{N^{2}}\right) \oplus \mathbb{Z}^{N^{2}-2N+2} / \langle ([1] \otimes [e_{ij}]) - [u_{ji}], 1 \leq i, j \leq N \rangle$$

$$\simeq \left(K_{0}(C_{\bullet}(G)) \otimes \mathbb{Z}^{N^{2}}\right) / ([1] \otimes \mathbb{Z}^{2N-2}).$$

The last quotient is obtained after identifying the classes $[1 \otimes e_{ij}]$ with $[u_{ji}]$. This gives the first part of the statement of Theorem D for $N \geq 4$. For the computation of the K-theory of quantum reflection groups $H_N^{s+} = \widehat{\mathbb{Z}}_s \wr_* S_N^+$, for $1 \leq s \leq +\infty$, which have K-amenable duals, note that, for $s < \infty$, since $C^*(\mathbb{Z}_s) \simeq C(\widehat{\mathbb{Z}}_s) \simeq \mathbb{C}^s$, we have $K_0(C^*(\mathbb{Z}_s)) \simeq \mathbb{Z}^s$ and $K_1(C^*(\mathbb{Z}_s)) \simeq \{0\}$. Hence,

$$K_0(C_{\bullet}(H_N^{s+})) \simeq \left(K_0(C^*(\mathbb{Z}_s)) \otimes \mathbb{Z}^{N^2}\right) / ([1] \otimes \mathbb{Z}^{2N-2}) \simeq (\mathbb{Z}^s \otimes \mathbb{Z}^{N^2}) / ([1] \otimes \mathbb{Z}^{2N-2})$$
$$\simeq \mathbb{Z}^{sN^2 - 2N + 2}.$$

The same computation works when $s = +\infty$ with $K_0(C^*(\mathbb{Z})) \simeq \mathbb{Z}$, for K_0 , but for K_1 , as $K_0(C^*(\mathbb{Z})) \simeq \mathbb{Z}$. \square

We are grateful to Adam Skalski for suggesting to us to use the uniqueness of the trace from [36] to deduce the following from Theorem D.

Corollary 7.1. If $M, N \geq 8$ and $s, t \geq 1$, then $C_r(H_N^{s+}) \simeq C_r(H_M^{t+}) \Leftrightarrow (N, s) = (M, t)$.

Proof. The dimensions of the K-theory groups may happen to be equal for different pairs of integers, so we need to go a bit further to differentiate between them. We use the value of the Haar state applied to the generators of $K_0(C_r(H_N^{s+})) \simeq \mathbb{Z}^{sN^2-2N+2}$. Thanks to the computation, we know that there are, in addition to the class of the unit [1], $N^2 - 2N + 1$ generators coming from the ones of $K_0(C_r(S_N^+))$, namely $[1 \otimes u_{ij}]$ for $1 \leq i, j \leq N-1$, they are equal to $[u_{ij}]$ in $K_0(C_r(H_N^{s+}))$. The trace of such elements is the same as the trace of the corresponding element in $C_r(S_N^+)$, thanks to 3.2, which is equal to 1/N. The $(s-1)N^2$ remaining generators are the ones coming from the N copies of $K_0(C^*(\mathbb{Z}_s) \otimes \mathbb{C}^N)$ and are of the form $[\delta_k \otimes e_j]$, for $k \in \mathbb{Z}_s$, $k \neq 0$, and $1 \leq j \leq N$. Such a class, in the i-th copy, is sent to $[\nu_i(\delta_k)u_{ij}]$, which is of trace 1/(sN). Thus, using the uniqueness of the trace of these algebras, as proved in [36], we get that the pair (N,s) can be retrieved from the data of the K-theory, and it allows to discriminate the algebras of the different quantum reflection groups. \square

Corollary 7.2. For all $m \geq 1$, and $N \geq 4$, the dual of $\widehat{\mathbb{F}}_m \wr_* S_N^+$ is K-amenable and we have:

$$K_0(C_{\bullet}(\widehat{\mathbb{F}}_m \wr_* S_N^+)) \simeq \mathbb{Z}^{N^2 - 2N + 2}$$
 and $K_1(C_{\bullet}(\widehat{\mathbb{F}}_m \wr_* S_N^+)) \simeq \mathbb{Z}^{N^2 m + 1}$.

If $1 \leq N \leq 3$, then the dual of $\widehat{\mathbb{F}}_m \wr_* S_N^+$ is also K-amenable, with

$$K_0(C_{\bullet}(\widehat{\mathbb{F}}_m \wr_* S_N^+)) \simeq \mathbb{Z}^{N!}$$
 and $K_1(C_{\bullet}(\widehat{\mathbb{F}}_m \wr_* S_N^+)) \simeq \mathbb{Z}^{N^2m}$.

In particular, for all $n, m \ge 1$ and $N, M \ge 1$, $C_{\bullet}(\widehat{\mathbb{F}}_n \wr_* S_N^+) \simeq C_{\bullet}(\widehat{\mathbb{F}}_m \wr_* S_M^+) \Leftrightarrow (n, N) = (m, M)$.

Proof. K-amenability of free groups has been proved by Cuntz in [17], when he introduced the notion of K-amenability for discrete groups. The K-theory for the maximal C*-algebra was initially computed in [16] and for the reduced C*-algebra in [40]. The result is $K_0(C^*(\mathbb{F}_m)) \simeq \mathbb{Z}$ and $K_1(C^*(\mathbb{F}_m)) \simeq \mathbb{Z}^m$. Using this and Theorem D, we get the first result. For the last statement, we first use equality of the K_0 -groups to deduce that N = M and then, equality of the K_1 -groups to deduce that n = m. \square

We can use the results of Section 6 to compute the KK-theory of C^* -algebras of a free wreath product of a fundamental quantum group of graph of CQG. The main theorem is the following, using the notations of Section 6 and denoting by E^+ and V the positive edges and vertices of the connected graph \mathcal{G} .

Theorem 7.3. If $G = \pi_1(\mathcal{G}, G_p, G_e, \mathcal{T})$ for any C^* -algebra A, there is a cyclic exact sequences:

Proof. As observed in Section 6 we have, at the level of full as well as reduced C*-algebras, a graph of C*-algebras $(\mathcal{G}, C_{\bullet}(G_q \wr_* S_N^+), C_{\bullet}(G_q \wr_* S_N^+), s_E^N)$ with conditional expectations (which are GNS-faithful only at the reduced level). By Theorem 6.3 the full/reduced C*-algebra $C_{\bullet}(G \wr_{*,\widehat{\Gamma}} S_N^+)$ is the full/vertex-reduced fundamental algebra of $(\mathcal{G}, C_{\bullet}(G_q \wr_* S_N^+), C_{\bullet}(G_r \wr_* S_N^+), s_E^N)$. Hence, we may apply the exact sequences (3) and (4). \square

We compute now K-theory groups of some free wreath products of an amalgamated free product with S_N^+ .

Corollary 7.4. If $G = G_1 *_H G_2$ then there is a cyclic sequence of K-theory groups:

$$K_0(C_{\bullet}(H \wr_* S_N^+)) \longrightarrow K_0(C_{\bullet}(G_1 \wr_* S_N^+)) \oplus K_0(C_{\bullet}(G_2 \wr_* S_N^+)) \longrightarrow K_0(C_{\bullet}(G \wr_* S_N^+))$$

$$\uparrow \qquad \qquad \downarrow$$

$$K_1(C_{\bullet}(G \wr_* S_N^+)) \longleftarrow K_1(C_{\bullet}(G_1 \wr_* S_N^+)) \oplus K_1(C_{\bullet}(G_2 \wr_* S_N^+)) \longleftarrow K_1(C_{\bullet}(H \wr_* S_N^+)).$$

Proof. As observed in Example 6.4 the loop subgroup is trivial in the case of an amalgamated free product. The proof follows from the first exact sequence in Theorem 7.3 with $A = \mathbb{C}$. \square

Corollary 7.5. For the compact quantum group $\widehat{\operatorname{SL}_2(\mathbb{Z})} \wr_* S_N^+$, which has K-amenable dual, we have for $N \geq 4$:

$$K_0(C_{\bullet}(\widehat{\operatorname{SL}_2(\mathbb{Z})}\wr_*S_N^+)) \simeq \mathbb{Z}^{8N^2-2N+2} \quad and \quad K_1(C_{\bullet}(\widehat{\operatorname{SL}_2(\mathbb{Z})}\wr_*S_N^+)) \simeq \mathbb{Z}.$$

For $1 \leq N \leq 3$, we have:

$$K_0(\widehat{\mathrm{CL}_2(\mathbb{Z})}\wr_*S_N^+)) \simeq \begin{cases} \mathbb{Z}^{69} & \text{if } N = 3, \\ \mathbb{Z}^{8N^2 - 2N + 2} & \text{if } N \in \{1, 2\}. \end{cases} \quad and \quad K_1(\widehat{\mathrm{CL}_2(\mathbb{Z})}\wr_*S_N^+)) \simeq 0.$$

Proof. We use the well known isomorphism $\mathrm{SL}_2(\mathbb{Z}) \simeq \mathbb{Z}_6 *_{\mathbb{Z}_2} \mathbb{Z}_4$ which implies the K-amenability of $\mathrm{SL}_2(\mathbb{Z})$ hence of the dual of $\widehat{\mathrm{SL}_2(\mathbb{Z})} \wr_* S_N^+$ as well by Theorem A. Moreover,

by Example 6.4, we have $\widehat{\operatorname{SL}_2(\mathbb{Z})} \wr_* S_N^+ \simeq H_N^{6+} *_{H_N^{2+}} H_N^{4+}$. We may use the K-theory of the quantum reflection groups computed in Theorem D to deduce the one for the group $\widehat{\operatorname{SL}_2(\mathbb{Z})} \wr_* S_N^+$. Applying Corollary 7.4 and K-amenability, we get the following cyclic exact sequence

$$K_0(C_{\bullet}(H_N^{2+})) \xrightarrow{\psi} K_0(C_{\bullet}(H_N^{6+})) \oplus K_0(C_{\bullet}(H_N^{4+})) \longrightarrow K_0(\widehat{\operatorname{SL}_2(\mathbb{Z})} \wr_* S_N^+))$$

$$\downarrow \qquad \qquad \downarrow \qquad \qquad \downarrow$$

$$K_1(C_{\bullet}(\widehat{\operatorname{SL}_2(\mathbb{Z})} \wr_* S_N^+)) \longleftarrow K_1(C_{\bullet}(H_N^{6+})) \oplus K_1(C_{\bullet}(H_N^{4+})) \longleftarrow K_1(C_{\bullet}(H_N^{2+})).$$

The maps in the bottom line are the diagonal embedding and the map $(x, y) \mapsto x - y$, which are respectively injective and surjective $\mathbb{Z} \to \mathbb{Z} \oplus \mathbb{Z} \to \mathbb{Z}$. Hence, the top line is a short exact sequence and the map ψ is injective, and we have

$$K_0(\widehat{\operatorname{C}_{\bullet}(\operatorname{SL}_2(\mathbb{Z})}\wr_*S_N^+)) \simeq \mathbb{Z}^{6N^2-2N+2} \oplus \mathbb{Z}^{4N^2-2N+2}/\operatorname{Im}(\psi).$$

The map ψ sends the generators of $K_0(C_{\bullet}(H_N^{2+}))$, coming from the classes of the form $[v \otimes e_j]$ in $K_0(C^*(\mathbb{Z}_2) \otimes \mathbb{C}^N)$ to the corresponding classes $[\beta^*v \otimes e_j] - [\alpha^*v \otimes e_j]$, where $\beta: \mathbb{Z}_2 \to \mathbb{Z}_6$ and $\alpha: \mathbb{Z}_2 \to \mathbb{Z}_4$ are the canonical embeddings. Hence, the quotient by $\text{Im}(\psi)$ identifies free copies of \mathbb{Z} and the group is isomorphic to \mathbb{Z}^{8N^2-2N+2} . \square

Remark 7.6. We can use Proposition 7.4 to give another proof of Corollary 7.2. It is also possible to compute first the K-theory of the amalgamated free products and then apply Theorem D to get the results of 7.4.

Let us now do a K-theory computation in the context of an HNN extension. Recall that the Baumslag-Solitar group BS(n,m), for $n,m \in \mathbb{Z}^*$, is defined by generators and relations:

$$BS(n,m) := \langle a, b \, | \, ab^n a^{-1} = b^m \rangle.$$

Let $\langle a \rangle < \mathrm{BS}(n,m)$ be the subgroup generated by a and view $\widehat{\langle a \rangle}$ as a dual quantum subgroup of the compact quantum group $\widehat{\mathrm{BS}(n,m)}$.

Proposition 7.7. The dual of the compact quantum group $\widehat{BS(n,m)} \wr_{*,\widehat{(a)}} S_N^+$ is K-amenable for any $n,m \in \mathbb{Z}^*$ and $N \geq 1$. Its K-theory groups are given by the following, for any $n,m \in \mathbb{Z}^*$: if $N \geq 4$ and n=m, then

$$K_0(C_{\bullet}(\widehat{\mathrm{BS}(n,m)}\wr_{*,\widehat{\langle a\rangle}}S_N^+)) \simeq \mathbb{Z}^{2N^2-2N+3} \ \ and \ \ K_1(C_{\bullet}(\widehat{\mathrm{BS}(n,m)}\wr_{*,\widehat{\langle a\rangle}}S_N^+)) \simeq \mathbb{Z}^{2N^2-2N+3},$$
and if $n \neq m$,

$$\begin{split} K_0(C_\bullet(\widehat{\mathrm{BS}(n,m)})_{{}^*,\widehat{\langle a\rangle}}S_N^+)) &\simeq \mathbb{Z}^{N^2-2N+3}, \\ K_1(C_\bullet(\widehat{\mathrm{BS}(n,m)})_{{}^*,\widehat{\langle a\rangle}}S_N^+)) &\simeq \mathbb{Z}^{N^2-2N+3} \oplus (\mathbb{Z}_{|n-m|})^{N^2}. \end{split}$$

For $1 \le N \le 3$, if n = m, then

$$K_0(C_{\bullet}(\widehat{\mathrm{BS}(n,m)})\wr_{*,\widehat{\langle a\rangle}}S_N^+)) \simeq \mathbb{Z}^{N^2+N!} \ and \ K_1(C_{\bullet}(\widehat{\mathrm{BS}(n,m)})\wr_{*,\widehat{\langle a\rangle}}S_N^+)) \simeq \mathbb{Z}^{N^2+N!}$$

and if $n \neq m$,

$$K_0(C_{\bullet}(\widehat{\mathrm{BS}(n,m)}\wr_{*,\widehat{\langle a\rangle}}S_N^+))\simeq \mathbb{Z}^{N!} \ \ and \ \ K_1(C_{\bullet}(\widehat{\mathrm{BS}(n,m)}\wr_{*,\widehat{\langle a\rangle}}S_N^+))\simeq \mathbb{Z}^{N!}\oplus (\mathbb{Z}_{\lfloor n-m\rfloor})^{N^2}.$$

Proof. Note that $\mathrm{BS}(n,m)$ is the HNN-extension $\mathrm{BS}(n,m) = \mathrm{HNN}(\mathbb{Z},\mathbb{Z},\theta_n,\theta_m)$, where $\theta_l:\mathbb{Z}\to\mathbb{Z}$ is the multiplication by $l\in\{n,m\}$. In particular, $\mathrm{BS}(n,m)$ is K-amenable and the CQG $\widehat{\mathrm{BS}(n,m)}$ is also an HNN-extension. We still denote by $\theta_l:C_{\bullet}(\widehat{\mathbb{Z}}\wr_*S_N^+)\to C_{\bullet}(\widehat{\mathbb{Z}}\wr_*S_N^+)$ the map defined by $\theta_l(\nu_i(k)u_{ij})=\nu_i(lk)u_{ij}$ for all $l\in\{n,m\},\ k\in\mathbb{Z}$ and $1\leq i,j\leq N$. By Example 6.4, the loop subgroup is $\Gamma=\langle a\rangle$ and, by using Theorem 7.3 (with $A=\mathbb{C}$), we get,

$$K_{0}\left(C_{\bullet}\left(\widehat{\mathbb{Z}}\wr_{*}S_{N}^{+}\right)\right) \xrightarrow{\theta_{n}^{*}-\theta_{m}^{*}} K_{0}\left(C_{\bullet}\left(\widehat{\mathbb{Z}}\wr_{*}S_{N}^{+}\right)\right) \longrightarrow K_{0}(\mathcal{B})$$

$$\downarrow \qquad \qquad \downarrow \qquad \qquad \downarrow$$

$$K_{1}(\mathcal{B}) \longleftarrow K_{1}\left(C_{\bullet}\left(\widehat{\mathbb{Z}}\wr_{*}S_{N}^{+}\right)\right) \xrightarrow{\theta_{n}^{*}-\theta_{m}^{*}} K_{1}\left(C_{\bullet}\left(\widehat{\mathbb{Z}}\wr_{*}S_{N}^{+}\right)\right)$$

where $\mathcal{B}:=C_{\bullet}(\widehat{\mathrm{BS}(n,m)}\wr_{*,\widehat{\langle a\rangle}}S_N^+)$. As the group $K_0(C_{\bullet}(\widehat{\mathbb{Z}}\wr_*S_N^+))$ is generated by the image of $K_0(C_{\bullet}(S_N^+))$ in it, and since we have, for $l\in\{n,m\}$, $\theta_l^*([u_{ij}])=[\theta_l(u_{ij})]=[u_{ij}]$ and $\theta_l^*([1])=[\theta_l(1)]=[1]$, where $[u_{ij}]$ and [1] are the generators of the K_0 group, the map $\theta_n^*-\theta_m^*$ is trivial at the K_0 level. It is however nontrivial at the K_1 level: $K_1(C(\widehat{\mathbb{Z}}\wr_*S_N^+))$ is generated by the images of the generator [1] of $K_1(C^*(\mathbb{Z}))$, the one of the generator of $K_1(C(S_N^+))$. The action of θ_n at the K-theory level is thus given by multiplication by n on the generators coming from $K_1(C^*(\mathbb{Z}))$, and acts trivially on the one coming from $K_1(C(S_N^+))$. Thus the sequence splits differently depending on if m and m are equal or not. If m=m, then the maps are trivial and the sequence splits in two short sequences as follows:

$$0 \to \mathbb{Z}^{N^2 - 2N + 2} \to K_0(\mathcal{B}) \to \mathbb{Z}^{N^2 + 1} \to 0 \quad \text{and}$$
$$0 \to \mathbb{Z}^{N^2 + 1} \to K_0(\mathcal{B}) \to \mathbb{Z}^{N^2 - 2N + 2} \to 0,$$

giving the first part of the result. If $m \neq n$ then the map is still trivial at the K_0 -level, but acts as the multiplication by m-n on the N first generators of $K_1(C(\widehat{\mathbb{Z}} \wr_* S_N^+))$ and as 0 on the last one. The sequence splits in two short sequences:

$$0 \to \mathbb{Z}^{N^2 - 2N + 2} \to K_0(\mathcal{B}) \to \mathbb{Z} \to 0$$
 and
$$0 \to \mathbb{Z}^{N^2} \xrightarrow{\psi} \mathbb{Z}^{N^2 + 1} \to K_1(\mathcal{B}) \to \mathbb{Z}^{N^2 - 2N + 2} \to 0,$$

the map ψ in the second being multiplication by (n-m) on each of the first N^2 terms of the sum, and the sequence becomes $0 \to (\mathbb{Z}_{|n-m|})^{N^2} \oplus \mathbb{Z} \to K_1(\mathcal{B}) \to \mathbb{Z}^{N^2-2N+2} \to 0$, and thus the result in the remaining case because the group on the right-hand-side is free. The proof for the cases $1 \le N \le 3$ works the same way. \square

Remark 7.8. The computation of this case can also be achieved thanks to the six-term exact sequence for the free wreath product with amalgamation of Theorem D, written in the beginning of the proof in section 7. The main point in the use of this exact sequence is that the maps $\sum s_v^* - r_v^*$ appearing in the sequence are injective and explicit in this special case, allowing an easy computation.

We can compare it to the non-amalgamated case, using the following proposition about the K-theory of the Baumslag-Solitar groups $BS(m,n) = HNN(\mathbb{Z}, \mathbb{Z}, \theta_m, \theta_n)$. We include a proof as we couldn't find the result stated except in the solvable case in [41].

Proposition 7.9. Let m and n be integers, then the Baumslag-Solitar group BS(m, n) is K-amenable and its K-theory is given, if n = m, by

$$K_0(C^*_{\bullet}(\mathrm{BS}(m,m))) \simeq \mathbb{Z}^2 \ and \ K_1(C^*_{\bullet}(\mathrm{BS}(m,m))) = \mathbb{Z}^2,$$

and if $n \neq m$, by

$$K_0(C^*_{\bullet}(\mathrm{BS}(m,n))) \simeq \mathbb{Z} \ and \ K_1(C^*_{\bullet}(\mathrm{BS}(m,n))) = \mathbb{Z} \oplus \mathbb{Z}_{|n-m|}.$$

Proof. The Baumslag-Solitar groups are the fundamental group of the graph with only one vertex and one edge, with \mathbb{Z} on each and with \mathbb{Z} being sent to the subgroups $n\mathbb{Z}$ and $m\mathbb{Z}$ by multiplication by m and n. Thus, the corresponding C*-algebras are the full and vertex-reduced fundamental C*-algebra of this graph with $C^*(\mathbb{Z})$ and the maps induced from the multiplications, which are K-equivalent. The exact sequence of K-theory is then

$$K_{0}\left(C^{*}(\mathbb{Z})\right) \xrightarrow{\theta_{m}^{*}-\theta_{n}^{*}} K_{0}\left(C^{*}(\mathbb{Z})\right) \longrightarrow K_{0}\left(C_{\bullet}^{*}(\mathrm{BS}(m,n))\right)$$

$$\downarrow \qquad \qquad \downarrow$$

$$K_{1}\left(C_{\bullet}^{*}(\mathrm{BS}(m,n))\right) \longleftarrow K_{1}\left(C^{*}(\mathbb{Z})\right) \xleftarrow{\theta_{m}^{*}-\theta_{n}^{*}} K_{1}\left(C^{*}(\mathbb{Z})\right),$$

which becomes

$$\mathbb{Z} \xrightarrow{0} \mathbb{Z} \longrightarrow K_0(C^*_{\bullet}(\mathrm{BS}(m,n)))$$

$$\uparrow \qquad \qquad \downarrow \qquad \downarrow$$

$$K_1(C^*_{\bullet}(\mathrm{BS}(m,n))) \longleftarrow \mathbb{Z} \longleftarrow (m-n) \mathbb{Z},$$

giving the result as the map (m-n) is injective if $m \neq n$ and 0 if m=n. \square

From this we can compute the K-theory of the free wreath products using Theorem D.

Proposition 7.10. Let m and n be integers and $N \geq 4$, then the dual of the compact quantum group $\widehat{BS(n,m)} \wr_* S_N^+$ is K-amenable and if n=m, then

$$K_0(C_{\bullet}(\widehat{\mathrm{BS}(m,n)}\wr_*S_N^+)) \simeq \mathbb{Z}^{2N^2-2N+2} \ and \ K_1(C_{\bullet}(\widehat{\mathrm{BS}(m,n)}\wr_*S_N^+)) = \mathbb{Z}^{2N^2+1}.$$

If $n \neq m$,

$$K_0(C_{\bullet}(\widehat{\mathrm{BS}(m,n)}\wr_*S_N^+)) \simeq \mathbb{Z}^{N^2-2N+2} \ and$$

 $K_1(C_{\bullet}(\widehat{\mathrm{BS}(m,n)}\wr_*S_N^+)) = \mathbb{Z}^{N^2+1} \oplus (\mathbb{Z}_{|n-m|})^{N^2}.$

For $1 \le N \le 3$, if n = m, then

$$K_0(C_{\bullet}(\widehat{\mathrm{BS}(m,n)}\wr_*S_N^+)) \simeq \mathbb{Z}^{N^2+N!} \ and \ K_1(\widehat{C_{\bullet}(\mathrm{BS}(m,n)}\wr_*S_N^+)) = \mathbb{Z}^{2N^2}.$$

If $n \neq m$,

$$K_0(\widehat{C_{\bullet}}(\widehat{\mathrm{BS}(m,n)}\wr_*S_N^+)) \simeq \mathbb{Z}^{N!} \ and \ K_1(\widehat{C_{\bullet}}(\widehat{\mathrm{BS}(m,n)}\wr_*S_N^+)) = \mathbb{Z}^{N^2} \oplus (\mathbb{Z}_{\lfloor n-m \rfloor})^{N^2}.$$

Acknowledgments

The authors would like to thank Kenny De Commer, Adam Skalski, Roland Vergnioux and Makoto Yamashita for their help and their suggestions during the redaction of this article.

References

- [1] T. Banica, Symmetries of a generic coaction, Math. Ann. 314 (4) (1999) 763–780.
- [2] M. Brannan, A. Chirvasitu, A. Freslon, Topological generation and matrix models for quantum reflection groups, Adv. Math. 363 (2020) 106982, 31.
- [3] M. Brannan, B. Collins, R. Vergnioux, The Connes embedding property for quantum group von Neumann algebras, Transl. Am. Math. Soc. 369 (6) (2017) 3799–3819.
- [4] N. Brown, K. Dykema, K. Jung, Free entropy dimension in amalgamated free products, Proc. Lond. Math. Soc. (3) 97 (2) (2008) 339–367.
- [5] R. Boutonnet, C. Houdayer, Amenable absorption in amalgamated free product von Neumann algebras, Kyoto J. Math. 58 (3) (2018) 583–593.
- [6] R. Boutonnet, C. Houdayer, S. Raum, Amalgamated free product type III factors with at most one Cartan subalgebra, Compos. Math. 150 (1) (2014) 143–174.

- [7] J. Bichon, Quantum automorphism groups of finite graphs, Proc. Am. Math. Soc. 131 (3) (2003) 665–673 (electronic).
- [8] J. Bichon, Free wreath product by the quantum permutation group, Algebr. Represent. Theory 7 (4) (2004) 343–362.
- [9] M. Brannan, Reduced operator algebras of trace-perserving quantum automorphism groups, Doc. Math. 18 (2013) 1349–1402.
- [10] T. Banica, R. Vergnioux, Fusion rules for quantum reflection groups, J. Noncommut. Geom. 3 (3) (2009) 327–359.
- [11] A. Chirvasitu, Cosemisimple Hopf algebras are faithfully flat over Hopf subalgebras, Algebra Number Theory 8 (5) (2014) 1179–1199.
- [12] M. Caspers, M. Klisse, A. Skalski, G. Vos, M. Wasilewski, Relative Haagerup property for arbitrary von Neumann algebras, Adv. Math. 421 (2023) 109017.
- [13] A. Connes, Une classification des facteurs de type III, Ann. Sci. Éc. Norm. Supér. (4) 6 (1973) 133-252.
- [14] A. Connes, Almost periodic states and factors of type III₁, J. Funct. Anal. 16 (1974) 415–445.
- [15] M. Caspers, A. Skalski, The Haagerup property for arbitrary von Neumann algebras, Int. Math. Res. Not. (19) (2015) 9857–9887.
- [16] J. Cuntz, The K-groups for free products of C*-algebras, in: Operator Algebras and Applications, Part 1, Kingston, Ont., 1980, in: Proc. Sympos. Pure Math., vol. 38, Amer. Math. Soc., Providence, R.I., 1982, pp. 81–84.
- [17] J. Cuntz, K-theoretic amenability for discrete groups, J. Reine Angew. Math. 344 (1983) 180–195.
- [18] K. De Commer, P. Kasprzak, A. Skalski, P. Soltan, Quantum actions on discrete quantum spaces and a generalization of Clifford's theory of representations, Isr. J. Math. 226 (1) (2018) 475–503.
- [19] M. Daws, P. Fima, A. Skalski, S. White, The Haagerup property for locally compact quantum groups, J. Reine Angew. Math. 711 (2016) 189–229.
- [20] V.G. Drinfel'd, Quantum groups, Zap. Nauchn. Sem. Leningrad. Otdel. Mat. Inst. Steklov. (LOMI), Differentsialnaya Geometriya, Gruppy Li i Mekh. VIII 155 (1986) 18–49, 193.
- [21] V.G. Drinfel'd, Quantum groups, in: Proceedings of the International Congress of Mathematicians, vol. 1, 2 (Berkeley, Calif., 1986), Amer. Math. Soc., Providence, RI, 1987, pp. 798–820.
- [22] P. Fima, A. Freslon, Graphs of quantum groups and K-amenability, Adv. Math. 260 (2014) 233–280.
- [23] P. Fima, E. Germain, The KK-theory of fundamental C*-algebras, Transl. Am. Math. Soc. 370 (10) (2018) 7051–7079.
- [24] P. Fima, On locally compact quantum groups whose algebras are factors, J. Funct. Anal. 244 (1) (2007) 78–94.
- [25] P. Fima, Kazhdan's property T for discrete quantum groups, Int. J. Math. 21 (01) (2010) 47–65.
- [26] A. Freslon, R. Martos, Torsion and K-theory for some free wreath products, Int. Math. Res. Not. (6) (2020) 1639–1670.
- [27] P. Fima, L. Pittau, The free wreath product of a compact quantum group by a quantum automorphism group, J. Funct. Anal. 271 (7) (2016) 1996–2043.
- [28] A. Freslon, Free wreath products with amalgamation, Commun. Algebra (2022) 1–23.
- [29] A. Freslon, A. Skalski, Wreath products of finite groups by quantum groups, J. Noncommut. Geom. 12 (1) (2018) 29–68.
- [30] D. Gromada, Quantum symmetries of Cayley graphs of abelian groups, Glasg. Math. J. 65 (3) (2023) 655–686.
- [31] C. Houdayer, Y. Isono, Factoriality, Connes' type III invariants and fullness of amalgamated free product von Neumann algebras, Proc. R. Soc. Edinb., Sect. A 150 (3) (2020) 1495–1532.
- [32] C. Houdayer, S. Vaes, Type III factors with unique Cartan decomposition, J. Math. Pures Appl. (9) 100 (4) (2013) 564–590.
- [33] M. Jimbo, A q-difference analogue of $U(\mathfrak{g})$ and the Yang-Baxter equation, Lett. Math. Phys. 10 (1) (1985) 63–69.
- [34] M. Kalantar, P. Kasprzak, A. Skalski, R. Vergnioux, Noncommutative Furstenberg boundary, Anal. PDE 15 (3) (2022) 795–842.
- [35] J. Krajczok, P.M. Sołtan, Examples of compact quantum groups with $L^{\infty}(G)$ a factor, arXiv preprint, arXiv:2203.10976, 2022.
- [36] F. Lemeux, The fusion rules of some free wreath product quantum groups and applications, J. Funct. Anal. 267 (7) (2014) 2507–2550.
- [37] F. Lemeux, P. Tarrago, Free wreath product quantum groups: the monoidal category, approximation properties and free probability, J. Funct. Anal. 270 (10) (2016) 3828–3883.

- [38] N. Ozawa, Examples of groups which are not weakly amenable, Kyoto J. Math. 52 (2) (2012) 333–344.
- [39] L. Pittau, The free wreath product of a discrete group by a quantum automorphism group, Proc. Am. Math. Soc. 144 (5) (2016) 1985–2001.
- [40] M. Pimsner, D. Voiculescu, K-groups of reduced crossed products by free groups, J. Oper. Theory 8 (1) (1982) 131–156.
- [41] S. Pooya, A. Valette, K-theory for the C*-algebras of the solvable Baumslag-Solitar groups, Glasg. Math. J. 60 (2) (2018) 481–486.
- [42] É. Ricard, Q. Xu, Khintchine type inequalities for reduced free products and applications, J. Reine Angew. Math. (2006) 27–59.
- [43] P. Salmi, Quantum semigroup compactifications and uniform continuity on locally compact quantum groups, Ill. J. Math. 54 (2) (2010) 469–483.
- [44] J.-P. Serre, Trees, Springer-Verlag, Berlin-New York, 1980, Translated by John Stillwell.
- [45] R. Tomatsu, Amenable discrete quantum groups, J. Math. Soc. Jpn. 58 (4) (2006) 949–964.
- [46] P. Tarrago, J. Wahl, Free wreath product quantum groups and standard invariants of subfactors, Adv. Math. 331 (2018) 1–57.
- [47] Y. Ueda, Amalgamated free product over Cartan subalgebra, Pac. J. Math. 191 (2) (1999) 359–392.
- [48] Y. Ueda, Factoriality, type classification and fullness for free product von Neumann algebras, Adv. Math. 228 (5) (2011) 2647–2671.
- [49] Y. Ueda, On type III₁ factors arising as free products, Math. Res. Lett. 18 (5) (2011) 909–920.
- [50] Y. Ueda, Some analysis of amalgamated free products of von Neumann algebras in the non-tracial setting, J. Lond. Math. Soc. (2) 88 (1) (2013) 25–48.
- [51] S. Vaes, Strictly outer actions of groups and quantum groups, J. Reine Angew. Math. 578 (2005) 147–184.
- [52] R. Vergnioux, K-amenability for amalgamated free products of amenable discrete quantum groups, J. Funct. Anal. 212 (1) (2004) 206–221.
- [53] C. Voigt, The Baum-Connes conjecture for free orthogonal quantum groups, Adv. Math. 227 (5) (2011) 1873–1913.
- [54] C. Voigt, On the structure of quantum automorphism groups, J. Reine Angew. Math. 732 (2017) 255–273.
- [55] R. Vergnioux, C. Voigt, The K-theory of free quantum groups, Math. Ann. 357 (1) (2013) 355-400.
- [56] J. Wahl, A note on reduced and von Neumann algebraic free wreath products, Ill. J. Math. 59 (3) (2015) 801–817.
- [57] S. Wang, Free products of compact quantum groups, Commun. Math. Phys. 167 (3) (1995) 671–692.
- [58] S. Wang, Quantum symmetry groups of finite spaces, Commun. Math. Phys. 195 (1) (1998) 195–211.
- [59] H. Wang, On representations of semidirect products of a compact quantum group with a finite group, Preprint, arXiv:1909.02359, 2019.
- [60] S.L. Woronowicz, Compact matrix pseudogroups, Commun. Math. Phys. 111 (4) (1987) 613–665.
- [61] S.L. Woronowicz, Tannaka-Kreĭn duality for compact matrix pseudogroups. Twisted SU(N) groups, Invent. Math. 93 (1) (1988) 35–76.
- [62] S.L. Woronowicz, Compact quantum groups, in: Symétries Quantiques, Les Houches, 1995, North-Holland, Amsterdam, 1998, pp. 845–884.